Public Document Pack



AGENDA

Committee Administrator: Democratic Services Officer (01609 767015)

Wednesday, 4 November 2015

Dear Councillor

NOTICE OF MEETING

Meeting PLANNING COMMITTEE

Date Thursday, 12 November 2015

Time 9.30 am

Venue Council Chamber, Civic Centre, Stone Cross, Northallerton

Yours sincerely

P. Morton.

Phillip Morton Chief Executive

To: Councillors Councillors

D A Webster (Chairman) J Noone P Bardon (Vice-Chairman) C Rooke

D M Blades Mrs I Sanderson

S P Dickins A Wake
G W Ellis Mrs J Watson
K G Hardisty S Watson

Other Members of the Council for information

PLEASE NOTE THAT THERE WILL NOT BE ANY MEMBER TRAINING

THE MEETING WILL COMMENCE AT 9.30AM WITH AN ADJOURNMENT FOR LUNCH AT 12.00PM RECONVENING FOR THE AFTERNOON SESSION AT 1.30PM (PLEASE SEE THE PLANS LIST INDEX FOR FURTHER DETAILS)

AGENDA

		Page No
1.	MINUTES	1 - 6
	To confirm the minutes of the meeting held on 15 October 2015 (P.13 - P.14), attached.	
2.	APOLOGIES FOR ABSENCE.	
3.	PLANNING APPLICATIONS	7 - 172
	Report of the Executive Director.	
	Please note that plans are available to view on the Council's website through the Public Access facility.	
4.	MATTERS OF URGENCY	
	Any other business of which not less than 24 hours prior notice, preferably in writing, has been given to the Chief Executive and which the Chairman decides is urgent.	

Agenda Item 1

Minutes of the meeting of the PLANNING COMMITTEE held at 10.00 am on Thursday, 15th October, 2015 at Council Chamber, Civic Centre, Stone Cross, Northallerton

Present

Councillor D A Webster (in the Chair)

Councillor P Bardon Councillor C Rooke

D M Blades Mrs I Sanderson
G W Ellis A Wake
K G Hardisty Mrs J Watson
J Noone S Watson

Also in Attendance

Councillor Mrs B S Fortune Councillor B Phillips

C Patmore

P.13 MINUTES

THE DECISION:

That the minutes of the meeting of the Committee held on 17 September 2015 (P.11 - P.12), previously circulated, be signed as a correct record.

P.14 PLANNING APPLICATIONS

The Committee considered reports of the Executive Director relating to applications for planning permission. During the meeting, Officers referred to additional information and representations which had been received.

Except where an alternative condition was contained in the report or an amendment made by the Committee, the condition as set out in the report and the appropriate time limit conditions were to be attached in accordance with the relevant provisions of Section 91 and 92 of the Town and Country Planning Act 1990.

The abbreviated conditions and reasons shown in the report were to be set out in full on the notices of decision. It was noted that following consideration by the Committee, and without further reference to the Committee, the Director had delegated authority to add, delete or amend conditions and reasons for refusal.

In considering the report(s) of the Executive Director regard had been paid to the policies of the relevant development plan, the National Planning Policy Framework and all other material planning considerations. Where the Committee deferred consideration or refused planning permission the reasons for that decision are as shown in the report or as set out below.

Where the Committee granted planning permission in accordance with the recommendation in a report this was because the proposal is in accordance with the development plan the National Planning Policy Framework or other material considerations as set out in the report unless otherwise specified below. Where the Committee granted planning permission contrary to the recommendation in the report the reasons for doing so and the conditions to be attached are set out below.

THE DECISION:

That the applications be determined in accordance with the recommendation in the report of the Executive Director, unless shown otherwise:-

(1) 15/01355/FUL - Construction of a detached dwelling with detached garage and associated access at Land adjacent to Westwood, Bagby for Ms Deborah Barker

PERMISSION GRANTED

(The applicant, Debbie Barker, spoke in support of the application.)

(2) 15/01917/FUL - Construction of a two storey extension to side of dwelling at 2 Pinewood Grove, Bedale for Mr P Balding

PERMISSION GRANTED

(3) 15/01809/OUT - Outline application for construction of a dwellinghouse with all matters reserved, for Mrs J Lancaster Carthorpe at Land opposite Rosedene

PERMISSION GRANTED

(4) 15/01821/FUL - Retrospective application for the alterations to dwelling, dormer extension, replacement windows, installation of conservation area roof lights and works to chimneys to form 2 self-contained flats at Kirkstone, Chapel Street, Easingwold for James Inman Joinery and Building Contractor

PERMISSION REFUSED for the following reasons:-

- 1. The scheme is contrary to Policies CP16 and DP28 of the Hambleton Local Development Framework; Part 12 of the National Planning Policy Framework and to the 1990 Planning (Listed Buildings and Conservation Areas) Act in that the design, scale and materials of the rear dormer window harmfully erodes the character and appearance of the Easingwold Conservation Area and fails to preserve and enhance the Easingwold Conservation Area.
- 2. The proposed development is contrary to the Hambleton Local Development Framework Policies CP1 and DP3 as it does not provide a minimum level of car parking and would give rise to an increase in congestion in Easingwold Town Centre that would be harmful to the safety and amenity of highway users and other residents. Furthermore contrary to the Hambleton Local Development Framework Policies CP16 and DP28 the congestion caused by parking of vehicles in public areas would harm the character and appearance of the Easingwold Conservation Area.

(The applicant, James Inman, spoke in support of the application.)

(5) 15/01335/FUL - Construction of a solar farm and associated infrastructure, access tracks, temporary construction compounds and security fencing at White House Farm Great Smeaton for Mr Mike Rogers

PERMISSION GRANTED

(The applicant, Andrew Leggett, spoke in support of the application.)

(The applicant's agent, Mike Rogers, spoke in support of the application).

(Mr Mann spoke objecting to the application.)

Disclosure of Interest

Councillor Mrs I Sanderson disclosed a personal interest and left the meeting during discussion and voting on this item.

The Chairman adjourned the meeting at 11.20am and the meeting reconvened at 1.30pm.

- (6) 15/00325/FUL Redevelopment of garage to provide convenience store, ATM, customer car park and associated petrol filling station at Greaves Garage, 36 Garbutts Lane, Hutton Rudby for James Hall and Company
 - PERMISSION GRANTED with amended Condition 3 to ensure the operating hours of the petrol filling station and the convenience store are the same.
 - (The applicant's agent, Tim Brown, spoke in support of the application).
 - (Geoff Cunningham spoke objecting to the application.)
- (7) 15/01543/OUT Outline application for the construction of a dwellinghouse at Land to the south west of Prospect House, Great Fencote for Mr D Ward
 - PERMISSION GRANTED subject to an additional condition concerning the repositioning of the hedge.
- (8) 15/01151/FUL Construction of a dwellinghouse at Church Farm, Myton on Swale for Mr and Mrs Revely
 - PERMISSION GRANTED
- (9) 15/01728/FUL Formation of additional 15 touring caravan pitches, single storey shower block, toilet block, disabled facilities, wash up area and drying room at The Oaks Fishing Lakes, Station Road, Sessay for Mr F&D Kay
 - PERMISSION GRANTED subject to an additional condition requiring details of a recreation area.
- (10) 14/02578/OUT Outline application for the construction of 25 residential dwellings with all matters reserved excluding access at White House Farm, Stokesley for Northumbrian Land Ltd

PERMISSION GRANTED

(The applicant's agent, Neil Morton, spoke in support of the application).

(Mike Cannavan spoke on behalf of Stokesley Parish Council objecting to the application.)

(Malcolm Smallwood spoke objecting to the application.)

(11) 15/01622/FUL - Construction of a dwellinghouse for at 1 Chapel View Cottages, Thirn Mr Asshheton Montagu Windsor Curzon-Howe-Herrick

PERMISSION REFUSED due to congestion and impact of additional on street parking and overlooking of neighbouring property.

The decision was contrary to the recommendation of the Executive Director.

(The applicant's agent, Dr James Gordon, spoke in support of the application).

(Chris Hartley spoke on behalf of Thirn Parish Council objecting to the application.)

(Linda Dyson spoke objecting to the application.)

(12) 15/01693/MRC - Proposed variation of condition 02 (to increase the number of holiday lodges from 11 to 34) of previously approved scheme (11/01989/FUL) for a change of use of an agricultural nursery to a caravan park (holiday lodges) with associated hardstanding, parking and landscaping at Hollin Barn Nurseries, Sutton Road, Thirsk for Evergreen Park Ltd.

PERMISSION REFUSED due to concerns about accessibility, highway safety, scale and for a development incompatible with the rural character and resulting impact on landscape.

The decision was contrary to the recommendation of the Executive Director.

(The applicant, Andrew Stephenson, spoke in support of the application.)

(Robin Bossomworth spoke objecting to the application.)

Disclosure of Interest

Councillor G W Ellis left the meeting before discussion and voting on this item.

(13) 08/04984/DCN - Proposed discharge of conditions attached to application 08/04984/FUL - Siting of 4 132m high wind turbines, associated works and a new vehicular access at OS Fields 0058, 6375, 4300 & 0001, land South of A684, Northallerton for Infinis Energy Services Ltd

PERMISSION GRANTED Conditions 6, 10, 11, 12, 14, 23, 24, 25, 27, 30, 31, 32, 33 and 34 were approved.

Authority to determine submissions in relation to Conditions 13 and 19 were delegated to the Executive Director (Planning).

(The applicant's agent, Julie Aitken, spoke in support of the application).

	(Mr Forster Holmes spoke on behalf of Osmotherley Area Parish Council objecting to the application.)
	(Mr Robert Birch spoke objecting to the application.)
The r	meeting closed at 5.10 pm
Chair	rman of the Committee
Oriali	man of the Committee



PLANNING APPLICATIONS

The attached list of planning applications is to be considered at the meeting of the Planning Committee at the Civic Centre, Stone Cross, Northallerton on Thursday 12 November 2015. The meeting will commence at 9.30am.

Further information on possible timings can be obtained from the Democratic Services Officer, Louise Hancock, by telephoning Northallerton (01609) 767015 before 9.00 am on the day of the meeting.

The background papers for each application may be inspected during office hours at the Civic Centre by making an appointment with the Executive Director. Background papers include the application form with relevant certificates and plans, correspondence from the applicant, statutory bodies, other interested parties and any other relevant documents.

Members are asked to note that the criteria for site visits is set out overleaf.

Following consideration by the Committee, and without further reference to the Committee, the Executive Director has delegated authority to add, delete or amend conditions to be attached to planning permissions and also add, delete or amend reasons for refusal of planning permission.

Mick Jewitt Executive Director

SITE VISIT CRITERIA

- 1. The application under consideration raises specific issues in relation to matters such as scale, design, location, access or setting which can only be fully understood from the site itself.
- 2. The application raises an important point of planning principle which has wider implications beyond the site itself and as a result would lead to the establishment of an approach which would be applied to other applications.
- The application involves judgements about the applicability of approved or developing policies of the Council, particularly where those policies could be balanced against other material planning considerations which may have a greater weight.
- 4. The application has attracted significant public interest and a visit would provide an opportunity for the Committee to demonstrate that the application has received a full and comprehensive evaluation prior to its determination.
- 5. There should be a majority of Members insufficiently familiar with the site to enable a decision to be made at the meeting.
- 6. Site visits will usually be selected following a report to the Planning Committee. Additional visits may be included prior to the consideration of a Committee report when a Member or Officer considers that criteria nos 1 4 above apply and an early visit would be in the interests of the efficiency of the development control service. Such additional site visits will be agreed for inclusion in consultation with the Chairman or Vice-Chairman of the Planning Committee.

PLANNING COMMITTEE Thursday 12th November 2015 9.30am

Item No	Application Ref/ Officer/Parish	Proposal/Site Description
1	15/01083/HYB Mrs J Laver Brompton/ Northallerton Page no. 11	Hybrid planning application for: 1. Full planning application for Phase 1 residential comprising 150 dwellings to the east of Darlington Road and 148 dwellings to the west of Stokesley Road, including earth works across the site, engineering works for drainage associated with Phase 1 residential, associated infrastructure, construction of strategic link road from Darlington Road to the west and Stokesley Road to the east including roundabouts and road bridge
		 crossing railway line and village green; and Outline planning application for a mixed use development of 900 dwellings (C3) which includes the residential development comprising Phase 1 referred to above, employment (B1, B2, B8), neighbourhood centre comprising shops (A1), restaurants (A3) and drinking establishments (A4), extra care facilities (C2), medical facilities (D1), primary school (D1), community uses including recreation playing pitches and allotments, car parking and means of access (all matters reserved apart from means of access)
		For: Mulberry Homes, Taylor Wimpey, Persimmon Homes and G, B, E & I Gill At: Land to the East of Darlington Road, West of Stokesley Road and North of Thurston Road Industrial Estate, Northallerton
		RECOMMENDATION: GRANT (OUTLINE and FUL)

13.30pm

Item No	Application Ref/ Officer/Parish	Proposal/Site Description
2	15/01324/FUL	Construction of dwellinghouse
2	Mr A J Cunningham	
	Dalton	For: Mr J Binks
		At: Dalton Lane, Dalton
	Page no. 95	
		RECOMMENDATION: GRANT
3	14/02285/FUL Mr T Wood Easingwold	Construction of 116 dwellings with associated access, open spaces and landscaping as amended by details received 30 June 2015
	Page no. 101	For: Persimmon Homes (Yorkshire)Ltd At: Land to the east of Kellbalk Lane, Easingwold
		RECOMMENDATION: GRANT

Item No	Application Ref/ Officer/Parish	Proposal/Site Description		
4	15/01838/OUT Mrs B Robinson	Outline consent for single detached dwelling		
_	East Harlsey	For: Mr S Barr		
	Lastrianscy	At: Land adjacent to the Cat and Bagpipes Inn		
	Page no. 119	East Harlsey		
		RECOMMENDATION: GRANT		
5	15/01542/OUT	Outline planning application for construction of one two storey		
J	Mrs H Laws	dwellinghouse as amended by email received on 23		
	Newby Wiske	September 2015		
	Page no. 125	For: Mr D Marwood		
		At: Land to the South of Willow Garth, Newby Wiske		
		RECOMMENDATION: GRANT		
6	15/01823/FUL	Construction of detached dwellinghouse and double garage		
0	Mrs B Robinson Potto	For: Mr J Graham		
	Follo	At: Land at the Old Orchard, Cooper Lane Potto		
	Page no. 133			
		DECOMMENDATION: CDANT		
	15/01897/OUT	RECOMMENDATION: GRANT Demolition of existing band room and construction of 4		
7	Mrs T Price	dwellings		
	Northallerton			
	D 400	For: Northallerton Silver Band		
	Page no. 139	At: The Band Room, Romanby Road, Northallerton		
		RECOMMENDATION: GRANT		
8	15/01446/FUL	Construction of detached three bedroom two storey dwelling		
U	Mr A J Cunningham Sessay	with attached garage to include access and means of enclosure		
	Jessay	enclosure		
	Page no. 145	For: Mr and Mrs Algie		
		At: Montrose, Main Street, Sessay		
		RECOMMENDATION: GRANT		
9	15/01943/REM	Application for Reserved Matters for the construction of 178		
J	Mr P Jones	dwellings		
	Stokesley	For: Taylor Wimpey (North Yorkshire) Ltd		
	Page no. 151	At: White House Farm, Stokesley		
	15/02139/FUL	RECOMMENDATION: GRANT Construction of a detached dwelling-house with domestic		
10	Helen Laws	garage.		
	East Harlsey	3		
		For: Mr & Mrs John Newcombe		
	Page no. 167	At: Bankside Farm, East Harlsey		
		RECOMMENDATION: GRANT		
L	L	commentscom onani		

Parish:BromptonCommittee Date:12th November 2015Ward:Northallerton North and BromptonTarget Date:15th October 20151Planning Officer:Janine Laver

Application Number: 15/01083/HYB

Proposal:

1. Full planning application for Phase 1 residential comprising 150 dwellings to the east of Darlington Road and 141 dwellings to the west of Stokesley Road, including earth works across the site, engineering works for drainage associated with Phase 1 residential, associated infrastructure, construction of strategic link road from Darlington Road to the west and Stokesley Road to the east including roundabouts and road bridge crossing railway line and village green; and

2. Outline planning application for a mixed use development of 900 dwellings (C3) which includes the residential development comprising Phase 1 referred to above, employment (B1, B2, B8), neighbourhood centre comprising shops (A1), restaurants (A3) and drinking establishments (A4), extra care facilities (C2), medical facilities (D1), primary school (D1), community uses including recreation playing pitches and allotments, car parking and means of access (all matters reserved apart from means of access).

The proposal is a major development, which is accompanied by an Environmental Statement.

Location:

Land To The East Of Darlington Road, West Of Stokesley Road, and North Of Thurston Road Industrial Estate, Northallerton

Applicant(s):

Mulberry Homes, Taylor Wimpey, Persimmon Homes, Messrs G, B, E & I Gill

EXECUTIVE SUMMARY

The application site forms the majority of a strategic allocation (NM5) within the adopted Hambleton Local Development Framework and the proposal comprises a total of 900 dwellings, mixed uses including retail, commercial, and extra care provision, a primary school, formal and informal recreation space including a village green, and a link road and bridge over the Northallerton-Middlesbrough railway line through the centre of the development.

The application is in hybrid form, with full planning permission sought for the link road and bridge, 291 of the 900 dwellings, a village green near Darlington Road and sustainable drainage features. Outline approval is sought for the balance of the development.

As the largest allocation site in the Local Development Framework, which has undergone extensive consultation, independent review, adoption and masterplanning, the principle of the uses applied for in the application, as well as the quantity of housing proposed, has been established.

The proposal was received in May 2015 and advertised to the general public by way of letter (334 sent in total), site notices displayed in the surrounding area, a press notice and Hambleton DC website releases. Statutory consultee notification was also undertaken in accordance with the relevant Town and Country Planning and Environmental Impact Regulations. To date 15 objections have been received from the public primarily raising issues of the lack of housing need in Northallerton, traffic congestion, the lack of a solution to queuing at the Low Gates rail crossing, off site flood impacts, and noise impacts to future residents from Allerton Steel operations. Statutory consultees raised some concerns regarding drainage and flooding, potential ecological impacts, lack of formal cycle provision within the scheme, noise impacts from Allerton Steel and the location of the school. Brompton and Northallerton Town Councils have also objected on various grounds.

The comments made by statutory bodies, interested parties and private individuals have been taken into account. However, it is concluded that for the reasons set out in this report the objections do not amount to reasons for refusal of the application either individually or collectively, and that the relevant planning issues that have been raised can be adequately addressed by the imposition of a range of planning conditions and/or legal agreement. Furthermore, design refinements will also undoubtedly occur in the detailed design phase in advance of submission of reserved matters.

It is considered that the Environmental Statement contained sufficient information to enable an assessment of the main or likely significant effects and appropriate mitigation measures, while subject to various conditions being met, and bearing in mind that reserved matters applications will need to be submitted for all development not included within the 'Full' part of the application, the proposal is satisfactory in terms of the mix of uses being sought and the traffic generated. Furthermore, with regards to the Full Application for 291 houses, the link road and bridge, SUDs basins and Village Green the proposal is considered to be satisfactory in terms of layout, architectural expression and urban design and will largely accord with the provisions of the Site Allocations Policy and the guidance within the adopted 2011 Masterplan for the site.

The development would also realise the North Northallerton Link Road, which is a major piece of infrastructure on the Council's priority list for spending receipts under the Community Infrastructure Levy and the subject of a £6 million Government Grant issued via the Local Enterprise Partnership. It would also help deliver a new primary school and recreational land, which would be of wider benefit to local communities.

The proposal provides the Council with the opportunity to meet part of its required housing provision, and based upon the viability appraisal can provide the district with some essential affordable housing, albeit below the policy level. In any event, due to the proposed review mechanism to be contained within an associated legal agreement, there may be opportunity for increased affordable housing provision at later phases within the development if the housing market improves.

Overall, the proposal is considered acceptable and it is recommended that planning permission be granted in OUTLINE for a mixed use development of 900 dwellings (Use Class C3), employment (Use Classes B1, B2, B8), neighbourhood centre comprising shops (Use Class A1), restaurants (Use Class A3) and drinking establishments (Use Class A4), extra care facilities (Use Class C2), medical facilities (Use Class D1), primary school (Use Class D1), community uses including recreation playing pitches and allotments, car parking and means of access (all matters reserved apart from means of access) on land to the East of Darlington Road, West of Stokesley Road, and North of Thurston Road Industrial Estate, Northallerton

It is also recommended that planning permission be granted in FULL for Phase 1 of the 900 dwellings granted approval in Outline by this permission, comprising 150 dwellings to the east of Darlington Road and 141 dwellings to the west of Stokesley Road, including earth works across the site, engineering works for drainage associated with Phase 1 residential, associated infrastructure, construction of strategic link road from Darlington Road to the west and Stokesley Road to the east including roundabouts and road bridge crossing the Northallerton to Middlesbrough railway line and a village green on land to the East of Darlington Road, West of Stokesley Road, and North of Thurston Road Industrial Estate, Northallerton.

1.0 PROPOSAL

- 1.1 The applicants have lodged a hybrid planning application comprising a) an 'outline' application for the entire site identified on the 'Proposed Application Boundary Plan' Nod. 0000-0001, dated Jan 2014, as well as b) a 'full' application for the land identified on 'Phase 1 Red Line Boundary Plan' Nod. 12003-SK-01, Rev B, dated August 2015. The total site area of the 'outline' application is 52.8 hectares.
- 1.2 An application for outline planning permission allows for a decision on the general principles of how a site can be developed. Outline planning permission is granted subject to conditions requiring the subsequent approval of one or more *'reserved matters'*.
- 1.3 The 'outline' application comprises:
 - 900 residential dwellings, comprising 2, 3, 4 and 5 bedrooms, mostly of two storey scale but some bungalows are also proposed;
 - 3,250 sqm of employment uses including B1 Commercial, B2 General Industrial and B8 Storage and Distribution;
 - A neighbourhood centre including A1 Shops, A3 Food and Drink, A4 Drinking Establishments and C1 Hotels uses (1,965 sqm floor area in total);
 - A care home (Use Class C2);
 - A primary school (Use Class D1);

- Community uses in the form of recreational playing fields/pitches, playgrounds, allotments and a Village Green;
- Associated car parking for the aforementioned uses;
- Flood and surface water retention basins/tanks, known as SUDs; and
- A link road and bridge to cross the Northallerton to Middlesbrough railway line.
- 1.4 The 'outline' application is detailed on the Plan entitled 'Illustrative Masterplan Scaled' Nod. 0000-0019 Rev A, dated May 2015, to which any planning permission should be tied. This Plan provides a guide to where the various uses sought may be located, with further detail provided in plans contained within the submitted Design and Access Statement.
- 1.5 However, it must be highlighted that the 'Layout', 'Scale', 'Appearance' and 'Landscaping' of the development are all matters that have been *'reserved'* for later determination, which means that the detail provided on the aforementioned 'Illustrative Masterplan Scaled' or within the Design and Access Statement is subject to change. However, those documents are intended to illustrate the quality of development that could be achieved within the parameters of the permission now sought. The Council is therefore only considering the principle of the uses and their potential locations, as well as the 'Access' to the wider site, which is a matter that is not reserved for later determination.
- 1.6 With regard to 'Access' the application is accompanied by details of the new main access points which is in the form of two new roundabouts on Darlington Road to the western extent of the site and Stokesley Road to the eastern extent of the site, which will serve development to both the north and the south of the strategic allocation. The extent of the initial infrastructure works is shown on 'Proposed Link Road Alignment and Access Arrangements Plan' Nod. 12003/GA/11 Rev G, dated September 2015, and produced by Optima Highways Solutions.
- 1.7 The 'full' application (hereafter known as Phase 1) was submitted in detail and comprises:
 - A total of 291 dwellings (which are included within the 900 dwellings sought as part of the 'outline' application, rather than in addition to the 900 dwellings).
 - The 291 dwellings will be split across the site with 150 dwellings provided to the east of Darlington Road (known in this application as the Persimmon Scheme) and 141 dwellings provided to the west of Stokesley Road (known in this application as the Taylor Wimpey Scheme). For clarity, the application originally included 298 houses in the Phase 1 application but during the course of the application the Taylor Wimpey housing layout was amended, which reduced the proposal by 7 dwellings.
 - The Persimmon Scheme is shown on the 'Proposed Site Plan' Nod. YOR.2457.004B, dated 9/9/15 and produced by Pegasus Urban Design. The Taylor Wimpey Scheme is shown on the 'Planning Layout Plan' Nod. 00C, dated 26/10/15 and produced by Taylor Wimpey.
 - The link road and bridge through the centre of the 'outline' site, is shown on the 'Proposed Link Road Alignment and Access Arrangements Plan' Nod. 12003/GA/11 Rev G, dated September 2015, and produced by Optima Highways Solutions;
 - Flood and surface water retention basins/tanks, known as SUDs; and
 - The Village Green.
- 1.8 The Persimmon residential scheme of 150 dwellings is broken down into the following sizes:

Size	Height		
35 x 2 bed dwellings	6 x 1 storey dwellings		
94 x 3 bed dwellings	100 x 2 storey dwellings		
21 x 4 bed dwellings	44 x 2.5 storey dwellings		
The layout comprises a mix of detached, semi-detached, and terraces of 3-4 dwellings, some			
with integral garages, detached garages, driveway only spaces, or shared parking courts.			

1.9 The Taylor Wimpey residential scheme of 141 dwellings is broken down into the following sizes:

Size	Height
18 x 2 bed dwellings	7 x 1 storey dwellings
42 x 3 bed dwellings	120 x 2 storey dwellings

75 x 4 bed dwellings	14 x 2.5 storey dwellings
6 x 5 bed dwellings	
The layout comprises a mix of detached and	semi-detached dwellings, some with integral

The layout comprises a mix of detached and semi-detached dwellings, some with integral garages, detached garages or driveway only spaces.

- 1.10 It is proposed to construct the dwellings using a range of soft red and brown brickwork interspersed with off-white render to break up the streetscene. From the house style information provided, roofs will be finished with slate or natural clay pantiles.
- 1.11 Landscape plans for the Phase 1 development were submitted during the course of the application, and have undergone several revisions. Therefore the plans submitted in October 2015 are the ones for consideration in this application; the plans numbers are listed in Section 1.17 below.
- 1.12 The development will be delivered in phases. Phase 1 comprises all the elements of the 'full' application. It was initially suggested by the applicants that the link road and bridge would not be constructed until 495 dwellings have been built; however, it has been agreed by the Council, the applicants and NYCC Highways that it will be constructed within the early period of the Phase 1 development (i.e. by the end of October 2017).
- 1.13 In terms of the remainder of the housing and employment development, this will be developed in accordance with market demand, although reserved matters applications will need to be submitted and approved prior to commencement of any development of those 'outline' elements. It is also anticipated that the remaining elements will be delivered as follows:
 - Sports facilities land will be acquired by the Council in Phase 2 and developed when enough funds have been accrued.
 - The new primary school land will be acquired by NYCC in Phase 1 and developed to be ready for a September 2018 intake.
 - The retail element of the neighbourhood centre will be delivered in Phase 2.
 - The aged care facility will be delivered in Phase 2.
- 1.14 The application was supported by a package of submission documents including:
 - A Planning Application Form and Landowner Notification Certificates;
 - Planning, Sustainability and Open Space Statement;
 - Design and Access Statement (incorporating Masterplan Parameters and Design Principles);
 - Statement of Community Involvement;
 - Flood Risk Assessment including Drainage Strategy (3 separate reports);
 - Geo-Environmental Report (2 separate reports);
 - Phase 1 Habitat Survey (also entitled Preliminary Ecological Survey)
 - Various Ecological and Biodiversity surveys, including bats and breeding birds;
 - Transport Assessment (subsequently updated on 09/09/15 and 24/09/2015);
 - Travel Plan (subsequently updated on 01/10/15);
 - Agricultural Land Use Classification Assessment;
 - Archaeological Desk Based Assessment;
 - Landscape and Visual Impact Assessment;
 - Tree and Hedgerow Survey;
 - Utilities Assessment;
 - A Retail Impact Assessment; and
 - A variety of plans including (but not limited to):
 - Proposed Application Boundary Plan, Nod. 0000-0001, Rev B, dated Jan 2014, and produced by Spawforths;

- Phase 1 Red Line Boundary Plan, Nod. 12003-SK-01, Rev B, dated August 2015, and produced by Optima Highways Solutions;
- Illustrative Masterplan Scaled, Nod. 0000-0019 Rev A, dated May 2015, and produced by Spawforths;
- Proposed Link Road Alignment and Access Arrangements Plan, Nod. 12003/GA/11 Rev F, dated April 2015, and produced by Optima Highways Solutions:
- Bridge Over Railway General Arrangement Plan, Nod. BHB-DRG-1000, Rev 03, dated March 2015, and produced by Buro Happold Engineering;
- Taylor Wimpey Planning Layout Plan, Nod. NN:00, dated 18 May 2015, and produced by Taylor Wimpey (superseded); and
- Persimmon Homes Proposed Site Plan, Nod. YOR.2457.004 Rev A, dated 01 May 2015, and produced by Pegasus Urban Design (superseded).
- 1.15 An Environmental Statement with appended technical reports and a non-technical summary was also subsequently submitted on 15th June 2015. This was advertised in accordance with Regulation 17 of the Town and Country Planning (Environmental Impact Assessment) Regulations 2011.
- 1.16 The application has been amended at various stages with the submission of new or clarified detail, such as a Retail Impact Assessment for a 1,723 sqm supermarket (and amended Traffic Impact Assessment), and revised plans as a result of more detailed design work, at the request of the planning officer or statutory consultees.
- 1.17 The additional and revised plans include:
 - Persimmon Homes Proposed Site Plan, Nod. YOR.2457.004 Rev B, dated 09 Sept 2015, and produced by Pegasus Urban Design.
 - Persimmon Homes Proposed Site and Street Scenes Plan, Nod. YOR.2457.003 Rev B, dated 09 Sept 2015 and produced by Pegasus Urban Design.
 - Persimmon Homes Proposed Planting Plan, Nod. YOR.2457.001, dated 09 Sept 2015.
 - Taylor Wimpey Planning Layout Plan, Nod. NN:00 Rev C, dated 26 Oct 2015, and produced by Taylor Wimpey.
 - Taylor Wimpey Landscape Plan Sheet 1 of 3, Nod. 2585/1 Rev H, dated 02 Nov 2015, and produced by Rosetta Landscape Design.
 - Taylor Wimpey Landscape Plan Sheet 2 of 3, Nod. 2585/2 Rev H, dated 02 Nov 2015, and produced by Rosetta Landscape Design.
 - Taylor Wimpey Landscape Plan Sheet 3 of 3, Nod. 2585/3 Rev H, dated 02 Nov 2015, and produced by Rosetta Landscape Design.
 - Landscape Masterplan for Spine Road, Nod. YOR.2457.010C Sheet 1 of 5, dated Oct 2015, and produced by Pegasus Urban Design.
 - Landscape Masterplan for Spine Road, Nod. YOR.2457.006C Sheet 2 of 5, dated Oct 2015, and produced by Pegasus Urban Design.
 - Landscape Masterplan for Spine Road, Nod. YOR.2457.007C Sheet 3 of 5, dated Oct 2015, and produced by Pegasus Urban Design.
 - Landscape Masterplan for Spine Road, Nod. YOR.2457.008C Sheet 4 of 5, dated Oct 2015, and produced by Pegasus Urban Design.
 - Landscape Masterplan for Spine Road, Nod. YOR.2457.009E Sheet 5 of 5, dated Oct 2015, and produced by Pegasus Urban Design.
 - Proposed Link Road Alignment and Access Arrangements Plan, Nod. 12003/GA/11
 Rev G, dated September 2015, and produced by Optima Highways Solutions.

2.0 SITE DESCRIPTION

2.1. The site is located in Northallerton approximately 1 km north of the Northallerton town centre (or in walking terms 15 – 20 Minutes away). The site has an overall area of approximately

52.8 hectares and forms a natural extension to the northern edge of the development limits of Northallerton. To the west and to the east the site is bounded by the A167 Darlington Road and the A684 Stokesley Road respectively. Thurston Road industrial estate lies to the south alongside a range of other uses including Hambleton District Council's offices, a leisure centre and a small cluster of residential properties. To the north lies open countryside with Northallerton Rugby Club and the settlement of Brompton beyond. The site is also bisected by the Northallerton to Middlesbrough train line running approximately north/south and roughly by area two thirds to the west and a third to the east.

- 2.2. The site is largely undeveloped and comprises farmland under several ownerships. Between Darlington Road and Northallerton Road the landscape is one of large arable fields with trimmed hedgerows and sporadic hedgerow trees. Immediately adjacent to Darlington Road smaller field sizes are present and some of the hedges have been replaced by fences and these fields are grazed rather than tilled. Between Northallerton and Stokesley Roads the site encompasses smaller fields at the southern end but larger fields at the northern end.
- 2.3. There are a number of mature trees dotted about the site, but there are only two small woodland blocks in the area and both are outside the site; one lies just west of the northern most part of the site and lies in the grounds of Strikes garden centre off the Darlington Road, and the other lies just east of Northallerton Road adjacent to the playing fields, but is actually a well-treed private garden. The odd dwelling and various farm buildings are scattered across the site, but for the most part the site is agricultural and appears as countryside on the fringe of settlements to the north and south.
- 2.4. The site also contains the following existing features which all generally align north-south:
 - The Northallerton to Middlesbrough railway line;
 - Northallerton Road which connects Brompton with Northallerton; and
 - Two watercourses including Brompton Beck (also known as North Beck).
- 2.5. In terms of topography the site is not constrained by any severe gradients, however ground levels do undulate across the full extent of the site following a general trend of sloping towards the Brompton Beck watercourse, which, as noted above runs approximately north/south through the centre of the site. In addition, the land also gently rises on the east and west fringes near Stokesley and Darlington Roads.
- 2.6. The centre of the site is covered by Flood Zone 3 designation (area of high risk of flooding); however, only the link road and bridge would be built on flood land, with compensatory flood storage created elsewhere to offset the loss in this location.
- 2.7. The site boundary is shown on 'Proposed Application Boundary Plan' Nod. 0000-0001, Rev B, dated Jan 2014. The application site does not include the development land parcels to the west of Darlington Road or to the east of Stokesley Road which are shown in the 2011 original Masterplan Framework document.
- 2.8. The entire site forms part of strategic site allocation NM5 North Northallerton Area, within the LDF, as set out in the Allocations Development Plan Document (DPD).

3.0 PLANNING HISTORY AND POLICY BACKGROUND

- 3.1. Section 38(6) of the Planning and Compulsory Purchase Act (2004) requires that proposals be determined in accordance with the Development Plan unless material considerations indicate otherwise.
- 3.2. The Development Plan for Hambleton is known as the Local Development Framework (LDF) and comprises:
 - (i) The Core Strategy Development Plan Document (adopted April 2007);
 - (ii) The Development Policies Development Plan Document (adopted February 2008); and
 - (iii) The Allocations Development Plan Document (adopted December 2010).
- 3.3. Supplementary Planning Documents (SPDs) relevant to this application include Open Space, Sport and Recreation (adopted February 2011); Affordable Housing (adopted April 2015); and Size, Type and Tenure of New Homes (adopted September 2015). Also relevant is the Council's Statement of Community Involvement (adopted March 2006 and revised July 2013) and the Community Infrastructure Levy Guide, Guidance Notes, Charging Schedule and

Regulation 123 List (dated March and April 2015).

- 3.4. Other material considerations which need to be taken into account include the National Planning Policy Framework (NPPF), March 2012, the associated Planning Practice Guidance (PPG) issued in March 2014 and regularly updated, and the Community Infrastructure Levy (CIL) Regulations 2010 as amended.
- 3.5. As noted above, the application site has been identified by the Council as a strategic site for mixed development within the Allocations DPD Policy NM5, as adopted by Full Council on 21 December 2010. The DPD states that proposal NM5 forms a strategic area for development in the Allocations DPD providing significant housing and employment development and also providing recreation, leisure and community facilities and strategic infrastructure.
- 3.6. The site represents the largest allocation in the Council's Local Development Framework and is highlighted as a priority in the North Yorkshire Housing Strategy and Local Investment Plan. The principle of the development of this site for housing, mixed uses and a link road has therefore been long established having been the subject of an independent examination in public by a Secretary of State appointed Planning Inspector. The examination agreed that the North Northallerton Development Area was the preferred option for further development of the town and found the LDF documents sound.
- 3.7. Allocation NM5 is divided into 7 sub areas, known as NM5A to NM5G. The exact wording of Policy NM5 is reproduced in full below:

NM5 - North Northallerton Area

This area is identified as a strategic site for mixed development, comprising the following uses:

NM5A: West of Northallerton - Middlesbrough Railway (5.7ha)

Allocated for housing development at a density of approximately 35 dwellings per hectare, resulting in a capacity of around 200 dwellings (of which a target of 40% should be affordable); for development throughout Phases 2 (2016-2021) and 3 (2021-2026).

NM5B: West of Stokeslev Road (6.5ha)

Allocated for housing development at a density of approximately 30 dwellings per hectare, resulting in a capacity of around 200 dwellings (of which a target of 40% should be affordable); for development in Phase 3 (2021-2026).

NM5C: East of Stokesley Road (5.5ha)

Allocated for housing development at a density of approximately 30 dwellings per hectare, resulting in a capacity of around 165 dwellings (of which a target of 40% should be affordable); for development in Phase 3 (2021-2026).

NM5D: East of Darlington Road (17.2ha)

Allocated for mixed housing and employment development;

Housing development: (11.4ha)

At a density of approximately 35 dwellings per hectare, resulting in a capacity of around 400 dwellings (of which a target of 40% should be affordable); for development in Phase 2 (2016-2021);

Employment development: (5.8ha) B1 and B2 employment development.

NM5E: North of Standard Way Industrial Estate (5.7ha)

Allocated for B1, B2 and B8 employment development.

NM5F: North Northallerton Link Road

This scheme will provide a single carriageway road of a width of approximately 8 metres, a footpath and a cycleway route, a bridge across the Northallerton-Middlesbrough railway and access to a number of sites within the North Northallerton Area.

NM5G: West of Northallerton Road (7.8ha)

Allocated for recreational development.

Development will be subject to the following requirements:

- i. Developer contributions towards improvements to the sewerage an sewage disposal infrastructure prior to any development taking place;
- ii. Flood alleviation measures being undertaken, funded by the Environment Agency and the developers, prior to any development taking place;
- iii. Contributions from the developer towards public open space and the North Northallerton Link Road:
- iv. Contributions from the developer towards the provision of additional school places and local health care facilities as necessary;
- v. Type and tenure of housing meeting the latest evidence on local needs;
- vi. Opportunities to incorporate sustainable and renewable energy generation should be maximised:
- vii. Incorporation of high quality landscaping on the boundaries and within each development site to maximise the visual appearance of the development; and
- viii. Reinforcement of the visual separation of the two settlements of Northallerton and Brompton and the separate identity of Brompton.
- 3.8. For the avoidance of doubt, allocations NM5C and NM5E do not form part of the subject planning application but are expected to be the subject of separate planning applications in the future. This planning application is therefore only concerned with development of the land allocations between Darlington Road and Stokesley Road.
- 3.9. It is also necessary to highlight that due to the passing of intervening legislation since the adoption of the LDF, such as the Community Infrastructure Levy Regulations, a number of the development contributions requirements contained within Policy NM5 will have altered or no longer be relevant.
- 3.10. In addition, it is necessary to highlight that in December 2013 the Council agreed to remove the phasing within Development Policy DP11 for the LDF's allocated housing sites through to 2026 (and thus also to the phasing suggested in Allocations Policy NM5) so as to help bring sites forward more quickly to boost housing supply. In this regard, the phasing identified within those policies is no longer applicable.
- 3.11. Finally, it is important to identify that following the adoption of the Allocations DPD a Masterplan Framework and Design Guidance document was prepared for the NM5 Allocation area by Taylor Young consultants, which provided conceptual design detail and expanded on the aspirations for the development of the entire site. The Masterplan was adopted by the Council in May 2011 and is a material consideration in the determination of this application.

4.0 RELEVANT PLANNING POLICIES

4.1 The following policies are of particular relevance to the determination of this application:

Core Strategy Policy CP1 - Sustainable development

Core Strategy Policy CP2 - Access

Core Strategy Policy CP3 - Community assets
Core Strategy Policy CP4 - Settlement hierarchy

Core Strategy Policy CP5a - Annual targets for housing completions by area

Core Strategy Policy CP6 - Housing distribution

Core Strategy Policy CP8 - Type, size and tenure of housing

Core Strategy Policy CP9 - Affordable housing

Core Strategy Policy CP14 - Retail and town centre development

Core Strategy Policy CP16 - Protecting and enhancing natural and man-made assets

Core Strategy Policy CP17 - Promoting high quality design
Core Strategy Policy CP18 - Prudent use of natural resources

Core Strategy Policy CP19 - Recreational facilities and amenity open space

Core Strategy Policy CP20 - Design and the reduction of crime

Core Strategy Policy CP21 - Safe response to natural and other forces

The Development Policies DPD sets out detailed policies for controlling development and delivering the vision, objectives and Core Policies of the Core Strategy DPD. Its policies play a key role in determining planning applications, by explaining practical ways of implementing the Core Policies. The following policies are of particular relevance to the determination of this application:

Development Policy DP1 - Protecting amenity

Development Policy DP2 - Securing developer contributions

Development Policy DP3 - Site accessibility
Development Policy DP4 - Access for all

Development Policy DP6 - Utilities and infrastructure

Development Policy DP8 - Development limits

Development Policy DP9 - Development outside development limits

Development Policy DP13 - Achieving and maintaining the right mix of housing Development Policy DP15 - Promoting and maintaining affordable housing

Development Policy DP16 - Specific measures to assist the economy and employment

Development Policy DP23 - Major out of town shopping proposals

Development Policy DP24 - Other retail uses
Development Policy DP29 - Archaeology

Development Policy DP30 - Protecting the character and appearance of the countryside Protecting natural resources: biodiversity/nature conservation

Development Policy DP32 - General design
Development Policy DP33 - Landscaping
Development Policy DP34 - Sustainable energy

Development Policy DP37 - Open space, sport and recreation

Development Policy DP38 - Major outdoor recreation Development Policy DP39 - Recreational links

Development Policy DP43 - Flooding and floodplains
Development Policy DP44 - Very Noisy Activities

The detail contained within the Allocations Development Plan Document referred to in Section 3 above is also directly relevant as well as the following Supplementary Planning Documents:

- Open Space, Sport and Recreation SPD, adopted February 2011
- Affordable Housing SPD, adopted 7th April 2015
- Community Infrastructure Levy Guidance Notes, Charging Schedule and Regulation 123 List, all adopted in 2015
- Size, Type and Tenure of New Homes, adopted 15th September 2015.

5.0 CONSULTATIONS

- 5.1 The application was advertised to the general public by way of letter, site notices, press notice, and Hambleton DC website releases. Statutory consultee notification was also undertaken in accordance with the relevant Town and Country Planning and Environmental Impact Regulations. In addition, the later submission of a Retail Impact Assessment and updated Traffic Assessment was subsequently advertised in the Local Press in accordance with Section 22 of the Environmental Impact Assessment Regulations 2011 and the Council's public access system has been continually updated when new plans/reports are received.
- 5.2 The following sections reveal details only from those consultees that provided a response. Note that some responses are provided in summary and some in full verbatim. For those that are provided in full verbatim they feature in *italicised* text. Officer comment may follow where clarifications/updates are considered to be necessary. All abridged consultee details and comments are, however, available in full on the Council's website using the Public Access system.

5.2.1 NYCC Highway Authority

NYCC Highways were integral to the development of the Masterplan for the site and have participated in a considerable number of meetings with the applicants and the Council's planning officers to design a scheme that would appropriately offset the impact of the development itself in traffic terms, but also provide improvements to Northallerton generally. The consultation reply is lengthy and is noted verbatim below. Suggested conditions will feature in Section 7 of this report.

This application will have significant impacts on the highway network in Northallerton, some of which will be positive. The documents submitted in support of the planning application included a Transport Assessment (TA) and Framework Travel Plan (FTP), which have assisted the Local Highway Authority (LHA) in assessing the impact of the proposed development. Amendments to both documents and submitted plans have been provided to secure proposals that are satisfactory to the LHA. The site is

allocated for development and includes a link road in the allocation. The submissions in support of the application cover the following matters:-

The Link Road

- ii. The site is accessed by a link road through the site connecting the A167 Darlington Road to the West with the A684 Stokesley Road to the East and includes a bridge over the Northallerton to Middlesbrough railway line. There will be roundabouts on the A167 and A684 where the link road joins these routes.
- iii. The LHA has been working with the developers for several years to secure an acceptable design for the link road; the submitted design reflects these discussions and is in line with current design principles. Consequently the purpose and design of the link road has evolved since its designation as a 'village road' in the Hambleton Allocations Plan and the Masterplan for the site.
- iv. The role of the link road has been acknowledged as a 'strategic link' in the overall Northallerton highway network. This has been a key decision as it acknowledges that the link road will serve more than the application site and provide some relief to key junctions in the town.
- v. The link Road will provide an east-west route for Northallerton north of Low Gates level crossing. Once the link road is open to through traffic the directional signing in the area will be amended, for example traffic travelling south on the A167 will be directed along the link road to the Friarage hospital, Thirsk, etc. and traffic travelling on the A684 will be directed along the link road to Standard Way and other businesses around Darlington Road. This will remove HGVs from Brompton Road, Friarage Street and Low Gates Level Crossing. The removal of HGVs from these routes is seen as a positive benefit to the highway network. The link will also remove non-HGV rat-running traffic from Quaker Lane. Once the development is completed and all buildings are occupied the modelling has shown there will be an overall reduction in vehicles crossing over the Low Gates Level Crossing.
- vi. A detailed traffic model of the existing highway network in Northallerton was created by Jacobs UK in 2006 to support Hambleton DC's allocation of the North Northallerton site for housing; this was updated and revalidated in 2012. The model has enabled the distribution of traffic flows to be considered in various situations with and without the proposed bridge over the railway. The traffic flows used in the model have been built up from surveyed existing flows with allowances for predicted traffic growth and other significant sites in Northallerton where development has been permitted. Only the significant sites that will have a quantifiable impact on the highway network have been specifically included in the analysis as the national forecasts for traffic growth include allowances for development led growth. This is a proportionate response. Smaller sites will have been permitted in the review period but will fall below the threshold of approximately 50 dwellings where a site specific TA is required. In these cases the annual traffic growth figures take account of the development traffic.
- vii. The analysis has shown that in 2026 with all the housing built and the link road open to through traffic there will be a small decrease in traffic passing through the Low Gates Level Crossing. This will be a positive benefit to the highway network notwithstanding the additional traffic from the development.
- viii. The applicant has also shown several scenarios whereby a number of houses could be built and occupied before the bridge needs to be in place. The trigger point has been established to be the development of 495 houses. This would be the point where junctions in the town would need relief without the link road. As it was unclear how these 495 dwellings would be distributed between the eastern and western sides of the site various scenarios were tested. They were:

West of the railway 495 dwellings 302 dwellings 136 dwellings 395 dwellings

ix. It is considered that the impact of these scenarios is acceptable as an interim situation prior to the completion of the bridge and opening of the link road to through

traffic. However external funding has been sought to allow construction of the link road in advance of the 495 dwelling trigger to provide the benefits of the link sooner thus maximising its impact. Realistically it is expected that the link road and bridge will be opened to through traffic in autumn 2017 due to the Growth Deal Funding contribution which has been sought to part fund the link road and rail bridge and which has a 'use by' date attached. This anticipated early completion of the road and bridge will be a positive benefit to the overall highway network.

- x. Overall, while the scenario testing has demonstrated that 495 dwellings could be built before the full link road is necessary, the delivery of 495 dwellings and no prospect of the link road open to through traffic is not acceptable. The occupation of a number of houses prior to the construction of the bridge is considered necessary however, to allow sufficient funds to be amassed by the developers to fund the bridge. However, now the Growth Deal funding is available the need to wait for the trigger point should not be necessary. There will need to be adequate controls either in the permission or through legal agreement(s) to ensure the through route is built in its entirety and development of the full link road does not stall with just two cul-de sacs either side of the railway. These controls will need to include adequate assurances for the County Council that all the land required to deliver the road and bridge will be available when required.
- xi. With regard to non-motorised users, the link road will have a footway to one side and a footway cycleway to the other. This is considered an appropriate level of provision for non-motorised users of the highway and will be linked with three formal crossing points of the link road at key locations: near the neighbourhood centre, near the school and near Northallerton Road. The shared footway cycleway will enable travel to the proposed school (north of the link road) by cycle from both sides of the link road using the formal crossing points. Furthermore, the benefit of a footway only is that some pedestrians prefer and sometimes feel safer not using a shared footway cycleway. Reserved matters applications for development of the land to the south will need to include links which direct the pedestrians and cyclists to the formal crossing points. These provisions for pedestrians and cyclists are considered acceptable.
- xii. No provision for horses has been made or is considered necessary as the development site does not link in with the local bridleway network and there is limited evidence of need. Nevertheless, it is considered appropriate to include a Condition requiring an Audit of Non- Motorised Users to be undertaken on the detailed design. The processes for the Audit are set nationally and will include a review of the provisions for all non-motorised users including cyclists and horses.

The bridge

- xiii. A new bridge is to be constructed over the Middlesbrough to Northallerton line. It will need to fully comply with both Network Rail's technical requirements and those of the LHA. The bridge will become part of the County Council's bridge stock and be maintained by the County Council in the usual way. There will need to be legal Agreements to secure the necessary permissions from Network Rail and NYCC for the delivery of the bridge following the granting of any permission.
- xiv. As indicated in paragraph x above, the non-delivery of the bridge is not acceptable to the LHA; it is an integral part of the overall link road.

Off-site highway works

- xv. In addition to the developer funded works linked to delivery of the link road the Transport Assessment identifies several other smaller items of infrastructure that the developers have agreed to deliver to mitigate the impact of the development. "Grampian" conditions will be used to secure works which are:
 - Mini-roundabout Brompton Road/ Northallerton Road/ Stokesley Road

Amendments to the existing mini-roundabout to improve capacity within the existing highway boundary;

Prior to the occupation of the 51st dwelling

 Mini-roundabout western end of Amendments to the existing mini-roundabout to improve capacity within the existing highway

Quaker Lane (North End)

boundary;

Prior to the occupation of the 51st dwelling

 Mini-roundabout eastern end of Quaker Lane (Brompton Road) Conversion of the mini-roundabout to a priority Tjunction with improved facilities for pedestrians across the end of Quaker Lane:

Within six months of opening of the Link Road to through traffic

Passenger transport issues

- xvi. There are existing bus services on the A167 and along Brompton Road between Northallerton and Brompton. It is proposed to provide additional services along the link road which are expected to be extensions of the Northallerton Town services. The exact details will be confirmed at a later date when the services are to be commissioned.
- xvii. To ensure the proposed additional bus services are given an opportunity to become established and financially viable 'pump priming funding' will be required. It is considered that funding of up to £100,000 per annum will be required for a period of five years. The trigger for the payments including their timing should be agreed in the \$106.
- xviii. Three pairs of bus stops are to be provided along the link road with shelters on the Northallerton bound side for waiting passengers.

The public right of way network

- xix. Formal comments indicating a right of way adjacent to the site have been issued by the LHA's Rights of Way team.
- xx. The rights of way network has no bridleways within the vicinity of the site and a footpath near the north western boundary of the site adjacent to the Strikes Garden Centre site. It is proposed to link this route in with the footway network on site. Within the residential phases the roads will be laid out in such a way that walking and cycling are encouraged. As noted above, the level of provisions for non-motorised users is considered to pass the tests at paragraph 204 of NPPF and be necessary, directly related and proportionate.

The Phase 1 Residential Area:- West of the A167 (Persimmon Homes)

- xxi. Pegasus Design drawing YOR.2457.004 Rev B has been submitted for this section of the site. The LHA considers this is suitable to base detailed design on. Some minor amendments are still required but all can be accommodated within the highway limits shown and none impact on the location of any dwelling. The layout will provide streets where vehicle speeds are naturally restrained to 20mph and walking routes are available on pedestrian desire lines feeding to the formal crossing points of the link road; this will also 'deliver safe routes to school' to the new school site.
- xxii. The parking provision is acceptable.
- xxiii. A Travel Plan for the Phase will need to be linked to the overarching Framework Travel Plan for the North Northallerton site and delivered.

The Phase 1 Residential Area:-East of the A684 (Taylor Wimpey)

- xxiv. The layout has needed to take account of the stream running through the site. It has also been agreed that it would be inappropriate to create any second access to the A684 beyond the proposed roundabout, as the roundabout will act as a 'gateway' to Northallerton, and creating a junction beyond the 'gateway' would reduce the benefits it provides, as well as potentially providing a rat run route. Pedestrian links to the A684 are being secured on key desire lines. The submitted drawing numbered NN:00 Rev C is considered suitable to base the detailed drawing submissions on.
- xxv. The parking provision is acceptable.
- xxvi. A Travel Plan for the Phase will need to be linked to the overarching Framework Travel Plan for the North Northallerton site and delivered.

xxvii. The LHA is now satisfied that the layout will provide streets where vehicle speeds are naturally restrained to 20mph and walking routes are available on pedestrian desire lines feeding to the formal crossing points of the link road; this will also 'deliver safe routes to school' to the new school site.

Future residential phases

xxviii. No details have been submitted with this outline application, with all matters except main access to the site reserved. The reserve matters applications will be required to deliver streets where vehicle speeds are naturally restrained to 20mph and walking routes are available on pedestrian desire lines feeding to the formal crossing points of the link road and 'safe routes to school' for the new school site. The LHA will provide advice when the relevant applications are submitted in the future.

The school

- xxix. A site has been identified which is acceptable to NYCC as LHA and Education Authority (LEA) subject to there being no direct access to the link road and the usual LEA requirements.
- xxx. Safe routes to school will be created within the new road layouts to encourage walking and cycling.
- xxxi. A School Travel Plan will be required to link to the Framework Travel Plan for the site and encourage the establishment of sustainable habits for travel to school. This will also cover the issue of school gate parking.

The Neighbourhood Centre and Employment Land

- xxxii. This is seen as a local facility to serve the residents of the development. As such it has been assumed that it will not generate any additional traffic outside the wider application site with the majority of trips being 'pass-by' in peak hours and a high proportion of walking and cycling. However a "sensitivity test" has been undertaken by the developer, to consider the impact of a larger food store, which would attract shoppers from outside the North Northallerton site. This "sensitivity test" has shown that a store of 1739m² could be built without any additional mitigation being needed on the highway network.
- xxxiii. No details for the area have been submitted with this outline application, with all matters reserved. The reserve matters applications will be required to deliver streets where vehicle speeds are naturally restrained to 20mph and walking routes are available on pedestrian desire lines feeding to the formal crossing points of the link road. This will need to include a link on the pedestrian desire line to the A167 Darlington Road.
- xxxiv. The layouts will also need to accommodate manoeuvring and deliveries for service vehicles and customer parking.
- xxxv. There will be a need to link in to the Framework Travel Plan which accompanied this application either through stand-alone Travel Plans for individual uses or through zone Travel Plans covering more than one site.

The LHA recommends the imposition of numerous planning conditions and various Section 106 legal requirements, one of which is the delivery of the link road and bridge within the first phase of the development. If the planning committee resolve to grant approval for the development, the Section 106 Agreement will need to be completed before a planning decision is issued.

With regard to the advice quoted above, it is important to clarify that while the LHA has reviewed traffic movements associated with a proposed supermarket on the site, as will be detailed in Section 6.2 of this report, the applicants have agreed to withdraw that element of the proposal from this application.

With regard to the assessment of Non Motorised Users, no consideration has been given to the suitability of the shared cycleway/footway on the south side of the link road. While planning officers accept that there is only a demonstrated need at the present time for one cycleway on the link road, it is considered that it should be relocated to the north side of the link road as this is where the school and majority of new houses would be located. Therefore, from a usability and safe routes to school perspective, it is more logical for the shared

cycleway/footway to be relocated to the north side. A planning condition should require this change to the link road layout on any planning permission granted.

In addition to the comments from the LHA on transport matters, the Corporate Director of Business and Environmental Services at NYCC suggested during the course of the application that an estimated commuted sum of £968,000 would be required from the developer by NYCC for maintenance liabilities for the bridge embankments and culverts. However, following discussions between the two councils, NYCC confirmed at the time of writing this report that it will now waive the requested sum due to its impact on the viability of the scheme. In addition, the £500,000 contribution sought for bus route pump priming identified in Part xvii of the consultation response provided above is also now waived for the same reason.

A more detailed discussion around developer contributions is included in Sections 6.3 and 6.4 of this report, which should help to explain how/if contributions can be sought.

5.2.2 NYCC Education

The Education Department at NYCC has been a key stakeholder in the preparation of this application by the developer consortium and has had a big influence on the proposed location of the primary school and the size of the school site. The LDF Allocation did not include a school but it was added through the 2011 Framework Masterplan process. The Education Department advise that they anticipate the school accommodating children from September 2018. A reserved matters application will need to be submitted showing detailed design of the school (amongst other things) in due course but for the purpose of this current application their advice was as follows:

The estimated number of pupils from a development of 900 2+ bedroom dwelling houses is 225. There is local shortfall of 217 places, thus there is a need for 217 new school places. The financial contribution sought from the developer is £2,950,332.00.

The Corporate Director of Business and Environmental Services at NYCC has added the following:

The estimated cost of construction is £5,000,000. NYCC will contribute £2,050,000, comprising the following:

- £955,000 NYCC basic need funding to the primary school provision
- £805,000 Section 106 education funding from York Trailers development
- £250,000 Additional NYCC top up funding for primary school provision
- = £2,050,000 Total NYCC education contribution.

This leaves a shortfall of £2,950,000 which is expected as a developer contribution. It is also standard practice that a serviced site is provided at no cost to NYCC. Road access to a serviced school site would be required by early 2017.

The provision of a school is identified as a CIL Regulation 123 infrastructure item, thus while the developer consortium have agreed to provide a serviced site, the cost of the build can only be taken from any available CIL funds. The Council has agreed to provide £2,000,000 of CIL towards the shortfall¹, which leaves £950,000 to be found. As mentioned above, a more detailed discussion around developer contributions is included in Sections 6.3 and 6.4 of this report, which should help to explain how/if contributions can be sought.

5.2.3 NYCC Heritage Services

NYCC Heritage Services has advised the Council that they have reviewed Chapter 9 of the ES, but notes that the conclusions of the chapter are drawn from the evidence base in the form of a Desk Based Assessment. As a result, NYCC recommended that a scheme of archaeological evaluation, which should comprise a geophysical survey, followed by trial trenching should be undertaken prior to determination of the planning application. Nevertheless, subsequent correspondence from NYCC Heritage Services advised that if the Council is minded to approve the application, then it is essential to ensure that the archaeological evaluation work is completed prior to the commencement of the development, and that appropriate mitigation is also agreed in advance and implemented either before

¹ Report to Cabinet 01/09/15: Community Infrastructure Levy – Contributions Towards the North Nothallerton Infrastructure Schemes - http://democracy.hambleton.gov.uk/documents/s2606/090115%20Item%208.pdf

and/or during development, as appropriate to the findings of the evaluation. The suggested planning conditions are noted in Section 7 of this report.

5.2.4 Historic England

Alongside County Heritage Services Historic England were also consulted. They advised that there are no direct physical impacts on designated heritage assets or negative impacts on their setting, that the views of NYCC should be sought regarding archaeological matters. They further advised that they are content to defer any archaeological conditions to NYCC.

5.2.5 Yorkshire Wildlife Trust

Yorkshire Wildlife Trust (YWT) were consulted and expressed some concern regarding the adequacy of the ecological submission. They state in *italics* that:

The Environment Statement covers the impacts, compensation and mitigation for the entire hybrid application, making it hard to assess whether the impacts of Phase 1 have been considered independently. The 'Illustrative Masterplan' map shows the area included in Phase 1 does not contain much of the greenspace and leaves little room for SUDS basins, especially in the eastern section.

The ES identified habitat for breeding birds on the site, including red listed ground nesting birds which will be losing habitat as a result of the development. However, the ES does not contain all of the information needed to fully assess the impacts of the development on breeding and wintering birds (e.g. the observation frequency map), it may be that the impacts are such that compensation land would need to be provided as the risks to ground nesting birds in particular cannot be easily avoided or mitigated for.

The ES highlights the potential impact on roosting and foraging bats, and recommends emergence surveys. If there is going to be removal of any buildings or trees that may house bats as part of the Phase 1 development then an emergence survey should be completed before planning permission for this phase is granted.

We support the conditioning of the ES recommendations for the creation of an ecological enhancement and management plan along with financial provisions for long term management and monitoring. A Construction Environmental Management Plan should also be created which will outline the measures to be taken to minimise the environment impact of construction activities. A detailed Section 106 will be required to ensure that Green Infrastructure and SUDS creation and long term environmental management are successful.

As this is a very large application we would expect the local authority to have professional ecological advice to assess the impacts of the development and the effectiveness of the proposed mitigation.

The Council engaged Arcus Consulting to review the ecological information submitted with the application in light of the recommendation of the YWT and County Ecologist (discussed in Section 5.2.7). Arcus raised concerns regarding the lack of bat emergence surveys and the inadequacy of the breeding birds survey and the developers' ecologist subsequently provided a response. This detail was then referred back to YWT, County Ecology and Natural England. A specific discussion of ecology matters is detailed in Section 6.6 of the report.

5.2.6 Natural England

Natural England (NE) has provided a response under various sub headings. That advice, with planning officer comment, is summarised below.

With regard to *Statutory Nature Conservation Sites* it was concluded by NE that the proposal is unlikely to affect any statutorily protected sites or landscapes.

With regard to *Soils* and *Land Quality* NE reiterate national planning advice that local planning authorities should seek to retain high quality agricultural land and instead use poorer quality land for development. NE also advises on soil handling and retention for the undeveloped parts of the site. However, on the basis that the allocation of this entire site for mixed use development has already undergone the necessary scrutiny and secretary of state examination, and on the basis that the undeveloped parts of the site will largely form the flood catchment areas, their advice on Land Quality and Soil handling has less relevance.

In relation to *Green Infrastructure* NE encourages the applicant to maximise opportunities to incorporate green infrastructure during the development of the detailed proposal. Green Infrastructure includes parks and gardens, informal recreation space, allotments, river

corridors for example. It is considered that the Phase 1 component through its grassland SUDS basins and the green corridor through the Taylor Wimpey Site attempts to provide for biodiversity within the future urban setting, while the assessment of the outline phases will occur at the reserved matters stage, although it must be noted that a large area is to be given over to informal and formal green spaces.

With regard to *Protected Species* NE advise that they have not assessed the application and defer to their Standing Advice. However, further consideration regarding Protected Species has been provided by the Yorkshire Wildlife Trust and NYCC Ecologist and is discussed elsewhere in Section 5.2 and in Section 6.6 of this report.

NE identify that the documents submitted demonstrate potential existence of *Priority Habitat* on the site and refer the Council to NPPF recommendations which state that LPAs should aim to conserve and enhance biodiversity and if significant harm from the development cannot be avoided then planning permission should be refused. As will be discussed in Section 6.6 the Phase 1 Habitat Survey is deemed to be satisfactory and mitigation can be provided. In addition, the draft landscape and SUDS proposals identify opportunities for biodiversity within the design. Furthermore, planning conditions could require the submission and implementation of habitat creation and management plans.

5.2.7 NYCC Ecology

NYCC does not have an agreement in place with HDC to provide ecology advice but they did provide some preliminary advice that "given the strategic importance of this site and the potential presence of legally protected species, HDC should seek its own independent ecology advice to assess the application's ecological reports." This was also recommended by the Yorkshire Wildlife Trust and was undertaken, as noted in Section 5.2.5. The discussion on ecology matters occurs later in this report in Section 6.6.

In any event, NYCC Ecology confirmed that "the land is not covered by any designations such as SSSI or SINC and that the value of the land in ecological terms is likely to be local. The habitats to be lost will be of farmland character – hedgerows, trees (possibly including mature), pasture, watercourses and ponds which themselves could support protected species such as bats, farmland birds, hedgerow birds, water vole and badger. In order to accord with the various legislation that surrounds protected species it will be necessary to have relevant surveys undertaken to assess the impact of the scheme and ensure that sufficient avoidance, mitigation, compensation and enhancement measures are included within the design."

Furthermore, NYCC Ecology state that "one of the most damaging impacts of large developments upon habitats and species is that of fragmentation along with on-going disturbance relating to increased human presence. There will be a need to provide sufficient green space dedicated to wildlife that will not be as affected by humans and their pets. These wildlife areas should try to ensure continuity with habitats outside of the development site."

NYCC Ecology provided a further response in light of the independent review of the application's ecology reports by Arcus Consulting, which is discussed in Section 6.6 of this report.

5.2.8 Environment Agency

The EA raised no objections to the application subject to conditions restricting surface water run off, provision of compensatory flood storage, and provision of box culverts in the embankment of the link road. The conditions are repeated in full in Section 7 of this report. Those conditions would need to be discharged either before the link road commences or before any dwelling is occupied.

5.2.9 Yorkshire Water

Yorkshire Water has advised that the local waste water infrastructure (notably the sewer system, as opposed to the Waste Water Treatment Works) does not currently have capacity to remove and treat foul water from the development beyond Phase 1. They therefore advise that reinforcement will be required to avoid detriment to existing sewerage in the vicinity. As no outline drainage details have been provided beyond Phase 1, Yorkshire Water propose a planning condition that no further development beyond Phase 1 should take place until approved sewer reinforcement works have been approved.

With regards to surface water, Yorkshire Water confirm that the public sewer network does not have any capacity available to accept any discharge of surface water from the proposal but it supports the approach to drain surface water into SUDS/watercourse. It recommends a planning condition detailing surface water management processes that must be achieved before the development can commence. Yorkshire Water also confirms that it can adopt the maintenance of SUDS as long as they are satisfied with the design.

Finally, Yorkshire Water have identified a 175 mm diameter public rising main and a 300 mm diameter Yorkshire Water maintained overflow on the Statutory Sewer Map that crosses the site. They advise that the road layout submitted on drawing 12003/GA/11 (Rev F) dated 27/05/2015 and prepared by Optima is not acceptable as the drawing shows the level of the proposed road to be raised by approximately 10 m over the line of the sewer. The additional overburden could cause failure of the sewer. Yorkshire Water objects to the layout of this element of the development and therefore suggest that the developer should amend the site layout or provide evidence that the sewer can be protected via an appropriate condition on a planning permission. One of the planning conditions suggested by Yorkshire Water states that construction of the access road shall not commence until the agreed sewer protection measures have been implemented.

The developer was asked to respond to the concerns of Yorkshire Water and their response was provided by iD Civils on 18th September 2015. It states that the location of the main has been identified and clashes with the western bridge abutment foundation. The bridge abutment cannot be moved due to the impact in the span and bridge deck, therefore the main would have to be diverted. Diversions of public sewers can be secured by the developers under a Section 185 agreement with Yorkshire Water, however, there is potential for this process to take up to 12 months to conclude. The recommended planning condition to prevent construction of the road until this diversion issue is resolved would, however, result in unacceptable delays to a key piece of the town's infrastructure. Yorkshire Water has therefore agreed that the condition could be revised to prevent only construction of the bridge until the sewer diversion is undertaken. If the

The suggested planning conditions are noted in Section 7 of this report.

5.2.10 NYCC Flood Risk Management

The NYCC Sustainable Urban Drainage Systems Officer confirmed that he agreed with the findings of the Environment Agency and the Internal Drainage Board and notes that raising land levels on the east of the site in an area where surface water flooding occurs appears to result in the loss of flood storage. In this regard it has been suggested that there is a danger of off site flooding from water displacement. The SUDS Officer therefore states that the developer is to ensure that any rainfall intercepted by the new development is only discharged at greenfield run off rates and that suitable exceedance flow paths exist for flow generated by extreme events. The SUDS Officer expects this to be dealt with by planning condition to be included on any permission granted seeking submission and implementation of a management and maintenance plan for surface water drainage for the site. The suggested planning condition is noted in Section 7 of this report.

5.2.11 Internal Drainage Board (Swale and Ure)

The Internal Drainage Board (IDB) have advised that watercourses passing through the site have the status of main river but the development is sited largely in the Board's district. The IDB advises that it has reservations about the proposal and suggest that the assessment of flood risk to the site is not based on reliable modelling with differences of up to 1 metre evident in the evidence submitted. The IDB considers that the Flood Risk Assessment (FRA) deals with the issue of loss of flood plain from construction of the new link road but does not make allowance for a similar loss as a result of raising ground levels and building footprints. It is suggested that the FRA places great reliance on the road gully network to remove surface water but gully systems are only designed to a 1 in 2 year standard and in flood conditions soon become partially or totally blocked.

As a result the IDB predicts significant surface water flooding would affect the area to the east of Northallerton Road. The raising of ground levels is considered by the IDB to merely pass flood water elsewhere within the site or increase flows through existing exceedance flood routes potentially affecting off site property. Consequently the IDB has advised that no development should commence until they have seen and approved a robust assessment of flow paths in exceedance conditions to ensure that the effect that both surface water and fluvial flooding might have on the completed development and off site areas can be addressed. The suggested planning condition is noted in Section 7 of this report.

5.2.12 Brompton Flood Group/Council's Drainage Officer

The Brompton Flood Group has objected to the application as they believe it would be detrimental to the whole of Northallerton, Brompton and Romanby and for other communities downstream by flooding. The Flood Group's objection is available in full on Public Access.It was circulated to the applicants who prepared a response, which has been subsequently reviewed by the Council's flood and drainage officer, the County SUDS officer and the IDB.

The Council's Drainage Officer's response on a point by point basis to the Brompton Flood Group's objection is included verbatim in *italics* below:

1. The main principles of development in relation to flood risk are that the development does not increase flood risk elsewhere and that the development itself is not subject to flood risk up to the design standard and further the consequences over and above the design standard are assessed and managed to minimise risk of flooding to the existing and proposed developments.

The proposed development is located outside of the flood zone 2 and 3 extents and the surface water flows within the development are being managed within the design standard. Highway infrastructure is unavoidably to be constructed in the flood zone 3 extents, this will occupy existing flood extents, compensatory flood storage will be provided as part of the development works in the immediate vicinity for this loss of flood storage.

The surface water discharge from the proposed development will be at greenfield rate or potentially less than greenfield rate as determined by Institute of Hydrology 124, so the flows to watercourse from the developed land will mimic the existing flows in quantity and flow rate from the undeveloped land.

The overall effect is that there will be no detriment to Brompton in flood risk terms due to the proposed development, or flood risk to the development itself within the designed standards.

- 2. The riparian owners are the land owners that abut the watercourse and have responsibility. Within the development area the developers as land owners will be riparian owners, this may change as houses are purchased and open spaces are transferred to their respective owners. The Environment Agency, North Yorkshire County Council and Drainage Board have powers in relation to the watercourses and a limited range of duties, riparian owners may be required to seek consent to work in watercourses.
- Maintenance of watercourses is undertaken in agreement between the riparian owners and authorities, EA, NYCC and DB with powers in relation to the watercourse, this will continue during the proposed development. Flows to watercourse as outlined in point 1 above will be unchanged.
- 4. The timing of surface water management measures is to ensure that there is no period of additional flood risk created during the development phase.
 - Implementation of surface water management measures is acceptable on a phase by phase basis, but constructed in advance to comply with the no additional risk caveat.
- 5. Development will include some raising of levels; the extent of this is being agreed in conjunction with the Environment Agency and ourselves. The raising of levels in flood risk terms is not to be to the detriment of existing dwellings or the development itself.
- 6. Design standards have been established for surface water management and run-off.
- Management of surface water run-off during the construction will be conditioned and/or included within the Site Waste Management Plan and Construction Phase plan.
- 8. The future maintenance of SuDS will be included with the planning conditions.
- Future management responsibilities for Public Open Spaces, SuDS and wetlands will be agreed as part of the planning approval. Management practice will be the responsibility of the individual maintaining organisations taking into account, statutory requirements and best practice guidance.
- 10. The development plans do not include for fundamental changes to watercourse, i.e. bed levels or route. The accommodation of the construction of the highway bridge is on the basis that it is not of detriment in flood risk terms to existing development.

11. There is no allowance within the North Northallerton Development Area scheme for flood risk management measures as described. The principles of the flood risk and new development are outlined in point 1, with the design of the development based on no detriment to existing development and managed flood risk on the development itself, to the design standard.

5.2.13 Network Rail

Network Rail are a key stakeholder in this application and have been in background discussions with the developer consortium and County Highways over this development proposal. Notwithstanding the fact that the developer would need to enter into various legal agreements with Network Rail and obtain relevant consents from Network Rail, they were consulted on the planning application. Their advice is that they have no objection in principle but that the following matters need to be taken into consideration:

Railway Bridge – Note is taken of the plan showing the general arrangement for the bridge over the railway line, and this is in line with the discussions held with our asset protection team. Given that the detailed design has to be agreed with NR in any event, we are happy for this to be taken forward by means of an appropriate condition, which would allow for any minor changes in design to the structure as a result of (for example) detailed ground investigation. It is very important, however, that the Basic Asset Protection Agreement to develop the design proposals is signed as soon as possible so as to prevent delay in the implementation process. We are also of the position that the construction of the new bridge has to be tied in with the closure of the user worked crossing at Halfway House (LEN3 43 Miles 1485 Yards, NGR 436821/495430). Closure of this crossing should be linked to the opening of the bridge in so far as connecting routes either side of the bridge should be made available for the crossing user at the same time as the bridge is opened for through traffic.

Drainage – There are no immediate issues arising from the preliminary drainage strategy that give us cause for concern. The discharge from catchment 8 pond may give rise to scour protection issues, and it may be that protection needs to be applied to the beck where it passes under the railway. We would expect this issue to be discussed in the detailed drainage strategy when produced as presumably required by an appropriate condition.

Fencing – It is unclear from the proposals whether unrestricted access to the land designated as "The Meander at Hawthorne" is envisaged as part of the development. If it is to be used as public open space then there will be a requirement for trespass-proof fencing to be erected on the western side of the railway, from the new road bridge to the site boundary. This can be removed if suitable fencing is in place along the eastern boundary of the housing development to prevent access onto the meadow, or reduced in length if the existing bridge over Hawthorne Beck (used by the farmer to gain access to his fields via the level crossing) is removed.

Northallerton Station – The station is generally well provided in terms of existing facilities, with the exception of disabled access to platform. However, this is being addressed by the DfT Access for All Scheme, scheduled for implementation by 2019. However, given the emphasis of the development (via the travel plan) for linkages to the station, and in particular by cycle, we consider it appropriate that the development should fund additional cycle parking at the station through the addition of a covered 5-stand for cycles – around £10k – the be provided upon occupation of 75% of the dwellings in Phase 2 of the development.

The request for developer contributions will form part of the negotiations with the developer, as discussed in Sections 6.3 and 6.4 of this report.

5.2.14 Hambleton DC Environmental Health

The application was specifically referred to Environmental Health for an assessment of noise through the construction period and the residential occupation period from key noise generators such as the link road and the industrial units on Thurston Road, but primarily from Allerton Steel. During the course of the application, Allerton Steel provided their own acoustic assessment, the applicants provided a subsequent response, and all of this material has been reviewed by an Environmental Health Officer (EHO). A discussion on noise will be undertaken in Section 6.7 of this report.

In summary though, the EHO confirms that to the north of the proposed link road the primary noise generator would be the road itself, while to the south of the road, noise from both the road and Allerton Steel would be dominant. This would necessitate different mitigation measures to maintain a reasonable level of amenity for proposed uses across the site, although as so much of the proposal is in outline, further acoustic testing will be required at reserved matters stages before the suitability of any of the uses or their actual location on the site can be confirmed.

5.2.15 HDC Housing Officer

The Housing Officer was involved in extensive pre-application discussions with the applicants up until the point of submission of the application to try to influence the level, location and size of the market, affordable and specialist housing provision on the site. However, the Officer has not been involved in the viability appraisal process (which is discussed in Section 6.4 below) which determines the extent of affordable housing that can actually be provided on the site. The Housing Officer's comments are therefore premised on the Council's desirable requirements and the layouts of the first phase of development, as submitted at the time of the application. The Housing Officer's comments, are produced verbatim in *italics* below.

Market Housing Mix and Older Person's Housing

Lifestyle changes have led to the formation of smaller households in Hambleton and this has also impacted on the type of housing that is needed to sustain communities and support economic growth.

To meet the needs of the changing population the number of new two and three bedroom market homes need to be increased. This will increase housing options for smaller families and couples for whom four and five bedroom houses are too large and unaffordable.

The Council is also concerned to ensure that all housing better meets the needs of the population in the light of demographic and lifestyle changes. Census data reveals that the population is ageing and this is increasing year on year. Lifestyle changes have also led to the formation of smaller households and this has also impacted on the type of housing that is needed to sustain communities and support economic growth.

There is evidence to support the following market mix on larger market sites:

10% two bedroom bungalows, 10% one bedroom & 60% two & three bedroom homes (predominantly two bedroom).

In towns such as Northallerton, which are key employment centres, there is also a need for a greater proportion of one bedroom homes for younger households.

In order to improve Hambleton's 'Housing Offer' in September 2015 the Council adopted a Size, Type and Tenure of New Homes SPD, which seeks to influence the size and type of new homes that are being built across the District. This SPD includes an indicative housing mix target for all sites of 25 or more dwellings and seeks 10% bungalow provision on all housing sites of 10 or more units.

Extra Care

Early discussions suggested that development of part of the site for an Extra Care scheme might be appropriate. Following further discussions between the Council, North Yorkshire County Council and Broadacres partners are now of the view that there is probably a greater need for a mixed age multi-tenure scheme not only to meet the needs of older people but also to help support those with learning disabilities to live more independently. This proposal requires more detailed consideration and would be progressed in a later phase.

Space Standards

The Size, Type and Tenure of New Homes SPD is also a tool to help deliver good quality homes across all tenures. This SPD signposts developers to the Nationally Described Space Standards which were issued by DCLG in March 2015 and which are applicable to all tenures. The Council advises that it will use these standards to guide the provision of good quality new homes with the intention of embedding them within the forthcoming local plan.

Housing Mix- Phase 1

The developers of Phase 1 are proposing the following mix:

Туре	Number of original scheme	%age of original scheme (rounded up)	Number of amended scheme	%age of amended scheme (rounded up)	SPD target mix
1 bed homes	0	0%	0	0%	10%
2 bed bungalows	13	4%	13	4%	10%
2 bed houses	40	13%	40	13%	35%
3 bed houses	145	49%	136	47%	25%
4 bed houses	94	32%	96	33%	10-15%
5 bed houses	6	2%	6	2%	No target
Total	298		291		

It is disappointing that no one bedroom properties are being proposed in this scheme as this would improve the housing offer for young economically active singles and couples working in the town.

The offer of some small 2 bed bungalows is welcomed. However, this offer falls short of the 10% target.

Whilst the current proposal offers a mix of 64% 2 & 3 bedroom homes (including bungalows) there is concern that there is a predominance of 3 bedroom properties. The mix could be improved by inclusion of a greater proportion of 2 bedroom and some one bedroom homes.

The proposed mix also includes a substantial proportion of 4 bedroom homes which are unlikely to provide an affordable option for local buyers.

There is concern that not all the house types meet the Nationally Described Space Standards.

However, the need to assess the target mix in the light of viability issues is recognised. It is also acknowledged that the Council's Size, Type and Tenure SPD has only just been adopted and there may be the opportunity to redress some of the issues regarding housing mix through negotiations on later phases of development.

However, viability aside, a scheme with an improved mix in phase 1 is likely to provide a greater level of social inclusion and community cohesion.

Affordable Housing

The Council adopted a new Affordable Housing Supplementary Planning Document in April 2015. This builds on the previous SPD (2008) and expands Policy DP15 providing more detail and clarity around the Council's requirements for affordable housing delivered through the planning system to ensure it is of good quality. The SPD addresses issues around size standards, transfer prices and provides guidance on tenure mix and the distribution of the affordable homes. It does not change existing policy which will be reviewed as part of work on a new local plan.

This application site lies within Northallerton where the Council's affordable housing target is 40% unless there is evidence provided through an Economic Viability Appraisal (EVA) submitted by the Applicant and reviewed by the Council's Consultant Valuer to demonstrate that this is not achievable.

The full planning application for Phase 1 is for 291 dwellings -150 dwellings to the East of Darlington Road (Persimmon Homes) and 141 to the west of Stokesley Road (Taylor Wimpey). To achieve the 40% target on Phase 1 a total of 116 homes would be required to be affordable, of which the Council would seek a tenure spit of 70% social rent (81 homes) and 30% (35 homes) intermediate tenure unless there was evidence to support a different tenure split which was also supported in writing by the Registered Provider to whom the homes would be transferred to. This tenure split is different to the 50/50 tenure split contained within Policy DP15 but has been applied by the Council in its negotiations in response to evidence provided by the 2011 North Yorkshire Strategic Housing Market Assessment and feedback/intelligence from Registered Providers working across The affordable homes should be a mix of one, two and three bedroom to meet the needs of singles, couples and families.

To meet the Council's affordable housing requirements the homes must be of a size that meets the Council's minimum standard or from the 1st October 2015 the National Described Space Standards, pepper-potted throughout the site preferably in clusters of no more than six to eight dwellings and the developer must transfer the properties to a Registered Provider at the Council's agreed Transfer Price. Transfer prices were reviewed as part of the consultation on the SPD. These are tenure neutral and designed to support a tenure mix on most sites of 70% social rented and 30% intermediate tenure.

			the table below:

Unit type	Minimum size	Transfer Price
1 bed	50m2	£50,200
2 bed	70m2	£65,200
3 bed	90m2	£79,200
4 bed	110m2	£84,200

The Council is also keen to ensure that affordable homes are of good quality, offering a good level of residential amenity to future occupiers and are energy efficient and affordable to run to those on lower incomes.

Guidance on issues such as the parking arrangements and boundary treatments for the affordable homes is also included within the SPD to minimize management issues in the future.

The planning application should be accompanied by a schedule of the affordable homes confirming plot numbers, property types/sizes/tenures and transfer prices as well as floorplans of the affordable homes indicating the size/dimensions of rooms and showing furniture layouts.

Conclusion

The scope to improve the proposed housing mix, by reducing the number of three bedroom homes, increasing the number of two bedroom homes (including bungalows) and providing some one bedroom properties should be considered.

The developers are urged to speak to the Council's Housing and Planning Policy Manager following the outcome of the development appraisal work to agree the level of affordable housing provision and its type and location which will be secured via a S106 agreement.

5.2.16 Police Architectural Liaison Officer

The PALO has provided commentary on design features of the development, notably fence heights, landscaping, natural surveillance and security. It is recommended that the residential development is designed in accordance with Building for Life 12 to provide residents with a safe and secure environment to live and to reduce the opportunities for crime and anti-social behaviour to occur. An assessment of design of the 'Phase 1 application will be undertaken in Section 6.8 of this report.

5.2.17 Ministry of Defence

The MOD have confirmed that it has no safeguarding objections to the proposal.

5.2.18 North Yorkshire Local Access Forum

The Local Access Forum have submitted three responses primarily raising concerns with the lack of provision for cycling on both sides of the link road and the lack of dedicated or shared off-road provision for horse riders. This matter has been discussed by NYCC Highways in Section 5.2.1 of this report and will be referred to again in Section 6.3.

5.2.19 National Cycling Charity

The National Cycling Charity were not actively consulted on this application but they have in any event provided comment that neither supports or objects to the proposal. They suggest that the shared cycleway on the link road should be on the north side instead of the south such that primary school children can use it; all children should be able to cycle to school safely through the development with roads line marked accordingly, and that there should be off road cycle ways through the green spaces designed as leisure routes.

5.2.20 NYCC Public Rights of Way

The NYCC PROW team have not raised an objection to the planning application but have included a planning condition about preventing obstruction to the PROW adjacent to the development. Full detail of the condition can be found in Section 7.

5.2.21 Ramblers Association

The Ramblers advised that they have no objection to this proposal and consider that the proposed footpath and cycleway are designed to align in purpose with existing.

5.2.22 Sport England

Sport England have provided a summary of the needs of the local clubs but recognise that the sports village element is in outline at this stage such that further consultation will occur in the future when reserved matters applications are made. However, they note that there is demand for additional pitches especially football and also the potential for rugby. However, they suggest that new facilities should be informed by robust evidence base in the form of a Playing Pitch Strategy, although in lieu of a Playing Pitch Strategy the views of the National Governing Bodies for Sport should be taken into account at the appropriate stage. Sport England would welcome the opportunity to enter into discussions with the applicant prior to the submission of any reserved matters application to ensure that the right facilities are proposed. In the meantime they do not raise an objection to the outline application.

5.2.23 Planning Casework Unit

The National Planning Casework Unit acknowledged receipt of the Environmental Statement and advised that they had no comments to make.

5.2.24 Hambleton, Whitby and Richmondshire Clinical Commissioning Group

The Clinical Commissioning Group has been consulted on this application and they advised that "there was no identified need for a new surgery in the area; however, they request financial support from the developer to help meet additional health care costs."

It must be highlighted that the provision of either new surgeries or improvements to existing facilities are classified as infrastructure, and since the introduction of CIL, the provision of such facilities would need to be identified on the CIL Regulation 123 List. At present no healthcare infrastructure is included on the CIL 123 list, although the provision of additional GPs is identified as a potential project. The allocation of CIL funds to the CCG is, however, a decision for the Council to take when it collects and comes to spend CIL and just because a project is included on the 123 list does not signify a commitment from the Council to fund, either in whole or in part, the listed project.

As the Council is currently in the process of undertaking a Local Plan review, it is the most suitable time for the CCG to engage with the Council to identify future need based on anticipated population and house growth in Hambleton and thus future CIL 123 priorities.

5.2.25 Allertonshire Civic Society

The Civic Society has provided mixed feedback regarding the application outlining that they have no objection to the level of housing, education, leisure and retail facilities as they meet local needs, while they are happy with the retention of open space between Northallerton and Brompton. However, they consider that the design of the link road is inadequate and consider that the bridge should be built at the outset.

5.2.26 Allertonshire School

Northallerton School and Sixth Form College has advised in writing that they have no objection to the application.

5.2.27 Brompton Town Council

The Town Council states that if its concerns cannot be addressed then it objects to the planning application and recommends that a refusal be issued. Their response is provided in full verbatim in *italics* in this section due to the number of independent points being raised. Officer comment will be provided on some of the points but expanded in Section 6 of the report.

The [Town] Council is of the strongest view that the proposed primary school cannot be sited so close to the industrial sites on Thurston Road, especially the heavy industrial works at Allerton Industries; noise from which can frequently be heard in Brompton. Noise is not good for the children either in class or at play. It is quite remarkable that someone has deemed it appropriate to site a new school in this location.

The County Council is responsible for school provision in Northallerton and has been part of the steering group for the development of this site for a number of years. The selection of a central site for the school was made on the basis that it would form a hub of activity within the residential allocation, could be easily reached on foot and by car and by being in the middle of the residential development would reduce journey times for the majority of families that would have a school place there. Furthermore, while some industrial noise may be audible from the school premises, Council's Environmental Health Officer considers that the road is the primary noise generator and that this can be mitigated through design and layout of the school buildings/outdoor recreation space on site.

In this respect it is understood that the District Council requested Allerton Industries to move to its current site from its former premises in Romanby Road – i.e. to be away from residential properties and the nearby primary school, etc. Why therefore was something that was deemed to be inappropriate at that time now considered to be acceptable?

Allerton Steel, as a neighbouring landowner was consulted during the land allocations process in 2008 and there was extensive public consultation around the development of the Masterplan in 2011. No objections were received from Allerton Steel to either of those consultation processes. The adopted Masterplan depicts land to the north of the existing Allerton Steel premises for employment uses, which could be used for their potential expansion. However, this is subject to land purchase by Allerton Steel from the relevant landowner and planning permission, which would also require noise mitigation measures as part of any expansion proposal.

In any event, the noise reports submitted by the Applicants and Allerton Steel demonstrate that while Allerton Steel can generate high levels of noise throughout the day and night, the buildings to the north of the link road can be constructed with noise attenuation materials to comply with relevant environmental health guidelines, and in the longer term when the land to the south of the link road and behind Allerton Steel is developed a further noise attenuation buffer will exist to bring noise levels at those sensitive receptors well within acceptable levels.

In addition as regards school provision, what provision, if any, has been made for the associated increased secondary school places as there seems to be an assumption that from this development and from the new Castlegate development families will only have children of primary school age – on what basis has this assumption been made and what will happen when these children get to secondary school age?

County Education Department are responsible for determining school place requirements. They have not identified a need for additional secondary school places and the District Council cannot therefore require provision to be made. The District Council is currently in the process of reviewing its local plan and future land allocations, and it might be identified through this process if, where and when a further secondary school will be required; however, there is currently no need identified.

Likewise, the Council also strongly objects to any dwellings earmarked for the elderly to be sited anywhere within the vicinity of such industries. If elderly residents are to live in the area then they would be constantly in fear of the noise, etc. from the factory as would be the children.

With regard to the location of aged care housing, this is depicted on the adopted Masterplan to the west of Allerton Steel. Based on existing noise readings for the company, noise levels for that land would exceed acceptable standards for sensitive residential land uses; however, there are a number of noise attenuation measures that could be applied, such as through building materials of the aged care facilities or acoustic fencing, that could mitigate the impact. As the aged care use is currently only being applied for in outline, with all matters reserved for subsequent approval, noise and mitigation will be revisited at the appropriate time. The Council will not approve buildings if noise standards cannot be brought within acceptable levels.

Greater consideration needs to be given to those children who walk or bike to school from Brompton to the Allertonshire School and their safety. Especially maintaining the fact that they do not have to cross any roads; as when the bridge and roundabout are built this could mean heavy goods vehicles in the vicinity and increased traffic. The council would wish to see a footbridge or underpass being constructed to enable pedestrians and especially school children to access the Allertonshire from Brompton safely. The council is not in favour of maintaining a pelican crossing system as cars don't always stop at such crossings.

It must be identified that neither the allocations document nor Masterplan for this site specified the provision of an underpass. The construction of an underpass would have implications for controlling flood water given the proximity of flood zones 2 and 3, it would require extensive land take to provide the necessary embankment support, and would require an extensive capital payment from the developer (as would the construction of a footbridge) which would impact on the viability of the scheme.

The current proposal, shown on Plan No. 12003/GA/14 Rev A, prepared by Optima and provided at Appendix H of the Transport Assessment, is the provision of a central refuge island crossing facility on the link road at the Northallerton Road junction, which means that pedestrians can safely cross the road in two stages if required; this provision is based on Optima Highway's survey of existing non motorised users along this existing route, national guidance, and an assessment of the predicted number of pedestrians and vehicles. However, the Movement Framework provided on Page 48 of the Design and Access Statement identifies this location as a potential Toucan crossing point. NYCC Highways have been asked to provide advice and they have suggested that either option can be satisfactorily accommodated, but an updated road safety audit, and non motorised users audit will be required before the road is built, which will determine the actual requirement.

As regards the 30mph speed limit, will there be a 20 limit for all of the schools in this vicinity? Surely this is an opportunity not to be missed; also who will properly police the speed limits?

The speed of the link road would be 30mph for its entire length to provide favourable journey times around the town and thus making it a viable alternative route. While the school land would abut the link road, the school building and access would be provided to the north of that site and not directly off the link road and would be subject to a 20mph limit on the residential roads adjacent to the school. With regards to speed limits near other schools, this is a County Highways matter and not something that can be addressed through this application.

The council would wish to see the current 30mph speed limit extended further along Stokesley Road especially in view of the anticipated increase in traffic volumes in the vicinity.

The 30mph speed limit is to be extended northwards on Stokesley Road and is identified on the submitted highway plans. The responsibility for policing speed lies with the Police and has no bearing on this application.

As regards traffic matters that will affect Brompton, the council requests that Bowe Bridge, Station Road be made single traffic to cope with possible rat run from Northallerton Road towards Darlington Road.

This is a matter for County Highways to consider in the future if evidence suggests that such a measure is necessary once the development is complete.

There is also a possible safety issue here with regards the school with any increase in traffic levels since Brompton Primary School for some unexplained reason has continually been refused a 20mph speed limit and indeed is one of the few schools on

such a busy highway that does not have this safety benefit for its children in North Yorkshire. The time is now opportune to address this issue on safety grounds.

See comments above.

Due to accidents, several times this year the A19 has been blocked and all traffic has been diverted along Stokesley Road which has been a horrendous situation for Brompton and this will not be improved by the link road – Brompton has suffered with vehicles trying to make a quicker journey by travelling through the village only to cause more chaos.

When accidents occur they can make journey times on other roads difficult. However, unless there are evidenced based accident hotspots, it is difficult to account for/mitigate unpredictable events on any road in or around Northallerton/Brompton. This is therefore not a valid reason for rejecting the application.

When the Plan was being discussed the road from Brompton to Northallerton via the cycle path was sold to many people on the understanding that an underpass would be in place. However, this is now not to be the case instead we are to have a pelican crossing – due to the financial cost – the council asks what price do we put on a person/child's life? You yourself at the recent meeting of Brompton Council said that things could change with regard to the application – consequently let one of the changes be the provision of an underpass.

See earlier comments.

If the development is to proceed the Council would ask that all footpaths be provided with lowered areas for those in wheelchairs and guardians who use buggies.

This is a standard requirement of the Manual for Streets and will be insisted upon by the County Highways Department through their road adoption process.

A far more basic question must be the proof of need for this development at a time when other properties on nearby new developments do not appear to be selling. There would seem to be no evidence of need for the properties intended to be incorporated into this development. There is a great need for rented accommodation in Northallerton, as well as affordable housing, especially for older people and those wanting to down size because of the bedroom tax. There is no certainty whatsoever that this development will resolve these issues.

A discussion of housing need will be provided in Section 6.1 of this report.

What would be the effect on house prices and the demographics of the population of Brompton?

With specific regard to property values this is not, in itself, a planning consideration, for there is no provision within the Town and Country Planning Act or Adopted Development Plan which requires the Council to consider the impact on property values and there is often no specific evidence that supports the premise that a new residential development would decrease the value of other residential properties nearby or would notably alter demographics of Brompton to the extent that special measures would be required.

The council will leave the Brompton Flood Group to comment in detail on the concerns as regards future risk of flooding and adequate drainage provision. However an assurance is requested that the proposed SUDS will be of sufficient overcapacity to cope with flooding and are designed as such by Hambleton District Council.

There is a question of who owns the responsibility for the maintenance after the initial period has stopped and who is paying for it? Brompton Council strictly requires that it be given cast iron guarantees that the changes brought about by this or any other new development will not cause a backflow of water towards Water End Green or any other locations within the Parish of Brompton. Furthermore, as flooding and the effects upon the village must be a prime consideration these need to be rigorously addressed before work starts rather than creating a problem and then looking for a solution.

It is also essential that all improvements be put in place for flooding before work on the 150 houses is started since there is no guarantee that the remaining 650 will ever be built.

The Council therefore looks for improvement of the beck to deal with water flow also throughout its entirety including within Brompton Parish. Such improvements to include the protection of Church View, Brompton by extending the flood wall from Linen Way to Bowe Bridge in a similar manner as to how Brompton Community Primary School has been provided with a flood wall.

Is there any possibility of farmers grants being linked in the future to help with slowing run off water from their farms to protect Brompton, Northallerton and Romanby?

Finally in connection with flooding, the council would expect sympathetic materials to be used for all drives and hard standings at each property.

Hambleton District Council are not responsible for the design of water storage/drainage measures and have referred the application to the internal Drainage Board, the Environment Agency, Yorkshire Water and County Flood Protection team. Their advice is detailed within this report and any design revisions/measures that they request by planning condition will be included on any permission granted. Any off site improvements suggested by the Parish Council that are unrelated to the proposed development cannot be sought from this permission. Paragraph 204 of the NPPF states that planning obligations should only be sought from developers where they meet all of the following tests: a) necessary to make the development acceptable in planning terms; b) directly related to the development; and c) fairly and reasonable related in scale and kind to the development. With regard to maintenance of SUDS, Yorkshire Water advise that, as long as the basins/tanks, etc. are designed to their satisfaction that they will adopt and maintain them.

It is also suggested that there is a need for more local provision of healthcare as there are currently two doctors surgeries with long waiting times to get appointments and a hospital that seems to become less and less capable of providing a satisfactory level of service.

The Clinical Commissioning Group has been consulted on this application and they advised that there was no identified need for a new surgery in the area.

5.2.28 Northallerton Town Council

The Town Council states that it objects to the planning application and recommends that it is refused. Their response is provided in full in *italics* with officer comments following, however some points will be expanded in Section 6.

The Council has serious concerns about any form of development constructed on existing flood plains particularly when such development is of a high volume residential proposal as in this case. In recent times very severe flooding has occurred along the beck that passes in the vicinity of this proposed development and at its location throughout Brompton village. There are no guarantees that the same level of flooding will not similarly occur at this location once the construction on the flood plain has taken place. More importantly it is unclear as to how residents in Brompton and North Northallerton would be protected to minimise the increased risk to their residential properties.

The only development on the floodplain is the link road and bridge and the Environment Agency have required compensatory flood storage to be provided. They have requested submission and implementation of this detail by Condition and no works will commence until the EA are satisfied that the flood water can be captured safely elsewhere. Various other conditions relating to water run-off have also been suggested by the relevant flood management authorities and such detail will have to be approved by those authorities before any development may commence. As Hambleton District Council is not a flood management authority it relies upon the expertise of those other agencies to ensure that the scheme does not create off site flood risks.

It is also unclear as to how it is intended to treat foul and surface water from this new development at a time when it is understood that Romanby Sewage works is at 92% of its capacity and when the Yafforth Road developments and development at the former Milk Board site on Romanby Road are also taking place and will need to be connected to the Sewage Works.

Yorkshire Water is the statutory authority for foul water and advise that there is capacity for Phase 1 of the development but not sufficient capacity for the later phases. However, they are under a statutory duty to ensure that there will be sufficient capacity and have to work with the

developers to plan the timing of foul water upgrade works. Yorkshire Water confirm that the capacity issue is not with the Waste Water Treatment Works but with the sewer network. With regards to surface water, various measures such as detention ponds and underground SUDS basins are proposed and such measures are supported by the various water management authorities, subject to surface water run-off rates being restricted to greenfield rates, which is recommended be required by planning condition. Section 6.5 will provide further detail on surface water issues.

Of particular concern to the Township of Northallerton is the issue of the Low Gates Level Crossing; and, whilst your Council has stated that it is not responsible for the rail network there is however an over-riding detrimental effect on the town enhanced by this proposed development and the Local Authorities do of course have some input into the road infrastructure especially when related to the siting and size of new developments. As a consequence it is inappropriate for the District Council to try to distance itself from this overriding aspect of the application.

As has been well documented, the town becomes gridlocked frequently throughout the daytime and early evening; this is severely worsened when traffic may have to be diverted along this route from planned or unexpected closures on the A1 or the A19. Any proposed link road and bridge will in the main of course only affect the new development and will do nothing to decrease the traffic flows within Northallerton; it being estimated the link road and bridge will only alleviate 3% of the traffic flows at this point – thereby being at a considerable financial cost to achieve very little.

In addition the proposed increase in rail traffic over coming yeas seems to have been completely disregarded.

The very future of Northallerton's economic future and prosperity in essence revolves around a solution being reached with regard to the traffic issues in the town.

The issue of Low Gates is a problem which the Council is well aware of but this application will not make the impact of Low Gates worse. In fact part of the purpose of the proposed road is to provide an alternative route to Low Gates for some travellers as well as enabling the mixed use development to proceed. This will allow necessary housing to be built and bring some relief (although not a solution) to traffic movements to the town. It would not be acceptable to refuse development on allocated sites based on existing constraints on the town when the proposal itself is also accompanied by a measure to mitigate the impact of that development, which the link road does. In addition, the Council cannot require the developer to resolve an existing problem in the local area if the proposed development is not exacerbating the problem; the Government is quite clear that infrastructure requirements must be identified through a CIL 123 Priority List and that additional off-site developer obligations (for example through Section 106 Agreement) can only be sought if the obligation is directly related to the development.

The Council will continue to liaise with Network Rail and County Highways over Low Gates but it does not consider that the problem of Low Gates is sufficient to refuse the proposed application. The most significant impact on Low Gates level crossing is the amount of time the gates are lowered, which is entirely dictated by rail traffic, not road traffic.

With regard to the suggestion that the link road and bridge would only alleviate 3% of traffic flows, it must be noted that that is factually incorrect. Optima Highways has spent three years working with HDC, NYCC and their consultants Jacobs to update and validate the Strategic Northallerton Town Centre VISUM model (which is the world's leading software for traffic analyses, forecasts and GIS based data management) to fully test the impact of the North Northallerton proposals. The key findings and the modelling are:

- In the 2026 design year the introduction of the NNDA and Link Road would reduce traffic flows over the A167 Low Gates level crossing thereby reducing queues and delays. Two way traffic flows at this location **reduce by 164 and 232 vehicles** during the busiest morning and evening peak periods respectively. This represents a reduction of 10% and 13% in the AM and PM peak hours respectively and significant betterment when compared to the without North Northallerton (and thus it's Link Road and Bridge) scenario;
- Traffic flows along Quaker Lane reduce even more substantially in both directions; up to 260 vehicles in the AM peak and 420 vehicles in the PM peak. This represents a reduction of 42% and 62% in the AM and PM peaks respectively and therefore significant

betterment when compared to the without North Northallerton (and thus it's Link Road and Bridge) scenario;

- Consequently there are reductions in the capacity, queue and delay values when compared to the baseline position at the High Street/Quaker Lane, Brompton Road/Quaker Lane, and Stokesley Road/Northallerton Road junctions. This is notwithstanding that the capacity at these junctions will be improved further by the proposed mitigation schemes identified in the Traffic Assessment; and
- Due to the weight restriction on Quaker Lane, the HGVs currently routing down to Friarage Street (from A19/Teesside via A684) and back up the A167 High Street, will reassign across the new bridge.

In summary then, the detailed strategic modelling using leading software demonstrates that the proposed Development with new Link Road and Bridge will provide a significant material reduction in traffic flows at Low Gates level crossing, as well as along Quaker Lane.

Equally importantly the Town Council finds it difficult to understand why the development within what is to be Phase 1 of the application – ie to the north of the Thurston Road Industrial Site, is even being considered as a location for residential development. If this residential development takes place then where in the future will Northallerton be able to expand its industrial facilities? Surely this land needs to be earmarked for future industrial use as once an alternative use such as housing takes place on this site there are then no alternative locations within the Northallerton area where industry can expand to upon an upturn in the national economy.

The very economic future of Northallerton and future job prospects will be curtailed once this appropriate industrial land has been built upon.

Indeed the very siting of this proposal is an issue – why add to Norhallerton problems when there are areas to the east, west and south of the town that could be utilised for housing development and which would not add to the towns problems, especially as these alternative locations would not be suitable for any future industrial development should it be required in years to come?

Such sites to the east, west and south of the town would by their very location have fewer transport issues either related to the road or rail networks.

Land to the north of Thurston Road has been earmarked for mixed use development since the site was Allocated in 2010 including residential, retail, light industry and commercial uses. Furthermore, land to the west of Darlington Road, (north of Standard Way industrial estate) which does not form part of this planning application remains allocated for commercial, general industrial and storage and distribution uses. As a result, there is still land allocated for employment purposes and it is likely that land to the north of Allerton Steel on Thurston Road will be occupied by commercial or light industrial uses and/or the possible expansion of Allerton Steel. As this part of the application is in outline only, the allocation of use will not be determined until reserved matters applications are made at some point in the future. In any event, the principle of mixed use on the land north of Thurston Road is already established; it is the detail that has to be approved at the relevant time.

With regard to other potential sites for employment uses, the Council is in the process of reviewing its Development Plan and recently went through the first round of 'call for sites' consultation. This is one of the first steps in preparing a new Local Plan and establishes what land is available for development within the District over the future plan timeframe. If there are sites in the east, west and south that are suitable for development they can, subject to need, potentially be allocated through this process. However, at the present time the Council has an adopted plan and adopted allocations which have come forward for development and they must therefore be given due regard.

The Council is also concerned that there are no cast iron guarantees in place that would lead to the eventual construction of any link road or bridge since it seems that the developer is only being required to start construction on such facilities. Whether in reality this leads to any more substantial construction other than a minimum few holes in the ground etc. very much remains to be seen.

This matter will be discussed in detail in Section 6.3 of the report. In summary, however, the traffic assessment has identified that the road and bridge would not be required before 495 houses are built on this site, but as the Council consider that it is an essential piece of infrastructure for the town as a whole to provide an alternative to Low Gates it is expected to

be built in the first phase. Furthermore, as the Council has secured £6 million of funding from the Local Enterprise Partnership (LEP) towards construction of the link road and bridge, which has to be spent on the construction of the road and bridge by March 2017, there is an incentive for the developer to undertake the works with contribution from the LEP in the first phase, particularly as it would provide better access to their developments and thus increase sales potential. The developers have agreed to commence construction of the road and bridge in the first phase and this will be secured by Section 106 Legal Agreement between the parties.

As regards Affordable Housing it is understood that a higher quota may be achieved through the future development of other parts of the building Phases which are not the subject of the initial constructions. This however leads to concerns that there is an issue if the future Phases of the developments do not come to fruition, and there are no guarantees that future building will actually ever take place. It is also the opinion of the Council that any affordable housing should be strictly allied to local need.

This matter will be discussed in Section 6.4. In summary, while the Council has a policy of seeking 40% affordable housing on all schemes above 15 dwellings, this is subject to a viability appraisal of the development proposal. A viability appraisal is a complex process which factors in matters such as land costs, normal and abnormal build costs, sales values, and contributions towards on and off-site infrastructure. The greater the costs and contributions sought from the development, the less viable a scheme becomes, and therefore the less affordable housing that can be provided. If the District Council were to ignore the viability appraisal (which in this case has been independently reviewed and negotiated by the District Valuer on the Council's behalf) and seek a provision in excess of that shown as viable, the decision could be challenged by the applicant and the Council would find it difficult to defend its request. For this reason, it is instead appropriate to insist on a viability review, secured by legal agreement with the landowners, at various points in the development process. This will allow the Council to reassess build costs and sales values again in the future and if it is identified that changes in the market make the development more viable then the Council can secure more affordable housing.

However, this means that there may be more affordable housing in later phases than in the first phase and thus not as evenly distributed throughout the development as the Council would like.

With regard to affordable housing serving local need, this is a requirement of the District Council and is ensured by legal agreement.

This point therefore leads nicely onto the question of whether there is indeed any proven need for this development to take place at all within the Northallerton area – who is going to live in these houses and what is actually the proven local need for such a large development?

To the Council's knowledge there has been no proven need for such a large-scale residential development to take place in Northallerton comprising the type of dwellings envisaged in the application, particularly if only minimal Affordable Housing is going to be achieved.

The issue of need is determined by national government and their requirement for Council's to maintain a 5 year housing land supply based upon objectively assessed housing need. The issue will be discussed further in Section 6.1, however, it must be noted that National Planning Practice Guidance says that a housing market area is a geographical area defined by household demand and housing preferences and that such needs are rarely constrained precisely by local authority boundaries. Instead it is considered that a housing market area is a geographical area reflecting the key functional linkages between places where people live and work. It is therefore advised that local planning authorities should establish their housing needs through joint working with other local authorities in the relevant housing market area. What this means is that some of the housing allocated in Hambleton will be taken up by people who work in other local authority areas, but this cannot be avoided.

Notwithstanding, the allocation of affordable housing will be restricted to those people who have a recognised connection to the local area, as per the criteria below:

- Currently live in the Northallerton Sub Area (i.e. Northallerton, Romanby and Brompton) and have been resident for at least 6 out of the last 12 months; or
- Have lived within the Northallerton Sub Area for at least 3 years out of the last 5 years; or

- Are employed in the Northallerton Sub Area. Employment is defined as meaningful permanent full or part time. Not casual or seasonal.
- Have an essential need to live close to another person, to provide or receive care or support, who currently lives in the Northallerton Sub Area, and who has been resident for the last 5 years, In this case the applicant can only bid to a Local Authority where the family member resides.
- Has a close family member residing in the Northallerton Sub Area that has done so for the last 5 years. (mother, father, adult son, adult daughter, adult brother, adult sister):
- One of the partner local authorities has accepted a duty to house the applicant from another council under the terms of Housing Act 1996 part 7.
- People who need to move to a particular locality in the Northallerton Sub Area, where failure to meet that need would cause hardship (to themselves or others) and that will resolve this need. (Examples of hardship are given in Appendix 20). Applicants would be restricted to applying for accommodation in the Local Authority area where employment / training / family member resides.

However, if a local person in need of the accommodation cannot be found, a cascade mechanism exists to widen the search area to the District as a whole. The above provisions would be secured by Section 106 Agreement.

Applicants who do not have a local connection may not qualify for access to the housing register.

Finally, as regards the proposed development itself the Town Council has serious concerns about the placing of both a Primary School and Aged Persons accommodation in close proximity to industry, particularly in the vicinity of the heavy industrial site at Allerton Industries. It is totally inappropriate for young children to have to spend their days so near to this very noisy site and it is also equally inappropriate for elderly people.

See Comments provided in Section 5.5.27 in response to Brompton Town Council.

5.2.29 British Transport Police

A late response provided below in *italics* has been provided by British Transport Police:

- BTP has serious concerns about the positioning of the new Primary school in between two very busy and very fast railway lines. The Middlesbrough Line and the East Coast Main Line.
- We would also like to see more detail about the proposed green spaces around the new bridge. If they have to be adjacent to the tracks their design will have to be very carefully considered to discourage trespass and stone throwing.
- The entire area would need large scale upgrading of railway fencing to prevent the increased numbers of children living locally from gaining access to the tracks.

As the location of the school is in outline only, with no detailed design of the building or its perimeter, it is considered that safety concerns can be overcome at the detailed design stage with involvement of British Transport Police at the relevant time. With regard to the concerns of accessibility of the railway line from the site, it is considered that this is most likely to be controlled through separate permissions and agreements with Network Rail, however, a planning condition can require further detail of the interface with the railway line prior to the occupation of any of the dwellings.

5.3 Public Notification

Members of the public were notified by way of letter (a total of 334 letters were sent), press advert, locally posted notices, and a website release on the Council website. To date the Council has received 15 letters of objection but no explicit letters of support. The objections relate to the following issues, some of which have been addressed in response to statutory consultees above, and some of which will be addressed or elaborated on in section 6:

Highways:

• The proposal would lead to greater congestion in Northallerton. A better solution would have been a Northallerton bypass to take traffic out of the town.

- An additional roundabout at Brompton Road and Northallerton Road would be better than creating an awkward crossroads.
- The proposed link road would be noisy and dangerous to cross, especially for children.
- Increased traffic on Northallerton Road will mean it is much more dangerous for cyclists and school children walking to schools in this area.
- The proposed link road will do nothing to alleviate congestion that is created by the Northallerton Middlesbrough railway line dissecting the town with the main crossing point being at Low Gates at North End. This is not an issue addressed in the application and no resolution is apparently under consideration by the Council.
- The solution to Northallerton's traffic congestion is to build a bridge over Low Gates.
- The solution to Northallerton's traffic congestion is to redirect the railway.
- The Council could offer some financial incentive to Network Rail towards the costs of diverting the line before building over the railway makes such a move impossible.
- How are children and elderly residents going to be able to cross the new road to town and back?
- The alignment of the link road does not accord with the alignment in the adopted Masterplan for the site.
- The road and bridge should be constructed before any houses are built.

Flooding/Drainage:

- A high number of properties would be built on floodplain and would therefore be liable to flood.
- Brompton Village regularly floods. The proposal is to build a substantial amount of hard surfacing on existing fields just up the hill from Brompton Village, which will result in overspill to Brompton.
- Land raising on Northallerton Road to construct the new houses could cause off-site flooding to adjacent existing houses because they presently have a ground level above the adjacent fields and are therefore not impacted by standing water that sits on the fields after periods of heavy rain. This would change if the field levels were changed to align with existing property ground levels.
- The Sewage works for the town is already at 98% capacity.
- The Council should use CIL to make up the missing funds for flood alleviation schemes in Brompton and Romanby.
- Housing Need:
- At a time when over 1,300 jobs have been lost locally, there is no need for new houses.
- There is already a surplus of houses on the private housing market in Northallerton, Brompton and Romanby, many of which have remained unsold for a considerable length of time, proving there is an oversupply.
- There are more appropriate locations within the District for the existing housing stock to be supplemented.
- The limited provision of affordable housing should be resisted, because if demand for housing does exist, it is surely likely to be in the affordable housing bracket.
- Utilising this site for housing prevents further industrial growth and thus enhanced employment opportunities.
- There are currently at least three housing developments underway or proposed in Northallerton (industrial ground opposite Standard Way, the former Prison, and the site of the Scout and Guide Group huts on Malpas Road); over development is not an answer to housing needs.

Loss of Green Fields:

• The pleasant break between Brompton Village and Northallerton would be lost turning Brompton into another suburb of Northallerton.

• The green belt of land between Northallerton and Brompton is important to maintain the identify of the village. Residents reserve the right to challenge in the courts any change to the decision to maintain the Brompton Green Belt.

Miscellaneous:

- House values nearby would be reduced enormously due to increased noise and traffic intensity.
- Noise from Allerton Steel potentially resulting in poor amenity for residents and children at the school and/or restrictions on the operating hours or potential expansion opportunities (which could lead to more local jobs) of Allerton Steel.
- There would be reduced amenity at neighbouring properties on Northallerton Road from the close proximity of the new buildings to existing dwellings.
- Shopping locally includes existing shops and supermarkets in the town. As there are many shops already empty, will additional outlets increase vacancy rates?
- The development seems to imply a significant negative environmental impact. How are the developers going to ameliorate or offset this devastation?
- The proposed development has moved outside of the development limits and cannot be approved as a result.
- The level of affordable housing is not declared but is rather 'subject to viability'. What is the level proposed?
- The phasing of development does not comply with the Development Plan Allocations Document. The Council approved dwellings from later phases in this phase.
- The quality of some of the documents submitted has been poor, particularly the mapping.
- The planning application falls foul of the principles established within the Rochdale cases (*R. v Rochdale MBC ex parte Milne* (No. 1) and *R. v Rochdale MBC ex parte Tew* [1999] and *R. v Rochdale MBC ex parte Milne* (No. 2) [2000] that sufficient information to enable 'the main,' or the 'likely significant' effects on the environment to be assessed...., and the mitigation measures to be described, and could therefore be subject to future challenge.

Noise:

Specifically with regard to noise, Allerton Steel also lodged an objection accompanied by their own independent acoustic report prepared by Dragonfly Consulting. The conclusion of the Dragonfly report is provided in *italics* below:

The assessment has shown that the existing operation of Allerton Steel Ltd will have significant adverse impact on the proposed residential development for this site.

Given the context of this assessment, it is considered that these noise levels are significantly above the Significant Observed Adverse Effect Level for this assessment.

The assessment provides a strong indication that the proposed residential dwellings will be significantly negatively impacted by the existing commercial noise sources to the south of the proposed development site, in particular Allerton Street Ltd.

Furthermore, that impact will increase once the noise from the proposed link road is factored into the assessment, resulting in noise levels in gardens and amenity areas in excess of those considered 'desirable' within BS8233:2014.

A full copy of the Dragonfly Consulting report is available on Public Access but will be discussed in Section 6.7 of this report.

6.0 OBSERVATIONS

The principle of development for mixed use in this area is established by the allocation of the majority of the site in the Allocations Development Plan Document, as identified in Section 3. The main planning issues, many of which have emerged through consultation, to take into account when considering this application therefore relate to the following:

- Housing Need (Section 6.1)
- Suitability of the Mix of Outline Land Uses (Section 6.2)

- Transport Considerations (Section 6.3)
- Affordable Housing Provision (Section 6.4)
- Drainage and Flooding (Section 6.5)
- Nature Conservation (Section 6.6)
- Noise (Section 6.7)
- Design and Landscaping (Section 6.8)

6.1 Housing Need

- 6.1.1 There have been a number of objections lodged with the Council suggesting that there is a lack of need for houses in Northallerton and/or that the proposal does not accord with the phasing timetable identified within Allocation NM5 and Development Policy DP11, such that the development should not be allowed to proceed. There have also been suggestions that because the proposed Development expands the Development Limits an outright refusal is warranted.
- 6.1.2 While it may be the case that there are many homes up for sale in Northallerton and that other sites, such as York Trailers, are under construction, based on population growth and projections, the Council is required to identify and help bring forward land for housing to meet future need.
- 6.1.3 Indeed, it is necessary to highlight that since the LDF was adopted, the Government has issued the National Planning Policy Framework (NPPF), which is a material consideration in the assessment of planning proposals. Paragraph 47 of the NPPF requires local planning authorities to boost significantly the supply of housing by using their evidence base to ensure that their Development Plan meets the full, objectively assessed housing need (hereafter referred to as FOAHN) for market and affordable housing, including identifying key sites which are critical to the delivery of the housing strategy over the plan period. Paragraph 47 also specifies that local planning authorities should identify and update annually a supply of specific deliverable sites sufficient to provide five years worth of housing against their housing requirements with an additional buffer of 5% to ensure choice and completion in the market. Paragraph 49 asserts that relevant polices for the supply of housing should not be considered up to date if the local planning authority cannot demonstrate a five year supply of deliverable housing sites.
- 6.1.4 Hambleton's LDF predates the NPPF and was prepared in the context of the Regional Strategy for Yorkshire and Humber (RSS), which has now been revoked. The RSS was used to inform the basis of the housing requirements set out in the Core Strategy at Policy CP5 (being 320 houses per annum between 2004 2011, 290 houses per annum between 2011 2016, and 260 houses per annum between 2016 2021) and was a constrained figure aimed at directing some of the growth that would have otherwise occurred in the region's rural areas, such as Hambleton, to the region's cities and conurbations. Hambleton's identified housing need within the LDF is consequently a restrained figure and not based on the area's FOAHN² and is therefore out of date. In that regard Policy CP5 carries no weight in the consideration of this application.
- 6.1.5 In recognition of the conflict between the LDF and the NPPF on housing need and supply requirements, in November 2013 the Council's Cabinet considered a report³ on the 5 year supply of deliverable sites. The report contained an analysis of the supply position, as at 1 October 2013, of sites with planning permission and sites allocated for future development.
- 6.1.6 The report concluded that the Council was not in a position to maintain a 5 year supply of deliverable housing sites and that there would be a continued deficit if the LDF phasing restrictions remained in place. The report therefore recommended the

² For clarity, National Planning Policy Guidance (NPPG) advises that the assessment of FOAHN involves three stages. The adjustment of the latest household projections on the basis of robust demographic evidence, further adjustments to account for job trends and a final adjustment to reflect changing market signals in quantity and price.

³ http://democracy.hambleton.gov.uk/Data/Cabinet/201311050930/Agenda/051113_Item_5.pdf

- relaxation of Phasing Policy DP11 in the DPD to help meet the shortfall by releasing sites allocated in Phases 2 and 3 early. The Cabinet endorsed the recommendations of the report and this was subsequently ratified by full Council on 10 December 2013.
- 6.1.7 Despite relaxing the phasing, and work having commenced on a new Development Plan, which includes reviewing housing site allocation and provision, Hambleton's level of housing need and site provision and delivery has been called into question through a number of recent planning appeals, for example at Easingwold, Great Ayton, Huby and Stokesley whereby the Council opposed housing applications on unallocated sites. The Council's position at those appeals was that it could demonstrate a 5 year housing land supply to meet the need due to phasing having been relaxed; the developers' argument was that the Council could not demonstrate a 5 year supply and that housing on these alternative unallocated sites should therefore be permitted.
- 6.1.8 Planning Practice Guidance confirms that establishing future need for housing is not an exact science and that no single approach will provide a definitive answer. Indeed, this is no more demonstrated than in the decisions for Great Ayton, Huby and Stokesley whereby the Inspector in the Huby appeal came to a different conclusion to the inspectors in Great Ayton and Stokesley on the figure for Hambleton's FOAHN. Notwithstanding, each of those decisions identified that the Council did not have a 5 year supply.
- 6.1.9 The Huby decision decided on 27 May 2015 (Planning Application Ref. 12/01243/FUL) identified that Hambleton had a requirement of 470 dwellings per annum, while the Great Ayton decision decided on 10 July 2015 (Planning Application Ref. 13/02275/OUT) and the Stokesley decision decided on 07 Sept 2015 (Planning Application Ref. 14/00337/OUT) identified that Hambleton had a requirement of 458 dwellings per annum. These decisions therefore confirm that the provision in Policy CP5 of the LDF (of 290 houses between 2011 2016 and 260 houses between 2016 2021) is too low and that relaxed phasing is justified.
- 6.1.10 However, in addition to the annual requirement, as noted above, the NPPF requires an additional 5% to be added to ensure choice and competition in the market, as well as the addition of house building shortfall from the previous years. When these figures were taken into account, the Inspector at Huby found that Hambleton has a housing requirement of 587 dwellings per annum, the Inspector at Great Ayton found a housing requirement of 608 dwellings per annum, while the inspector at Stokesley found a housing requirement of 643 dwellings per annum.
- 6.1.11 When all unpermitted allocated sites, undeveloped sites possessing planning permission, and windfalls were considered, it was determined in Huby that Hambleton has only 4.8 years worth of housing supply, it was determined in Great Ayton that Hambleton has only 4.4 years worth of supply, and it was determined in Stokesley that Hambleton has only 4.24 years worth of supply. What these decisions therefore demonstrate is that there is a need for the housing proposed at North Northallerton, it is greatly emphasised in National Planning Policy, and that to avoid sporadic development in unsustainable and unplanned locations, the allocated sites should be encouraged to come forward. The proposal is therefore compliant with Policy CP1 which seeks to ensure that development is sustainable, it is consistent with Policy CP2 which seeks to minimise the need to travel, and continues to meet the general requirements of Policy CP6 which seeks to deal with the sustainable distribution of development.
- 6.1.12 With regard to the expansion of the Development Limits, Policy CP4 requires an 'exceptional case' to be made in line with certain sub clauses and Policy DP9 has similar requirements. However, as highlighted in the Huby and Stokesley decisions, the development limits were defined in the context of the restrained development strategy, do not accord with the the NPPF, and cannot be considered up to date where the Council has to meet a higher FOAHN. The use of some greenfield land outside of the Development Limits is therefore inevitable and other exceptional circumstances should not be precluded; so Policies CP4 and DP9 are deemed to have limited weight in this instance. This is particularly so given that the application site accords, for the most part, with the allocation and is consistent with aims and objectives of the allocation.

- 6.1.13 In addition, it is necessary to bear in mind that the expansion of the site has to be viewed in the context of the infrastructure requirements being sought from the developers, specified through Policy DP2 and Allocation NM5F. The Link Road and Bridge envisaged by Allocation NM5F was intended to be delivered from pooled contributions from various developers and accrued over time; however, the Council and NYCC Highways are requesting that the applicants deliver the road and bridge in the first phase in recognition of the wider benefit that it would have for Northallerton and the other site allocations in the NM5 area. The applicants have agreed to provide the land and deliver the construction of the entire road and bridge, but this places an additional financial burden on them above that required by the new Community Infrastructure Levy (CIL) tariff (see Section 6.3 for more detail). In addition, the primary school has been added to the Allocation into one of the core residential areas. Therefore in order to make up the additional infrastructure funds, and in order to replace the residential land now allocated to the school, the proposal includes expansion of the Development Limits for additional market housing.
- 6.1.14 Had Allocation NM5C, which relates to land owned by the Church Commissioners, and Allocation NM5E to the west of Darlington Road also come forward at the same time the cost burden for funding the infrastructure would have been shared by an additional developer, and expansion of the development limits may not have been required or justified. As that is not the case, the Council has to weigh up benefit and harm of expanding the Development Limits and, as will be demonstrated elsewhere in this report, the planning officer considers that the benefits of allowing the road and bridge to come forward in a timely manner, outweigh the harm of expansion, particularly when also considered in light of Hambleton's housing shortfall.
- 6.1.15 Thus, given that the site is allocated for development of the type proposed, given that it would immediately contribute towards the delivery of required housing, and given that it can deliver important infrastructure for the town, the need for the Development is demonstrated and no other material considerations would lead to a different conclusion.

6.2 Suitability of the Mix of Outline Land Uses

- 6.2.1 In addition to the proposed residential use of the land, the proposal incorporates a Local Neighbourhood Centre containing retail and services, a primary school, an extra care facility, open space/recreation land and commercial/employment use. This is commensurate with the NPPF (Paragraph 38), which states that for larger scale developments, planning policies should promote a mix of uses in order to provide opportunities to undertake day to day activities including working on site. Where practical, particularly within large scale developments, key facilities such as primary schools and local shops should be located within walking distance of most properties.
- 6.2.2 The proposed uses will be discussed in turn. However, as each of the uses would be subject to reserved matters applications, it is important to note that there is limited dialogue on design or layout.

Local Neighbourhood Centre

- 6.2.3 The proposed Local Neighbourhood Centre (LNC) is proposed to equate to a floor area of 1965 m² (as identified in Chapter 3 of the Environmental Statement), and is proposed to comprise Use Classes A1 (Shops), A3 (Restaurants and Cafes), A4 (Drinking Establishment), C1 (Hotel) and D1 (Non-Residential Institutions), which includes clinics, health centres and crèches for example. The LNC would also be located to the east of Darlington Road, at the western entrance to the site, adjacent to the proposed village green, within allocated site NM5D.
- 6.2.4 While NM5D is allocated for mixed housing and employment development, the aforementioned uses are not specifically endorsed within that allocation. However, the adopted 2011 Masterplan for the site identified a need for a community hub on the site and the proposed location for the LNC accords with that in the Masterplan. Given the extent of this allocation in the Local Plan and the fact that it will contribute towards meeting the target of achieving at least 51% of all new development within Northallerton, it is essential that this new neighbourhood has all the necessary facilities within close proximity to make the new development as sustainable as possible. This corresponds with the principles of Policies CP1, CP2 and CP3 for developments to include a sustainable and complementary mix of uses, minimise the

- need to travel particularly for everyday needs, and provide a focus for activities and improve community well being. In that regard, and while the applicants have not submitted specific details for the actual floorspace of each suggested use class, each of the uses are supported in principle.
- 6.2.5 During the course of the application, however, the applicants submitted a Retail Impact Assessment (RIA) for a supermarket within the LNC. The RIA stated that the supermarket is anticipated to comprise up to 1,723 m² gross retail floorspace and would most likely be occupied by a discount food retailer. When having regard to the floor area suggested by the Environmental Statement for the LNC, the proposed supermarket appears to constitute the bulk of the Centre space, leaving little room for the other mix of uses.
- 6.2.6 Local Plan Policy DP23 states that retail proposals in excess of 500 m² gross in size will not be permitted outside the primary retail areas designated under Policy DP21 unless the proposal can be demonstrated to be acceptable under national planning policies in PPS6. For clarity, PPS6 has been revoked but its main principles have been translated into the NPPF.
- 6.2.7 The NPPF is very clear on a 'town centre first' approach. Paragraph 24 of the NPPF states that local planning authorities should require applications for main town centre uses to be located in town centres, then in edge of centre locations and only if suitable sites are not available should out of centre sites be considered. This is known as the 'sequential test'. Secondly, local planning authorities are then required to consider the impact of the proposal on existing, committed and planned public and private investment in a centre or centres in the catchment area of the proposal; and the impact of the proposal on town centre vitality and viability. This is known as the 'impact test'.
- 6.2.8 The Council engaged England and Lyle Planning Consultants to review the applicants' RIA and provide advice on the robustness of the assessment. It was identified that while the applicants' impact assessment was generally correct, in that the proposed supermarket in North Northallerton would not harm the vitality and viability of the town centre, it had failed the sequential test because it had dismissed the availability of part of the former HM Prison Site, which has identified through its adopted Masterplan the provision of a foodstore to the southern part of the site. Furthermore, as the former HM prison site is under the sole ownership of Hambleton District Council the redevelopment of the southern part of the site for a foodstore could be brought forward as a first phase of development in advance of development options for the wider Northallerton Prison site.
- 6.2.9 Pursuant to the Paragraph 27 of the NPPF, where an application fails to satisfy the sequential test it should be refused. The applicants were therefore advised that the Council could not support its application for a supermarket in excess of 500 m² gross in size, as it would be contrary to Local and National Policy.
- 6.2.10 The applicants disagree with the planning officers on this point but have advised that they will not pursue the larger retail unit at this stage. This does not prevent the applicant from making a fresh application for the supermarket at a later stage, but for the purpose of this application, it is considered necessary to include a planning condition on any permission granted to restrict the size of any individual retail unit on the site to no more than 500 m². This is in line with Local Plan Policy DP24, which states that proposals for small scale retail developments (below a threshold of 500 m²) outside the primary retail area boundaries will be permitted where these serve neighbourhoods and residential areas and would not jeopardise the vitality and viability of the primary retail area.
- 6.2.11 It is considered that the addition of the mixed use area at the scale proposed, with a small supermarket, is likely to attract users from a relatively small catchment, most likely within the body of the site and would be a valuable community facility. A small supermarket with the other A Class uses is therefore supported in principle.

Commercial/General Industrial Use

6.2.12 According to Chapter 3 of the Environmental Statement, the proposal includes the provision of 3250 m² of employment floorspace, in use classes B1 (Business) which includes offices, B2 (General Industrial) and B8 (Storage and Distribution). It is

unclear from the illustrative Masterplan submitted with the application where the commercial/industrial uses are proposed to be located, particularly as Allocated Site NM5E to the north of Standard Way which is allocated for all of these uses, is excluded from the application. However, as noted in Section 3.7 above, Allocation NM5D to the east of Darlington Road provides for B1 and B2 uses, thus it is likely that the land to the south of the link road shown on the Illustrative Masterplan as 'detail to be determined' is the relevant location.

- 6.2.13 This location would in theory be appropriate for employment uses because it can integrate with similar uses that front Thurston Road. Furthermore, it would form an interface between heavy manufacturing on Thurston Road and the new residential neighbourhood to the north, which could in practice also act as a noise buffer for the dwellings to the north, as will be discussed later in this report. Additionally, and as promoted by Local Plan Policy DP16, the provision of mixed use schemes can improve accessibility between homes and jobs and they embody sustainable development principles. They are therefore to be encouraged particularly if it can be demonstrated that there would be no unreasonable adverse impacts resulting from the proximity of the uses to the residential development to the north. However, as the proposals are in outline, it is difficult to determine whether there would be any adverse impacts arising, although noise is an obvious concern (particularly from B2 uses), and design and highways movements from the B8 uses is also noted given that B8 uses tend to be larger footprint uses with heavy goods vehicle movements.
- 6.2.14 When dealing with multi-stage consents that have been subject to an Environmental Impact Assessment, National PPG states that:

"The likely significant effects of a project on the environment should be identified and assessed at the time of the procedure relating to the principal decision. However, if those effects are not identified or identifiable at the time of the principal decision, an assessment must be undertaken at the subsequent stage. Under the Town and Country planning system this could be prior to the approval of reserved matters following a grant of outline planning permission; or of matters required by a condition attached to a full planning permission."

- 6.2.15 With that in mind when considering proposals with limited information the Council can either refuse the development or they can seek further environmental information at the time of discharge of conditions or submission of the relevant reserved matters application.
- 6.2.16 With regard to proposed B8 use class, it is important to recognize that the Allocations Document and the adopted Masterplan for the site located the larger footprint uses in the less sensitive location to the west of Darlington Road whilst higher quality, finer grained offices and workspaces were identified for the east of Darlington Road along the interface with the new residential neighbourhoods. No justification has been provided by the applicant to vary this allocation or layout, although it is feasible that B8 uses can be small scale and should not necessarily be deemed inappropriate on this site.
- 6.2.17 That said, it is noted that the Transport Assessment (TA) submitted with the application has assessed the development impact of the 900 dwellings, an extra care facility of 60 beds, and B1 and B2 employment uses, but it has not assessed the impact of B8 uses in this location or provided any assessment of the suitability of the width and splays of the proposed access roads off the south side of the link road into this part of the site for storage and distribution type vehicles. While it is acknowledged that this would form part of a reserved matters application, it is questionable whether access/egress form the southern side of the link road for storage and distribution vehicles can be achieved safely and while maintaining the desired free flow of the road. In this regard, it is considered necessary to prohibit by planning condition the use of any of the site for B8 uses until such a time that a revised Transport Assessment has been provided to the Council's satisfaction to demonstrate that the link road and any side access roads can safely accommodate the necessary vehicles (notably HGVs) and that a revised acoustic assessment has been provided demonstrating that noise from those vehicles or any refrigerated storage unit would not detrimentally impact on the surrounding residential uses elsewhere on the site.
- 6.2.18 Similarly, with regard to the B1 and B2 uses, while they accord with the objectives of

the allocation and adopted Masterplan, no specific acoustic survey results have been able to be provided to satisfactorily conclude that such uses would not harm the amenity of the proposed sensitive land uses, such as the dwellings and the school. While it is noted that Chapter 11 of the ES has identified maximum internal noise levels within the proposed industrial units, this will require further assessment as part of any reserved matters application. This should be required through condition on any planning permission granted. Notwithstanding, the uses are in principle acceptable in this location.

Play and Open Space

- 6.2.19 The proposal includes approximately 19 hectares of open space/recreation land, consisting of a village green, outdoor sports pitches, formal and informal amenity space, all age play areas and allotments. Local Plan Policy DP2 identifies the need for developers to provide public open space, and sport and recreation facilities to serve new residents, Policy CP19 gives support to proposals that lead to the provision of additional recreational assets, Allocation NM5G, which is located within the site, is allocated for recreational development, while the development of a Northallerton Sports Village is one of the Council's priority infrastructure projects that will be funded through CIL receipts. The proposal to include such extensive play and open space, and the proposed dedication of the land by the applicants to facilitate the future development of the sports village and allotments will therefore make that project a realistic prospect and will provide notable benefits to the community.
- 6.2.20 With regard to the location of local equipped play areas, indicative locations are shown on Page 68 of the submitted Design and Access Statement. One is proposed immediately to the north of the primary school land and one is proposed within the sports village. These locations were identified through discussions with officers of Hambleton DC and NYCC as the areas most accessible to new residents of the development. The uses and locations are therefore supported in principle.

Primary School

- 6.2.21 The application seeks outline permission for the provision of a primary school for a two form entry, but on land of at least 2 hectares to allow for future expansion if required. The location of the proposed school is the result of input of the NYCC Education Department, who considered that its location to the west of the railway line and to the north of the link road would best serve the majority of the proposed new residents.
- 6.2.22 There is local concern that education infrastructure would not be sufficient, however, NYCC Education is satisfied that there is sufficient capacity for high school places to accommodate children from the development. However, this is not the case for primary school provision, which is why it is advanced on this site.
- 6.2.23 However, despite identifying the plot for the school, no detailed design has been undertaken. One of the key recommendations of the NYCC Education Officers though, is that the school should be accessed from the residential roads to the north, east or west of the site and not from the link road itself, and based on the illustrative layout in the Design and Access Statement this is seemingly achievable. Consideration will also have to be given to noise attenuation measures in the design process of the school, as discussed in Section 6.7 of the report, due to high noise levels from the road, but this is also achievable.
- 6.2.24 NYCC require the land to be dedicated during the first phase of the development so that the school is built to be operational for intake of pupils in 2018. However, like the Sports Village, the school is one of the Council's priority infrastructure projects and will therefore need to be funded from CIL receipts, although there would be funding options open to NYCC to overcome the delay on the receipt of CIL funds. Notwithstanding that, the proposed use of the land for the school has policy support, is in a sustainable location and would at some point in the future meet the shortfall of school places that currently exists.

Extra Care Facility (i.e. Nursing Home)

6.2.25 The proposal includes the provision of a 60 bed extra care facility somewhere on the south side of the link road as part of the mixed use allocation identified in Allocation NM5D. This accords with the requirement in Local Plan Policies CP8 and DP13 for

housing proposals to meet the needs of all sectors of the community, including the needs of elderly people. Objections have been lodged regarding the possible location of the aged care facility so close to Allerton Steel on Thurston Road, because of the potential noise impacts to elderly residents; however, as will be discussed in Section 6.7 of this report, it is feasible for an extra care facility to be designed with sufficient acoustic attenuation measures, such that it is not considered reasonable to refuse this part of the proposal or direct it to be relocated elsewhere on the site until more detailed acoustic testing has been undertaken as part of a reserved matters application.

Summary

6.2.26 In summary, the proposed uses sought in full and outline are entirely commensurate with the Local Plan Allocations Document and the adopted Masterplan for the site, and are in principle worthy of support subject to other relevant considerations, such as drainage and flooding, noise, highways access, etc.

6.3 Transport Considerations

- 6.3.1 The proposal includes the construction of a new link road and bridge over the Northallerton to Middlesbrough railway to help mitigate the impact of the development itself and to provide some relief to Northallerton Town Centre. This is a key aspiration of Allocation NM5F.
- 6.3.2 As already discussed in Section 6.1 above, the need for the level of housing proposed is demonstrated by the Council's current lack of a 5 year land supply and shortfall of new homes, which has largely resulted from the economic downturn earlier in the decade. However, housing should not be supported in isolation of the necessary infrastructure to connect it to the settlement of Northallerton nor to the significant detriment of the existing highway network.
- 6.3.3 While the aspiration of the Council (as per Local Plan Policy CP2) and central Government in the NPPF is to design developments so as to minimise the need the travel, it is reality that most households own a car and use it on a daily basis. Planning housing growth must therefore include planning, revising and evolving new and existing roads but this takes time and the benefits are not necessarily realised in the short term.
- 6.3.4 There have been numerous objections submitted arguing that the proposal would lead to greater congestion in Northallerton and that because the link road does not solve the problem of queuing traffic at Low Gates level crossing in the town the proposal should be refused. The officer has already provided a response to these suggestions in response to the Town Councils' submissions in Section 5 of the report; however, it is important to reiterate that the issue of Low Gates is a problem which the Council is well aware of but that this application will not make the impact of Low Gates worse. This is because the most significant impact on Low Gates level crossing is the amount of time the gates are lowered, which is entirely dictated by rail traffic, not road traffic. The Council will continue to liaise with Network Rail and County Highways over Low Gates but it does not consider that the problem of Low Gates is sufficient to refuse the proposed application. The focus of the highways assessment in this report and the decision making involved for this application must therefore be on the impact of the development itself on the highway network and whether that can be alleviated, rather than on whether it solves existing highway problems. Resolutions to existing problems are of course desirable but not mandatory.
- 6.3.5 As identified in Paragraph 32 of the NPPF all developments that generate a significant amount of movements should be supported by a Transport Assessment and decisions should take account of whether:
 - The opportunities for sustainable transport modes have been taken up depending on the nature and location of the site, to reduce the need for major transport infrastructure;
 - Safe and suitable access to the site can be achieved for all people; and
 - Improvements can be undertaken within the transport network that cost effectively limit the significant impacts of the development. Development should only be

prevented or refused on transport grounds where the residual cumulative impacts of development are severe.

- 6.3.6 As an historic town, the road network in Northallerton has evolved over time consisting mainly of a number of radial routes emanating from the centre of town and connecting it to places such as York, Darlington, Richmond, Harrogate, etc. While the network was once well planned and sufficient, the congestion now caused through level crossing closures and traffic volumes at peak times is now a constraining factor for housing, employment and business growth. Indeed, with the Low Gates crossing at the northern end of the High Street being closed for a minimum of 20 minutes per hour and Network Rail indicating that freight movements may increase on that line, further barrier down time is anticipated. The town's housing growth needs and employment land growth needs will undoubtedly generate additional traffic and traffic modelling has highlighted that the road network is not equipped to take the extent of this growth.
- 6.3.7 A key objective of Allocation NM5F is therefore to add capacity to the local road network to enable growth, as well as provide an alternative for transport users to move around North Northallerton or between Northallerton and other locations without being delayed at the level crossings, such as Low Gates. The proposed 1.4 km addition of the link road would therefore add that capacity and would allow the mixed use development to proceed to meet the needs of the existing and future community.
- 6.3.8 The route of the road and how it sits in the context of the North Northallerton development area has been established through an extensive masterplanning exercise, supported by the applicants' traffic consultants, NYCC Highways, Network Rail and the Homes and Communities Agency's Advisory Team for Large Applications (ATLAS). Preliminary work to enable a detailed design has been undertaken and this has been supported by additional technical work including geotechnical surveys and flood impact assessments. The link road is principally designed to 'Design Manual for Roads and Bridges' (DMRB) standards and has sought to balance the urban requirements of the route, i.e. managing traffic speeds and promoting pedestrian/cycle movements against the strategic requirements to ensure it remains an attractive route for commercial and other vehicles wishing to avoid Low Gates level crossing and the town centre. It would therefore be 7.3 m wide, have a 30 mph speed limit, have a 3 m wide shared footway/cycle way to one side and a 2 m wide footway to the other, a 70 m stopping sight distance, street lighting, and on carriageway bus stops (i.e. no bays).
- 6.3.9 The link road would be accessed from the creation of two new roundabouts, one on Darlington Road and one on Stokesley Road and both would reduce/manage vehicular speeds as drivers transfer from a rural to an urban environment thereby enhancing highway safety and encouraging pedestrian/cycle trips; they would also provide a defining location for a change in the speed limit. Where the link road crosses Northallerton Road though, it is proposed to form a 'ghost island' right turn priority 'T' junction with the northern section of Northallerton Road leading to/from Brompton. The remaining southern section of Northallerton Road between Stokesley Road mini-roundabout and the link road will become a cul-de-sac to vehicular traffic and a turning head provided.
- 6.3.10 There have been suggestions from the public that the T Junction at Brompton Road and the link road should be replaced with a roundabout. It is, however, important to note that the link road has been designed to primarily function as an alternative route through Northallerton (particularly for HGVs), at the request of NYCC highways, who will become the adopting authority. This has had the effect of altering the original vision of the road as a low speed village street, to a more strategic through route, which therefore needs to benefit from the least number of formal stopping/slow points. In addition, further reasons why Northallerton Road does not form an 'all movements' crossroad or roundabout junction with the link road is because there is a desire to create a safe route for pedestrians and cyclists travelling between Brompton and Northallerton, and by leaving the through road open there are concerns that the southern section of Northallerton Road would effectively become a rat run.
- 6.3.11 The design of the link road has been approved in principle by NYCC Highways as the LHA that will adopt and maintain the infrastructure. Their comments have been provided in full in Section 5 above and will inform the planning conditions attached to

any planning permission granted. Nonetheless, the planning officer considers that the location of the shared cycleway/footway on the south side of the link road would have more benefit on the north side of the link road given that the majority of housing and the school is located on the north side. Advice was sought from NYCC Highways regarding whether there would be any design impediment to swapping the single footway and shared cycleway/footway over and none was identified. To encourage safe routes to school and thus minimise the number of road crossings that children on bikes would have to make, a planning condition should be attached to any planning permission granted requiring the single footway to be located on the south side of the link road and the dual cycleway/footway to be located on the north side, unless through the detailed design of the road for construction it is determined by NYCC Highways that this cannot be safely achieved.

- 6.3.12 With regard to assessing the impact of the proposed uses on the existing network, and in turn determining the point at which the link road and bridge would be required, extensive modelling has been undertaken by Optima Highways Consultants in consultation with NYCC Highways and this is summarised in the Transport Assessment that was submitted with the application. It was determined that the point at which material capacity concerns arise on the overall highway network and thus the trigger point at which the proposed link road and bridge is required to become operational, is after the construction of 495 dwellings. Thus in principle there is no reason why the full part of the planning application for 291 houses should be refused.
- 6.3.13 As noted in Section 5.2.1 though, the road and bridge is intended to be delivered in the first phase and will be required as such through Section 106 Agreement. After construction of the road and bridge it is determined that the proposed link road will provide the following key strategic benefits:
 - Sufficient network capacity to accommodate the 900 houses proposed on the wider site;
 - Removal of a proportion of traffic movements from the town centre, in particular east-west flow reductions on Quaker Lane;
 - Removal of HGV movements routing through the town and along Friarage Street;
 - Reduction in the number of vehicles utilising Low Gates level crossing thereby reducing queues and delays at this existing constraint; and
 - The associated environmental benefits of reduced noise and vehicle emissions within the town.
- 6.3.14 The traffic flow reductions are also summarised as follows:
 - Reductions in the ratio of flow to capacity, queue and delay values when compared with the current baseline position at the High Street/Quaker Lane, Brompton Road/Quaker Lane and Stokesley/Northallerton Road junctions;
 - Reductions in the ratio of flow to capacity, queue and delay values when compared with the current baseline position at the Low Gates level crossing; and
 - Much of the remaining highway network either remains within capacity or is relatively unchanged compared to the baseline position.
- 6.3.15 As reported in the Transport Assessment and supported by NYCC Highways, the development and link road scenario provides an appropriate level of mitigation across the wider highway network and offsets the impact of the additional development trips, as was envisaged when the Site was allocated under Policy NM5 of the LDF.
- 6.3.16 It is clear from the modelling and NYCC Highways review of the data that even without the link road and bridge in place, the existing road network will sustain some housing being built ahead of the link road and bridge being delivered (this figure is identified as 495 houses). Notwithstanding, HDC and NYCC Highways wish to bring forward the delivery of the road and bridge in order to maximise the economic potential of the town and to improve movement within the town in the immediate term, rather than wait to reach critical mass. It is as a result of this that the link road and bridge are identified on the Council's 'CIL Regulation 123 List', which is discussed below.

- 6.3.17 The cost of developing the link road and bridge is estimated at £12.3 million (2015 prices). However, as a key infrastructure priority on the CIL Regulation123 List, it must now be funded through Community Infrastructure Levy (CIL) receipts, rather than negotiated developer contributions; furthermore, CIL receipts are largely dependent on housing sales and the rate of housebuilding.
- 6.3.18 For readers of this report who are not familiar with CIL, it is a planning charge introduced by the Planning Act 2008 as a tool for local authorities in England and Wales to help deliver infrastructure to support the development of their area. It came into force on 6 April 2010 through the Community Infrastructure Levy Regulations 2010. Prior to the introduction of CIL, the amount of money a developer contributed towards infrastructure could be negotiated, however, CIL is a set charge and the developer cannot be expected to top it up with additional contributions for the same piece of infrastructure. Hambleton's CIL Charging Schedule equates to £55 per square metre of gross internal floor area of each private market housing dwellings.
- 6.3.19 As CIL is only applicable to private market housing the amount of CIL that can be collected reduces when more affordable housing is provided as part of the development. Thus, while the pursuit of 40% affordable housing is a Council objective, the delivery of key infrastructure projects can potentially be impacted by that objective because less market housing means less CIL, which means that the CIL 'pot' fills more slowly. Furthermore, the CIL pot is intended to fund not only this key piece of highways infrastructure but also the primary school and the recreational facilities, noting that these are also identified as key infrastructure projects on the Council's CIL Regulation 123 List.
- 6.3.20 A simplistic example of the CIL that could be generated from this development is provided below, but note that this is based on average gross internal floor area calculated from the Phase 1 proposal and cannot be relied upon for accuracy.

CIL with 40% Affordable Housing

Total Units: 900
60% Private Market Units: 540 Units
Average m²/Unit: 91.79
CIL Rate/m²: £55.00
Total CIL Contribution: £2,726,163

CIL with 30% Affordable Housing

Total Units: 900
70% Private Market Units: 630 Units
Average m²/Unit: 91.79
CIL Rate/m²: £55.00
Total CIL contribution: £3,180,524

CIL with 20% Affordable Housing

Total Units: 900
80% Private Market Units: 720 Units
Average m²/Unit: 91.79
CIL Rate/m²: £55.00
Total CIL contribution: £3,634,884

CIL with 10% Affordable Housing

Total Units: 900
70% Private Market Units: 810 Units
Average m²/Unit: 91.79
CIL Rate/m²: £55.00
Total CIL contribution: £4, 089,244

- 6.3.21 The figures shown above, are indicative only but are sufficient to aid the understanding of what can be achieved from these developers towards a £12.3 million road and a £5 million primary school, not to mention the costs for the proposed Sports Village.
- 6.3.22 In recognising the benefits of the road to Northallerton as a whole and not just for this development, the Council has secured Growth Deal Funding from the Local

Enterprise Partnership to the value of £6 million towards the cost of building the road and bridge to try to ensure that the road and bridge can be built in the first phase of development. However, this is time limited and must be spent on the building of the road and bridge by March 2017. The applicants have also agreed to provide the land and deliver the road and bridge in lieu of a financial receipt within the first phase (this is known as an infrastructure payment) and this has been agreed by the Council, but they need to be able to commence house building at the same time to generate the required cash flow.

- 6.3.23 While there have been suggestions made that the developers are being given a Growth Deal handout to make their development acceptable, it must be reiterated that 495 houses could be built before a link road is deemed to be necessary. Furthermore, the link road is identified as a town wide infrastructure improvement project, which therefore means that it should be funded from CIL collected from this development and other permitted developments as well. The Growth Deal funding is therefore a means to bring this infrastructure forward not to reduce what the developers have to pay.
- 6.3.24 In fact it is important to highlight that pursuant to Section 73A(3) of the Community Infrastructure Levy (Amendment) Regulations 2014, where CIL is paid by way of an infrastructure payment the amount of CIL paid is an amount equal to the value of the infrastructure provided. As the value of the infrastructure provided by the developers in this case (the balance of £6.3 million) would be greater than the CIL receipts illustrated in section 6.3.20 above, the developer would actually be providing a contribution greater than the Council's CIL Charging Schedule requires.
- 6.3.25 Notwithstanding the provision of the link road, various other highway improvements are required to make the development acceptable, such as improvements to the existing High Street/Quaker Lane mini roundabout junction upon occupation of the 50th dwelling; improvements to the existing Brompton Road/Northallerton Road/Stokesley Road mini roundabout upon occupation of the 50th dwelling; and the conversion of the mini roundabout at the eastern end of Quaker Lane with Brompton Road to a priority T Junction within 6 months of opening the link road to through traffic. These would be required by a planning condition on any planning permission granted.
- 6.3.26 With these measures in place, with the provision of bus stops along the link road to provide bus companies with circular routes around the town and most residents access within an approximate 400 m walk distance, and with the provision of footpaths and a cycleway to provide sustainable transport links to and from the town centre as well as through the site, the proposal is considered to provide excellent accessibility by car, public transport and foot and cycle to a vast range of services, facilities and employment opportunities without causing unreasonable detriment to the surrounding area. The proposal is therefore considered to accord with Local Plan Policies CP2 and DP3 that aim to encourage a variety of transport options and improvements in accessibility and the provisions of Paragraph 32 of the NPPF.
- 6.3.27 In terms of the delivery of the link road and bridge the County Council's preference is for the developers to enter into a Section 278 Agreement, which would dedicate the land as highway land at an early stage. However, Regulation 12(a) and 12(b) of the Community Infrastructure Levy (Amendment) Regulations 2014 now restricts the use of highway agreements under Section 278 of the Highways Act 1980 and does not allow the local planning authority to use its planning powers to require a developer to enter into such an agreement in relation to infrastructure that the charging authority intends to fund through CIL. In addition, the aforementioned Regulations also prevents to local planning authority from attaching a 'Grampian' style condition preventing or restricting the carrying out of development until a 'highway agreement' has been entered into.
- 6.3.28 Notwithstanding that, lawyers for the applicants, the Council and NYCC have agreed that the delivery of the road and bridge can take place under a Section 38 Agreement, which means that rather than the County adopting the land for highways prior to the commencement of the road building, the developers ask County highways to adopt it once it has been constructed. The two processes are not dissimilar in that the works must be constructed to a design and standard agreed by the highway authority, the developer will be responsible for carrying out the works at their own expense, and for

- maintenance costs until adoption and similarly a bond will be required between the developer and the Local Highway Authority to ensure that the proposed works can be satisfactorily completed in the event of any default or unforeseen occurrence.
- 6.3.29 As previously noted, the delivery timelines will be specified as part of a Section 106 Legal Agreement with the landowners and the Council, which will be completed post planning committee but prior to any planning decision being issued.

6.4 Affordable Housing Provision

- 6.4.1 Local Plan Policy CP9 states that housing development of 15 or more dwellings (or sites of 0.5 ha or more) in service centres must make provision for affordable housing. It clarifies that the LDF seeks to achieve 40% affordable housing in Northallerton, although it also acknowledges that the actual provision on individual sites will be determined through negotiations, taking into account viability and the economics of provision. This is consistent with Paragraph 50 of the NPPF which states that where local planning authorities have identified that affordable housing is needed, it should set policies for meeting this need on site. However, it then adds that such policies should be sufficiently flexible to take account of changing market conditions over time.
- 6.4.2 Local Plan Policy DP13 provides further advice on the provision of housing and states that housing proposals must provide for a mix of dwellings, in terms of size, type and tenure, which meets the needs of all sections of the local community, promotes sustainable communities and social cohesion; while Policy DP15 provides specifics on the key definitions and principles on which the provision of affordable housing will be achieved.
- 6.4.3 These policies are further supplemented by the Council's Affordable Housing Supplementary Planning Document (adopted April 2015), which states that there is a need for affordable housing across all of Hambleton as evidenced in the November 2011 North Yorkshire Strategic Housing Market Assessment (SHMA) which identified a need for a minimum of 1,600 additional affordable homes across Hambleton from April 2011 to March 2016, or 320 per year.
- 6.4.4 The application was submitted with no identified amount of affordable housing that could be provided on this site and objections were received stating that the limited provision of affordable housing should be resisted, because if demand for housing does exist, it is surely likely to be in the affordable housing bracket. During the course of the application period the applicant has undertaken a viability assessment, which the Council has referred to the District Valuer for consideration. This will be discussed following clarification on the viability process.
- 6.4.5 For the readers of this report who are not familiar with viability appraisals they are a financial test of the ability of a development project to meet its costs including the cost of planning obligations (such as affordable housing, or highway improvements), whilst ensuring an appropriate site value for the landowner and a market return to the developer in delivering that project. There are a variety of items that form part of a viability appraisal and these include, but are not limited to land purchase costs, normal build costs, abnormal build costs (such as land raising to overcome groundwater issues), house sales values, professional fees, developer profit, planning obligations and CIL financial requirements.
- 6.4.6 Where planning obligation liabilities (such as the provision of 40% affordable housing) reduce the site value to the landowner and return to the developer below an appropriate level, the development is regarded as unviable and the land will not be released and/or development will not take place. Paragraph 205 of the NPPF states that where obligations are being sought or revised, local planning authorities should take account of changes in market conditions over time and, wherever appropriate, be sufficiently flexible to prevent planned development being stalled.
- 6.4.7 In such situations, and given the Government's requirement for local planning authorities to boost significantly the supply of housing and maintain a 5 year land supply for delivery of that required housing, the Council is left with a dilemma as to whether to simply refuse permission, or whether it should recognise that the benefits of the scheme coming forward even without a particular contribution outweigh the need to secure that contribution? It is clear from national policy, numerous appeal decisions, and the Courts, that viability is an important question in this context.

- 6.4.8 The applicants submitted various draft viability appraisals during the course of the application, however, the final assessment was submitted on 5th October 2015. Based upon their calculations, the development can deliver between 11.2% and 12.3% affordable housing. The District Valuer (DV) provided his response on 9th October 2015 and concluded that the proposal can deliver between 16.78% and 18.22% affordable housing. For clarity, the reason that both parties have provided two figures is because both have assessed viability firstly with the inclusion of the NYCC Highways request for a commuted sum of £968,000 for bridge maintenance (as noted in Section 5.2.1 above) and secondly without that sum included.
- 6.4.9 Also included in the appraisals are the contributions sought by NYCC Highways for bus pump priming (£500,000) and garden sheds to house bicycles in the 40% of houses that do not have a garage (£273,600).
- 6.4.10 The two key areas of difference between the developers and DV are land values and sales values, although the Council has questioned the figure proposed by the developers for abnormal foundations costs to the west side of the Site, which equates to approximately £2,900,000. The Council has engaged Mouchel engineering consultancy to verify the requirement for and validity of the abnormal foundation costs, but at the time of writing this report their response was not available. It is therefore recommended that the Planning Committee delegate authority to the Executive Director to agree the final contribution in consultation with the Chairman of the Planning Committee.

Viability Differences

	Developers	District Valuer	Difference in Affordable Housing Provision
Land Value	£210,000/acre	£170,000/acre	3.74% / 34 houses
Sales Value	£215/sq ft	£212/sq ft	1.8% / 16 houses
Affordable Housing Contribution	11.2% with £968k commuted sum; 12.3% without £968k commuted sum	16.78% with £968k commuted sum; 18.22% without £968k commuted sum	

6.4.11 In determining land values, the Government's Planning Practice Guidance on Viability states that Central to the consideration of viability is the assessment of land or site value. The most appropriate way to assess land or site value will vary from case to case but there are common principles which should be reflected.

In all cases, land or site value should:

- Reflect policy requirements and planning obligations and, where applicable, any Community Infrastructure Levy charge;
- Provide a competitive return to willing developers and land owners (including equity resulting from those wanting to build their own homes); and
- Be informed by comparable, market-based evidence wherever possible. Where
 transacted bids are significantly above the market norm, they should not be used
 as part of this exercise.
- 6.4.12 The developers were asked to provide evidence of how they had calculated land values and their solicitors Walker Morris provided hard copies of the options agreements in place with the landowners, which demonstrated that £210k/acre is the minimum land value that the landowners are willing to sell for. The developers also advise that this figure has been equalised across the site because, while it is recognised that some parcels of land, such as that for the sports village would have a

- significantly lower land value than the residential land for example, each landowner must have the same incentive to sell if the site as masterplanned is to be delivered as a whole. There is merit to this argument.
- 6.4.13 The DV on the other hand, has based his land value figure on benchmark land values from low, medium and high value land in and around Yorkshire, as well as from the wider area of Durham, Leicester and North East Lincolnshire for example. He considers that having assessed the evidence, the figure proposed by the consortium (equivalent to £210,000 per gross acre), is in excess of his expectations for the purposes of a benchmark land value. Whilst he acknowledge that £210k/acre appears to be the price agreed for the land, he reiterates that he does not believe this is appropriate as a benchmark land value, as it appears to be above the market average for this type of land is this type of location.
- 6.4.14 It is difficult to reconcile the two positions because while Planning Practice Guidance states that land value should be based on market based evidence, it also states that landowners will want a competitive return, which in reality is the minimum value that they are willing to sell their land for. While there is merit in the cases put forward by both parties, it is considered that the Council would be taking a very large risk in accepting the DV's land value over that of the developers' and trying to impose that lower land value on the landowners, for a landowner is under no obligation to sell his/her land and it risks the development site not becoming available or only available in parts.
- 6.4.15 With regard to the difference in sales values, the DV originally advised that in his opinion the average price per sq ft was £220, whereas the developers proposed a figure of £210. Both parties were asked to provide evidence to support their positions and the developers engaged DTZ Property Consultants to undertake their appraisal. Both parties submitted a report to the Council providing evidence from comparable development schemes, taking into account the type of property, the location and delivery. While the parties have not managed to find common ground the DV's valuation dropped and the developer's valuation rose. On the basis that the parties did not draw their information from exactly the same sources and some of the different figures sourced may or may not have taken into account incentives, which are often offered by developers to attract new purchasers, it is again difficult to reconcile the two positions. However, the developers have advised in no uncertain terms that they are not prepared to increase their anticipated sales revenues. They consider that to act against the advice of their independent expert would not only question their integrity as an independent source, but that also the developers would have difficulty justifying it to their board members.
- 6.4.16 Notwithstanding the above, the developers have made an offer to the Council of the provision of 15% affordable housing (i.e. 135 dwellings) subject to the Commuted Sum, Bus Contribution and Cycle Storage provisions being withdrawn by NYCC and there being no requirement to make up the shortfall in funding for the primary school (as identified in Paragraph 5.2.2 above).
- 6.4.17 Since NYCC issued its consultation responses, it has agreed to retract their requests for maintenance payments for the bridge embankments and culverts (£968,000) and Travel Plan Measures (£773,600). However, NYCC is insistent that the shortfall in funding for the new primary school (£950,000) must be met in order to provide the school, which is an essential piece of infrastructure. The shortfall in funding is very likely to come from the funding provided towards affordable housing, which would reduce the level below 15%. Given the importance of the school to the development it is recommended that planning committee delegate authority to the Executive Director, in consultation with the Chairman of Planning Committee, to negotiate further with the developers.
- 6.4.18 What remains clear from the costs and revenues appraisals undertaken by the developers and the DV on behalf of the Council is that viability is a significant issue for this site and that based on current market conditions and infrastructure requirements it is highly likely that any figure agreed on now would result in an affordable housing provision well below the 40% target.
- 6.4.19 However, whilst a site of this size represents a significant challenge due to the scale of the costs involved, it also offers the Council the opportunity not available on smaller

- schemes to seek a viability review at a later phase(s) such that if viability were to improve as a result of market conditions this can be captured to get closer to the target affordable provision and is not taken as an enlarged profit margin by the developer.
- 6.4.20 Indeed, the Homes and Communities Agency's Good Practice Note on 'Investment and Planning Obligations: Responding to the Downturn' suggests that with phased development affordable housing could be deferred to later phases, or the percentage contribution could be lower in phase one and then subject to viability appraisal in respect of later phases. The HCA make clear that they think a planning obligation which allows phased contributions with further viability testing at later stages, is acceptable. A viability review mechanism would therefore be required through terms of a Section 106 Legal Agreement.
- 6.4.21 Of course though, it must also be appreciated that if market conditions worsen during the course of the development (likely to be a build period of 10 years) then a lower affordable housing provision may have to be agreed.
- 6.4.22 In this regard while it may seem sensible to finalise the contribution now to provide 'certainty' and guard against reductions in provision if the market drops later, unfortunately current Government guidance (Paragraph 205 of the NPPF) does not support this. Indeed, the Growth and Infrastructure Act 2013 introduced specific provisions to allow modification or discharge of affordable housing requirements to ensure that developments are not stalled in such situations and require Councils to be flexible through variations in agreed provision levels to ensure that development can proceed. Furthermore, the Act does not set out provisions for Councils to seek the same re-evaluation if it has a fixed provision and the market improves.
- 6.4.23 It is therefore recommended that the Planning Committee agree to delegate authority to the Executive Director in consultation with the Chairman of Planning Committee to negotiate the contribution following a Section 106 Requirement for viability reviews to be undertaken on a phased basis throughout the course of the development, ascertaining the correct level of provision for each phase as it comes forward based on the prevailing market conditions at that time. However, what might not be achievable by taking this approach is a consistent spread of affordable homes across the whole site, particularly if the later phases prove more viable resulting in more affordables in one area. This is contrary to Local Plan Policy DP13 to provide for a mix of dwellings and promote sustainable communities; however, is arguably worth varying if affordable housing needs continue to be so pressing in the future.
- 6.4.24 In conclusion, and for the avoidance of doubt, the provision of up to 15% affordable housing is put forward by the applicants, but this is premised on the development not having to meet the Education shortfall of £950,000. It is considered that based on the viability information to date that this is a reasonable provision at the present time. Notwithstanding, it is recommended that the Planning Committee delegate authority to the Executive Director, in consultation with the Chairman of Planning Committee, to continue to negotiate the figure, which will form part of a Section 106 Legal Agreement to be completed before any planning decision is issued.
- 6.4.25 Furthermore, if the committee agree to approve a review mechanism through the provision on a Section 106 Agreement then it accepts that an absolute figure for final affordable housing delivery will not be known at the time of the grant of consent. This is considered to be a reasonable and flexible approach that reflects government guidance to ensure that the development is viable and can proceed.
- 6.4.26 In the circumstances, it is considered that there is no proper policy basis upon which to found a refusal on the basis of the affordable housing offer, for while the 40% requirement of Local Plan Policy CP9 cannot be met, the provision of some affordable housing would be a benefit that carries a small amount of weight in favour of permitting the development.

6.5 Drainage and Flooding

6.5.1 There have been numerous responses received highlighting the potential for the Development to worsen flooding in the surrounding area and suggesting that foul water capacity is unavailable for a scheme of this size. It is not surprising that flooding issues have been raised by the public and consultees as a cause for concern for the

development of this site, particularly given the watercourses that flow through the site, the evidenced flooding of Brompton, and the location of Flood Zones 2 and 3 adjacent to and partly within the site itself. However, Schedule 3, Part 7 of the Water and Flood Management Act 2010 requires that construction work which has drainage implications may not be commenced unless a drainage system for the work has been approved by the approving authority, and from April 2015 local planning policies and decisions on planning applications relating to major development are required to ensure that sustainable drainage systems (SUDS) are used for the management of surface water.⁴

- 6.5.2 While the LDF was adopted before the aforementioned legislation, its policies are nonetheless in alignment. Specifically, Policy CP21 states that proposals must take particular account of the need to ensure protection from, and not worsen the potential for flooding; Policy DP6 on utilities and infrastructure seeks to ensure new development is capable of being accommodated by existing or planned services and clearly states that new development must not have a seriously harmful impact on existing systems; while Policy DP43 outlines that development will not be permitted if it would have an adverse effect on watercourses or increase the risk of flooding elsewhere and if there is a high risk of the development itself being affected by flooding.
- 6.5.3 The applicant has submitted a Flood Risk Assessment (which was subsequently updated in light of statutory consultee comments), several reports outlining its provision of SUDS (which are now a mandatory requirement), and geo environmental reports which highlight ground water issues on the east of the site which necessitate the need to land raise before it can accommodate new housing. All of this detail has been provided to the relevant statutory consultees (Environment Agency, Internal Drainage Board, NYCC Lead Flood Risk Authority and Yorkshire Water) and those consultees have reviewed each others' responses to try to provide a consistent response. Furthermore, Council's own drainage expert has reviewed all of the relevant flood and drainage documents along with the consultation responses and provided an overarching response that will be referred to in this section as necessary.
- 6.5.4 It is important to highlight that the development area (i.e. the actual land being utilised by buildings and roads) is wholly located in flood zone 1 (that of the lowest flood risk) and only the embankments of the bridge would sit within the area of flood zone 3 (that of highest flood risk) with the approach roads elevated above ground level and the flood plain. The volume of water lost within the embankment footprint in flood zone 3 is required to be compensated for elsewhere at a similar level to the volume lost in order that the flood characteristics adjacent to the bridge are not altered.
- 6.5.5 As identified in the Flood Risk Assessment, the site boundary is such that the available land for compensatory storage is constrained, particularly on the eastern side of the rail crossing. However, through the use of engineered bridge embankments, as opposed to traditional earth embankments, the land required to accommodate the necessary flood compensation storage is available in three discrete areas (see Plan No. 4033-FRA05, prepared by iD Civils Design and found at Appendix L of the Flood Risk Assessment) and may provide an additional 10% volume. In addition, and in order to avoid any flooding of the railway line, it is proposed that large box section culverts are constructed in the embankments to maintain the natural flow route south. This method has been ratified by the Environment Agency.
- 6.5.6 With regard to surface water, as noted above, the developer is required to consider sustainable drainage techniques for the new development and to ensure that SUDS are incorporated into the development. As highlighted in the NYCC SUDS Design Guidance, surface water runoff not collected for use must be discharged to one or more of the following in the order of priority shown:
 - a) Discharge into the ground (infiltration).
 - b) Discharge to a surface water body.
 - c) Discharge to a surface water sewer, highway drain or other drain.

See https://www.gov.uk/government/speeches/sustainable-drainage-systems. For an explanation of 'major' development please refer to Article 2(1) of the Town and Country Planning (Development Management Procedure) (England) Order 2015.

- d) Discharge to combined sewer.
- 6.5.7 Furthermore, the peak runoff rate from the developed site for the 1 in 1, 1 in 30 and 1 in 100 year rainfall events to include for climate change and urban creep must not exceed the peak greenfield runoff rate from the site for the same event. Greenfield runoff rate is maximum 1.4 l/s/ha unless catchment modelling demonstrates greenfield runoff to be greater than this.
- 6.5.8 With regard to the Darlington Road (west) part of the scheme geo-technical surveys have revealed that the site is underlain by a mixture of soft/firm clays with groundwater at cc 2.3m below ground. The natural clays below the site will not provide soakaway rates sufficiently high to enable the use of infiltration drainage and therefore this technique has been avoided.
- 6.5.9 Instead it is proposed that the surface water from Phase 1 West will be discharged to the Sheepcote Stell watercourse which falls within the Swale and Ure Drainage Board's designated area. Flows will be attenuated by means of a detention basin with a flow restriction device at the outfall to the watercourse. The discharge rate will be based on 1.4 litres/second/hectare as agreed by the Environment Agency, NYCC, Swale and Ure Drainage Board and HDC's own flood advisor.
- 6.5.10 As identified in the submitted SUDS and Drainage Strategy prepared by iD Civils Design for the Darlington Road scheme, and specifically on Plan No. 4377-C-D7-01 in Appendix C of the iD Civils SUDS Report, the detention basin is proposed to be located to the east of the Phase 1 Persimmon Scheme and north of the indicative location for the primary school. The capacity of the basin would accommodate up to 1 in 100 year (plus 30% climate change) event and provide capacity for Phase 2. As discussed in the Flood Risk Assessment, Phase 1 West and the later phase(s) to the north will be drained via a single outfall and the network needs to be designed to enable provision for Phase 1 independently but so that minimal work is required to modify the Phase 1 drainage to enable the later phases to be accommodated.
- 6.5.11 Yorkshire Water confirms that it can adopt the maintenance of the SUDS detention basin under a Section 104 agreement as long as they are satisfied with the design. This will therefore be subject to planning conditions and other legislative requirements. The Council's flood and drainage officer has confirmed that the arrangements for managing surface water for the Darlington Road (West) Scheme is satisfactory. Therefore compliance with Policies DP6 and DP43 is achieved.
- 6.5.12 With regard to the Stokesley/Northallerton Road (east) part of the scheme geotechnical surveys have revealed that the site is underlain by a mixture of soft/firm clays, silts and granular material with a water table at cc 1.5m depth. The variable nature of the natural ground below this part of the site and the high water table will preclude the use of infiltration drainage as a means of surface water disposal.
- 6.5.13 Instead it is proposed that the surface water from Phase 1 East will be discharged to the unnamed watercourse located between Stokesley and Northallerton Roads, which falls within the Swale and Ure Drainage Board's designated area. Flows will be attenuated by means of four underground attenuation tanks, each will have a flow restriction device at the outfall to the watercourse. The discharge rate will be based on 1.4 litres/second/hectare as agreed by the Environment Agency, NYCC, Swale and Ure Drainage Board and HDC's own flood advisor, which in practice will be 5 litres per second per tank as this is the technical minimum that can be achieved with flow control devices and still be appropriate for adoption. It is important to note, however (particularly with regard to concerns of off-site flooding from land raising), that the natural greenfield rate identified in the Institute of Hydrology '124 Report', which is the Environment Agency recommended method for a site of this size, has been calculated as 3.6 litres/second/hectare. Therefore, the adoption of the NYCC/IDB rate of 1.4 l/s/ha therefore represents a 60% reduction on the current greenfield rate. Furthermore, due to the granular nature of the existing sub soils on the east, the Flood Risk Assessment identifies that water would not naturally pond on the site, but would percolate to the watercourse via natural groundwater flows.
- 6.5.14 NYCC SUDS officer originally stated a preference for the use of ponds over tanks, as tanks don't provide the same type quality water treatment. However, due to high land take and safety issues of ponds within housing developments and because contamination from housing sites is not likely to be an issue (noting that there is no

- commercial development within the eastern site) NYCC accepts the benefits of underground tanks. As a result, a revised response was received confirming that underground storage tanks would be suitable.
- 6.5.15 As identified in the submitted SUDS and Drainage Strategy prepared by iD Civils Design for the Stokesley/Northalleron Roads scheme, and specifically on Plan No. 3525-C-D9-01 in Appendix C of the iD Civils SUDS Report, two tanks would be provided on the north side of the link road and two on the south side within the housing development layout. The land above ground would be landscaped so that the spaces appear as green space. The capacity of the tanks would accommodate the 1 in 100 year plus climate change event and would be adoptable by Yorkshire Water under a Section 104 agreement.
- 6.5.16 The Council's flood and drainage officer has confirmed that the arrangements for managing surface water for the Stokesley/Northallerton Roads (east) Scheme is satisfactory, although as the site will be elevated above the level of Stokesley Road the submission of final design will be required showing that any displaced water can be channelled through safe exceedance routes and discharge to the watercourse. The IDB has similarly suggested a planning condition to this effect, which is included in Section 7 of this report.
- 6.5.17 Extensive consultation has taken place with the relevant flood and drainage authorities (as identified in Section 5) and in summary it has been concluded that the development can be appropriately designed to comply with LDF Policies, National Guidance and Legislation on flood prevention, although some further detail, which will be required by Condition, will need to be endorsed. While there is strong feeling and frustration from the local community (specifically the Brompton Flood Prevention Group), in relation to a lack of remedy provided by the regulatory bodies to date, there is no evidence to demonstrate that the development would exacerbate this existing problem. The proposal would not therefore conflict with Policies CP21 and DP43, both of which are in compliance with the NPPF and which seek to ensure that the development does not have an adverse impact on flooding. This matter does not therefore weigh in favour of a refusal of the application.
- 6.5.18 Specifically in relation to foul water, as identified in Section 5.2.9 and 5.2.28 of this Report, Yorkshire Water has advised that there is capacity for Phase 1 of the development but not sufficient capacity for the later phases. However, they are under a statutory duty to ensure that there will be sufficient capacity and have to work with the developers to plan the timing of foul water upgrade works. Yorkshire Water confirms that the capacity issue is not with the Waste Water Treatment Works but with the sewer network. The proposal can therefore comply with DP6 because it is capable of being accommodated by existing or planned services.

6.6 Nature Conservation Impacts

- 6.6.1 LDF Policy CP16 states that developments will be supported where they preserve and enhance the District's natural (and man made) assets, while Policy DP31 is worded in the negative and states that permission will not be granted for development which would cause significant harm to sites and habitats of nature conservation, geological or geomorphological value, together with species that are protected or under threat. While these policies pre-date the NPPF, they accord with its provisions, noting that the NPPF focuses on achieving sustainable development which should 'contribute to protecting and enhancing our natural ... environment' and 'help to improve biodiversity (Para 7), by 'moving from a net loss of biodiversity to achieving net gains for nature' (Para 9). Indeed, conserving and enhancing the natural environment is one of the 12 core planning policies.
- 6.6.2 As part of the planning submission, Chapter 8 of the Environmental Statement identified the extent of the ecological assessment, the procedures followed, the surveys undertaken, the significance of the impact of the proposed development on the biodiversity of the site, and any mitigation required. The Chapter was supported by more detailed ecological survey reports, which identified the likely absence of the following protected species: badgers, great crested newt, reptiles, water vole and schedule 1 birds. The submission noted that there are no designated sites within or bordering the Development and concluded that current and recent use of the site means that the majority of the Site (agricultural grassland and arable) presents limited

ecological value; however, there is potential for the proposed development to impact on bats, nesting birds and over wintering birds. The mitigation proposed included precommencement bat roost surveys and vegetation clearance between September to February to avoid nesting season (March to August).

As noted in Section 5 of this report, the Yorkshire Wildlife Trust, County Ecologist and Natural England were all consulted and it was recommended that the Council seek independent advice on the content of the ecological submission as well as on the suggestion that bat surveys could be undertaken at a later period in time; the latter apparently being contrary to best practice to fully consider the impact on European Protected Species. The Council therefore engaged Arcus Consulting to review the material and their response is provided below in *italics*.

The aim of the review is to establish whether or not the submitted ecological information provides a robust and accurate assessment of the potential ecological effects of the Development.

With regards to habitats and vegetation the Phase1 Habitat Survey was undertaken in accordance with best practice and the extent and timing of the survey was appropriate. The valuation and assessment of impacts is correct and justified.

With regard to Great Crested Newt, it was noted that the desk study results returned no records of the species and it is unlikely that further surveys would qualitatively alter the conclusions of the assessment. GCN's have therefore been appropriately scoped out of the assessment.

Turning to Otter and Water Vole, surveys were undertaken in accordance with best practice and extent and timing of the surveys were appropriate. The reported valuation and assessment of impacts is correct and justified; limited and generally unlikely.

In relation to badgers, surveys were also undertaken in accordance with best practice and extent and timing of the surveys were appropriate. The results were presented clearly and there are no records of badger activity or evidence. Therefore badgers were appropriately scoped out of the assessment of ecological effects but addressed in terms of legal compliance and so preconstruction surveys have been recommended.

With regard to invertebrates, predominately agricultural habitats are unlikely to support invertebrate assemblages of note and so it is appropriate to exclude invertebrates from detailed assessment. However, this was not justified in the report or ES submitted with the application. Notwithstanding, measures to enhance invertebrates should be included in the recommended Ecological Enhancement Management Plan. Measures could focus on Local BAP priorities and should be delivered through habitat enhancement and features such as nest boxes and log piles.

In relation to birds, the winter bird surveys were undertaken in accordance with best practice guidance and the extent and timing of winter bird survey considered appropriate. However, it has been identified that no walkover route maps have been presented for winter season surveys so it is not clear if the "observation frequency" map is skewed by observer coverage.

Turning to breeding birds, the surveys were not undertaken in accordance with best practice, with the extent and timing of the survey being sub-optimal. Surveys should have commenced in April rather than late May and a minimum of four visits should have been carried out for a site such as this. By commencing surveys in late May valuable evidence of breeding attempts by some key farmland species (e.g. lapwing and skylark) may have been missed. Breeding surveys should have followed the Common Bird Census (CBC) methodology, with observers walking within 50 m of all habitats. As it stands it is unclear whether surveyors walked within fields or just along field edges because no walkover route map is provided, although the text suggests they did not. It is not clear if the "observation frequency" map is therefore skewed by observer coverage. Breeding territories of Birds of Conservation Concern (BoCC) should

be provided to allow assessment. The use of "observer frequency" map is unsuitable.

Furthermore, a Barn owl survey is recommended in the Preliminary Ecological Assessment (PEA), however it is not clear if this was completed, as it is not mentioned within reports other than the PEA. Failure to complete a barn owl survey could lead to legal offences if birds are breeding and disturbed during the breeding season by construction works.

It is therefore considered that the ES does not contain all of the information needed to fully assess the impacts of the development on breeding and wintering birds (e.g. the observation frequency map is not suitable) ... The number and location of pairs of breeding birds is required to be able to assess these effects, but this information is not presented. The main effect of the development is permanent habitat loss. If, post-construction, there is no habitat for the birds, they will be displaced. Although some species will move into gardens (e.g. house and tree sparrow and starling) others will be permanently displaced (e.g. lapwing and skylark). We disagree with the operational assessment changing from moderate adverse to moderate positive over time. The development will displace birds (including Birds of Conservation Concern) and some of these species will not move into gardens/habitat referenced in the proposed Ecological Enhancement Management Plan. The information is not quantified and as a result it is not clear if the impacts will be significant in the local context. Consequently, it is not clear whether or not the proposed mitigation and enhancements will be suitable for the species involved.

Greater clarification on survey effort and walkover route needs to be provided. A breeding bird survey following the CBC methodology (with at least four visits) may need to be repeated during the 2016 breeding season commencing in early-April (depending on the above clarifications sought).

Lastly, turning to bats, two different elements of bat survey and assessment were undertaken to inform the baseline. For clarity, these are addressed in turn. It is essential to highlight though that bats are a European Protected Species and under Regulation 39 of the Habitats Regulations, subject to certain defences or in the absence of a licence (Regulations 40 and 44), it is unlawful to:

- a) Deliberately kill; or
- b) Deliberately capture; or
- c) Deliberately disturb; or
- d) Deliberately take or destroy the eggs of a wild animal of a European protected species; or to
- e) Damage or destroy a breeding site or resting place of a wild animal of a European protected species.

Bat Roosts

A requirement for the assessment of trees and buildings for their suitability to support roosting bats was identified during the PEA. The assessment and categorisation of roosts was undertaken in line with best practice guidance. The recommendation for further survey — activity surveys comprising roost presence/likely absence surveys — of the features with moderate potential to support roosting bats were in line with best practice. However, recommendations for further roost activity surveys on features with low to moderate roost potential (buildings in Sheepcote Close and Hartington House) were not in line with best practice. Several buildings scheduled for demolition were assessed as having a low potential for roosting bats, however further activity surveys were not recommended, contrary to Bat Conservation Trust guidance, which recommends a single emergence survey. This deviation from guidance is not explained. Roost activity surveys at any of the features identified were not undertaken and no justification for their omission is provided.

The scope of work was therefore not in line with prevailing best practice and the baseline may be considered incomplete. Therefore in terms of the impact of the development on bats it is considered that an assessment of the loss of potential

bat roosts has been informed by an incomplete baseline. To accommodate this information gap, the authors advise that they have assumed a worst-case scenario that features of moderate roost potential are occupied by common pipistrelle and soprano pipistrelle bats. An assessment based on the assumed presence of a European Protected Species is neither appropriate nor presents a worst-case scenario. The assessment of effects on roosting bats is therefore not in line with best practice.

It is stated in the ES chapter that roost presence/likely absence surveys will be undertaken prior to demolition of buildings or removal of trees. This will however be undertaken post-consent. A recommendation is made in the applicants' reports that surveys of potential roost features should be included as a planning condition.

Surveys for European Protected Species to inform an ecological baseline should not be a condition of a planning consent. To do so would be in contradiction of planning policy and The Conservation of Habitats and Species Regulations.

The Local Authority should not grant consent under the terms of The Conservation of Habitats and Species Regulations and as described in the ODPM circular 06/2005, paragraphs 98 and 99. In particular, paragraph 99 iterates the importance of the requisite information being gathered before planning consent is granted.

Recent case law including Woolley vs Cheshire East Borough Council 2009 and Morge vs Hampshire County Council 2010 have provided clarification on the responsibility of the LPA in discharging its duties in cases involving European Protected Species, the findings from which are applicable in this case.

Roost presence/likely absence surveys should be undertaken and the assessment amended prior to the application being determined. Alternatively, allowances should be made in the development layout to preserve potential roost feature(s) where their removal would require additional information to be gathered.

Clarification should be sought from the consultants where recommendations for further surveys deviate from best practice. Guidance documents are open to interpretation and the appropriate course of action can, and should, be informed by professional experience and judgement. Deviation from the guidance in this instance is in contradiction to the consultants methods as stated in their report(s). Such deviations should therefore be acknowledged within the report with sufficient justification presented.

Foraging and Commuting Bats

An assessment of the habitats on and adjacent to the development area was undertaken during the PEA and subsequent survey to inform the preparation of the Initial Bat Assessment Report (May 2015). The habitat assessment and all subsequent activity surveys were completed in line with best practice. Low levels of activity primarily from common species were identified.

With regard to potential impact, the valuation and assessment of impacts is correct and justified; there would be a moderate adverse effect. Mitigation to reduce likelihood of displacement (due to lighting) and enhancements as part of the Ecological Enhancement Management Plan will adequately address potential impacts and provide potential positive effects.

- 6.6.4 The Arcus Report was provided to the statutory consultees and the applicant for comment.
- 6.6.5 The NYCC Ecologist advised that overall Arcus have provided a very good review of the ecological work undertaken and that she would agree with their recommendations in general and the planning conditions that they propose. However, with regard to birds, while the NYCC Ecologist agreed with Arcus that the applicants' assessment of a moderate positive outcome for species such as lapwing and skylark seems incorrect, she considered that the information provided by the applicant seems to be sufficient to make an assessment on the importance of the site for birds. She advises that these are farmland species that are not likely to be/able to be accommodated

- within the new development. In that regard the applicant could look to provide compensation off site. In any event site clearance and construction must avoid the breeding/nesting season.
- 6.6.6 With regard to bats, the NYCC Ecologist agreed with the Arcus Report and advised that conditioning bat surveys is not recommended and would go against current policy because it would not be possible to make a full assessment of the impact upon bats in accordance with the Conservation of Habitats and Species Regulations 2011 (as amended) 'the three tests'.
- 6.6.7 Yorkshire Wildlife Trust advised that the report by Arcus is thorough and the conclusions reached are correct. They agree with the advice that bat emergence surveys must be carried out prior to determination of the application and at the same that a barn owl survey should be carried out showing the locations of the sites with potential to support nests.
- 6.6.8 With regard to birds, the Trust also agree with Arcus that the breeding bird survey is currently insufficient to enable a full assessment of the potential impacts on the habitat and bird populations on site, as well as the potential effectiveness of the mitigation and compensation measures, noting that the full Common Bird Census (CBC) methodology requires 10 visits between March and July, ideally with at least 10 days between each visit, and mapping of individual bird territory, although a scaled down version of a CBC survey (five visits) would be sufficient here.
- 6.6.9 Natural England provided a response, which again referred the Council to their standing advice for protected species; however, in relation to bats they advised caution in conditioning protected species surveys and, like Arcus, also referred to Para 99 of Government Circular 06/2005. Natural England highlight that the Circular makes clear that the presence and extent of protected species should be established before permission is granted, and that ecological surveys should only be left to coverage under planning conditions in exceptional circumstances. However, Natural England also identify that Para 99 also suggests that developers should only be required to undertake surveys where they are necessary.
- 6.6.10 The applicant has provided a response to the Arcus findings on Birds and Bats, and these will be detailed separately below, with officer comment thereafter.

Birds

- 6.6.11 The applicant states that they did not provide a walkover route map, as neither Natural England, the Planning Inspectorate nor other Local Planning Authorities have requested them on previously consented projects using this methodology. They suggest that "field data in farmland is "naturally" skewed with relatively low densities of birds in open fields and relatively high densities of birds in field boundaries and other habitats with greater structural complexity. For this reason and in accordance with Natural England/Defra guidance on determining survey effort, every major internal hedgerow, drain and field boundary within the site boundary was surveyed as well as a good representative sample of the on-site farmland habitats."
- 6.6.12 Furthermore they suggest that "while Common Bird Census (CBC) is a methodology that can be used there are no formal recommendations for survey methodology in relation to breeding birds (see Natural England/Defra 'Planning and development guidance' on 'Wild birds: surveys and mitigation for development projects'). Three survey visits are therefore considered acceptable."
- 6.6.13 In relation to barn owls, the applicants' ecologist confirmed that "signs of barn owl presence were searched for during the winter bird survey (although not specifically stated in the ES). No evidence of occupation was found in the only two potentially suitable buildings within the survey area and no suitable trees or nest boxes were found."
- 6.6.14 It is noticeable that there is some disagreement between various qualified ecologists on the extent, methodology and reporting of surveys on breeding and nesting birds with the applicants' relying on past practices, which may have been acceptable to

https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/7692/147570.pdf

other planning decision makers, and Arcus and Yorkshire Wildlife Trust on the other hand relying on guidance in the CBC methodology. Indeed, it is a difficult task to balance two conflicting positions, particularly when the applicants' have provided evidence of large scale developments (such as a Nationally Significant Infrastructure Project) where their process of transect routes and observation frequency mapping has been accepted and endorsed, and when the alternative CBC territory mapping is internationally acknowledged as the most efficient and practical way of estimating breeding bird numbers on farmland in small areas.

- 6.6.15 However, in researching the two methods, the planning officer notes that the British Trust for Ornithology, the Joint Nature Conservation Committee and the Royal Society for the Protection of Birds launched a new Breeding Bird Survey methodology in 1994 to use line (aka route) transects rather than the more intensive territory mapping method used by the CBC. The British Trust for Ornithology advises that this allows the surveyor to cover the ground quickly and efficiently recording many birds (noting that the time required for territory mapping can be up to seven times that of transects), limits double counting of birds as the surveyor is continually on the move, suits extensive, open and uniform habitats, and is suited to situations where access is good. On the basis that there are no designated sites within or bordering the development and that transect surveys are a recognised survey methodology, the planning officer considers it reasonable to accept the results in the applicants' breeding bird and over wintering survey reports and considers it would be unreasonable to delay decision making by requiring the applicant to undertake the more onerous CBC territory mapping.
- 6.6.16 Having said that, it must be mentioned that the provision of a transect route map should have been provided, as this might have overcome the concerns of the statutory consultees and Arcus that the results were skewed by surveyor coverage of the site. The Council should therefore consider the provision of a route/walkover map as a requirement for any other application on other sites that requires a breeding bird survey to be submitted.
- 6.6.17 While it is noted that the development will displace birds (including Birds of Conservation Concern) and that some of these species will not move into gardens/habitat, there remains considerable undeveloped agricultural land around the development site that would be attractive to displaced birds. On that basis, it is considered that displacement is not a ground for recommending refusal of the development.

Bats

- 6.6.18 Following the Arcus review the applicant was requested to carry out bat emergence surveys to the buildings in Sheepcote Close and Hartington House, as well as Tree 16 of the arborists report, on the basis of the advice in Para 99 of Government Circular 06/2005 that the presence and extent of the protected species should be established before the development is permitted, and because without that baseline information it was considered difficult to determine if the requirement in Planning Policy DP31 to avoid causing significant harm to protected species could be met.
- 6.6.19 The applicants agreed to undertake the emergence surveys at Hartington House and one dusk emergence and pre-dawn re-entry survey was undertaken on each building in September 2015. It was concluded that no bats were observed emerging from or returning to any of the buildings surveyed. It was therefore recommended that the buildings should be demolished or development works started as soon as practicable, to reduce the risk of bats exploiting what little potential exists and potentially roosting in the intervening period, but that if demolition does not occur by September 2016 that the surveys be repeated.
- 6.6.20 Arcus reviewed the submitted report and confirmed that the methods are in line with best practice and the conclusions of the report are appropriate and commensurate with the results. It is therefore suitable to support the planning application.
- 6.6.21 With regard to Tree 16, the applicants expressed that they were reluctant to undertake emergence surveys because the tree was being retained. However, the tree was identified for felling in the arborists report and so clarification was sought. The applicants provided an updated arborists report on 6 October 2015 clearly identifying the tree would be retained. The applicants' ecologist also advised that

"although Tree T16 has been identified as having definite bat roost potential and classified as a Category 1 tree (Hundt, 2012) it will be retained as part of the development, as it is located in the village green. As the tree will be retained no further surveys for bats are required for this tree. Even if a small roost is present (bat activity surveys undertaken in this area showed that a major bat roost is not present) the current masterplan suggests that it will not be directly impacted and any indirect impacts are likely to be minimal."

- 6.6.22 In relation to the buildings at Sheepcote Close, the applicants were reluctant to provide emergence surveys and offered the following explanation: "The initial bat assessment undertaken at Sheepcote Close in March 2014 assessed the property as a whole as having low potential to support roosting bats. While some of the buildings provide some low to moderate potential, no evidence of roosting bats was identified during the comprehensive internal and external survey. The farmhouse and associated buildings are occupied and maintained by the current owners and as such are unlikely to deteriorate in the foreseeable future. Given that the buildings at Sheepcote Close will not be affected by the development for 5-8 years, during which time the buildings will remain occupied, used and maintained, it seems appropriate to condition a reassessment of the buildings for bats and, if required, emergence and reentry nocturnal surveys prior to commencement of the phase of development which effects these buildings."
- 6.6.23 While Para 99 of Circular 06/2005 states that ecological surveys should only be left to coverage under planning conditions in exceptional circumstances, it must be noted that those buildings are not located in the Phase 1 development area, and also that they are well removed from the Phase 1 development area, such that, even if there were bat roosts within those buildings it is reasonable to conclude that they would most likely be unaffected by the Phase 1 construction. Furthermore, it must also be noted that Para 99 of the said Circular also states (in *italics*) that:

[B]earing in mind the delay and cost that may be involved, developers should not be required to undertake surveys for protected species unless there is a reasonable likelihood of the species being present and affected by the development.

- 6.6.24 With this in mind, and noting the comprehensive photography of the internal spaces of those buildings, it is considered reasonable in this instance to condition a reassessment of the Sheepcote Close buildings for bats and, if required, emergence and re-entry nocturnal surveys, prior to commencement of the phase of development which affects those buildings. It is also noted that the Council has adopted this approach on similar large outline planning applications.
- 6.6.25 Overall, it is considered that no significant harm from the proposed development to nature conservation interests, together with species that are protected or under threat, has been identified and where doubt has been raised by the consultees suitable planning conditions can be attached to any permission granted to ensure that any impact can be mitigated, or, as a last resort, compensated for. In that regard it is considered that the provisions of the NPPF and the provisions of Local Policies CP16 and DP31 to preserve and protect biodiversity are met.

6.7 Noise

- 6.7.1 On the basis that the allocation site lies adjacent to the commercial and industrial premises on Thurston Road, it is not surprising that concerns have been raised by members of the public and Allerton Steel with regard to the noise impacts from the existing businesses on the new land uses, specifically the new dwellings and the school, and the potential for future noise complaints from residents that could lead to restrictions on the otherwise unencumbered operating hours/methods of those businesses. Furthermore, given the construction of a strategic link road around the town and the proximity of many of the new dwellings and the school to that link road, there is potential for the noise from the road to also become a noise nuisance for occupiers of adjacent sensitive buildings.
- 6.7.2 Policy CP1 Sustainable Development, of the Local Plan seeks protection of health, economic and social well-being, amenity and safety of the population, while Policy DP1 states that all development proposals must adequately protect amenity, particularly with regard to privacy, security, noise and disturbance. Furthermore,

- Policy DP44 states that noise sensitive development will not be permitted in areas where potential for harmful noise levels is known to exist.
- 6.7.3 The Government recognises that one aspect of meeting the objective of achieving sustainable development is the need to manage noise. The Noise Policy Statement for England (NPSE) produced in 2010 and now abridged in the NPPF has three aims:
 - Avoid significant adverse impacts on health and quality of life;
 - Mitigate and minimise adverse impacts on health and quality of life; and
 - Where possible, contribute to the improvement of health and quality of life.
- 6.7.4 The NPSE highlights that the broad aim of noise management is to minimise noise 'as far as reasonably practical'. It clarifies that this concept can be found in the Environmental Protection Act 1990, where in some circumstances, there is a defence of 'best practicable means' in summary statutory nuisance proceedings. What is therefore clear from the NPSE is that all reasonable steps should be taken to mitigate and minimise adverse effects of noise but it does not mean that such adverse effects cannot occur.
- 6.7.5 Para 123 of the NPPF states that planning decisions should aim to (amongst other things) recognise that development will often create some noise and existing businesses wanting to develop in continuance of their business should not have unreasonable restrictions put on them because of changes in nearby land uses since they were established. However, this does not mean that because new developments would make or experience some noise or because existing businesses might have to adapt to its new neighbours that the proposal should be regarded negatively.
- 6.7.6 Indeed, Paragraph 002 of PPG on Noise states that while noise can override other planning concerns, neither the NPSE or the NPPF expects noise to be considered in isolation, separately from the economic, social and environmental dimensions of proposed development.
- 6.7.7 Notwithstanding that, Paragraph 006 of PPG on Noise advises that the potential effect of a new residential development being located close to an existing business that gives rise to noise should be carefully considered. This is because existing noise levels from the business even if intermittent may be regarded as unacceptable by the new residents and subject to enforcement action. To help avoid such instances, appropriate mitigation should be considered, including optimising the sound insulation provided by the new development's building envelope. Whilst planning consents should not be subject to conditions that are covered by other legislation, such as the Building Regulations, local planning authorities may include conditions relating to noise and acoustics when granting consent.
- 6.7.8 As noise is a complex technical issue the information prepared by the applicants' acoustic consultants (PDA) and criticised by Allerton Steel's acoustic consultants (Dragonfly) has been reviewed by Council's Environmental Health Team, who are experienced specialists in this area. There has been some dispute between the two consultants regarding the correct standard of noise assessment for the proposed development with PDA having only assessed BS8233: 2014 'Guidance on Sound Insulation and Noise Reduction for Buildings' and Dragonfly insisting that BS4142: 2014 'Methods for Rating and Assessing Industrial and Commercial Sound' should also be considered to determine the level of noise generated by Allerton Steel above background noise levels. However, PDA consider that, as Allerton Steel already exists and their noise is already part of the context for the site, its noise already forms part of the background noise levels.
- 6.7.9 Clause 8.5 of BS4142: 2014 provides legitimacy to PDA's argument, for it states that "where a new noise sensitive receptor is introduced and there is extant industrial and/or commercial sound, it ought to be recognised that the industrial and/or commercial sound forms a component of the existing environment. In such circumstances other guidance in addition to or alternative to this standard can also inform the appropriateness of both introducing a new noise sensitive receptor and the extent of required noise mitigation." This approach was confirmed as acceptable by Environmental Health prior to the submission of the application.

6.7.10 BS8233: 2014 provides the following recommended indoor ambient noise levels for living and dining rooms for daytime use and bedrooms for night time as follows:

Location	07:00 to 23:00	23:00 to 07:00
Living Room	35 L _{Aeq, 16 hour}	-
Dining Room/Area	40 L _{Aeq, 16 hour}	-
Bedroom	35 L _{Aeq, 16 hour}	30 L _{Aeq, 8 hour}

- 6.7.11 For external areas used for amenity such as gardens and patios, it is desirable that noise levels do not exceed 50 dB L_{Aeq, T}, with an upper guideline value of 55 dB L_{Aeq, T}.
- 6.7.12 The EHO has advised that the noise levels reported by PDA (which includes noise from Allerton Steel and the proposed Road) at the dwellings adjacent to the link road are between 56 dB(A) to 64 DB(A), while Dragonfly's report identifies the overall noise impact of between 60 dB(A) and 64 dB(A). The EHO has also advised that noise exposure predictions submitted by both consultants for the Phase 1 residential scheme identifies traffic noise as the dominant noise source during day and night time measurements, regardless of any noise from Allerton Steel.
- 6.7.13 Thus, while the noise predictions far exceed those recommended by BS8233: 2014, they confirm that Allerton Steel would not be the underlying cause, which in some regards demonstrates that the two developments can safely co-exist. Notwithstanding that, there is a requirement for the developer to reduce the internal noise levels of those properties affected. The Noise Contour Maps included within the PDA report are a suitable indicator of where the impact would be felt.
- 6.7.14 PDA, Dragonfly and the EHO all conclude that dwellings on the north side of the link road could be suitably designed and/or orientated to comply with BS8233: 2014. Measures can include all, or some of the following:
 - Revised position / orientation of the buildings in relation to the noise source(s);
 - b) Provision of acoustic barriers;
 - c) Increasing the sound insulation of the building; and /or
 - d) Planning of the interior layout of the building.
- 6.7.15 The EHO considers that the internal noise levels could be achieved with the use of c) and d) alone but the noise levels in the gardens, as identified in PDA's own report, will be above those recommended and, without acoustic barriers, may be unachievable. However, it is noted that the rear gardens to houses will benefit from shielding from the dwellings themselves and fencing, which will bring the noise down to 55 dB $L_{Aeq\ 16}$ hours and below in the majority of cases. It is a commonly accepted benchmark that a physical barrier such as a solid brick wall or specialist acoustic fence that is high enough to block 'the line of sight' between the noise source and the receiver can decrease noise by at least 10 dB. To put this into perspective, motorway traffic produces 70 dB and conversational speech rates at 60 dB, thus 10dB is a really significant reduction.
- 6.7.16 With regard to improving the sound insulation of the dwellings, this is best done with closed, double glazed or secondary glazed windows; but this might also require the installation of mechanical ventilation. While Planning Policy Guidance Note 24 'Noise' has been superseded by the NPPF, its table from Annex 6 clarifies the acoustic properties of different types of glazing, and hence it is reproduced below for demonstration:

Table 1: Typical Noise Reduction of a Dwelling Façade with Windows Set in a Brick/Block Wall

Noise Source	Single Glazing	Thermal Double Glazing	Secondary Glazing
Road Traffic	28	33	34
Civil Aircraft	27	32	35

Military Aircraft	29	35	39
Diesel Train	28	32	35
Electric Train	30	36	41

Note: The values in the Table are the difference between dB(A) levels measured outside and inside typical dwellings; they have not been corrected for reverberation time or window area, and so cannot be compared with values obtained under other conditions. The Table is intended to give an idea of the insulation likely to be achieved in practice - not under ideal conditions. Secondary glazing systems in particular will perform better in installations where sound insulation is not limited by poor sealing or by flanking sound paths such as through doors or acoustically weak parts of window bays. The values for single glazing are representative of well sealed windows.

- 6.7.17 Because the sound insulation of a window (and other components of the building envelope) varies with the frequency (or pitch) of the sound, the overall noise reduction provided by a window will depend, among other factors, on the spectrum of external noise. The table above therefore shows typical reductions in noise levels from common sources which would be expected from various types of window installations fitted in brick/block walls in a dwelling. The insulation provided by any type of window when partially open will be in the region of 10 15 db(A).
- 6.7.18 Clearly, the detailing of which glazing specification would be required for each window is an item that belongs in the design-stage considerations and will depend on the location of the dwelling *vis a vis* the road. However, the achievement of suitable internal noise levels should be the subject of a Condition on any Planning Permission granted.
- 6.7.19 With regard to the later development phases and the indication on the pre-application Masterplan that was displayed at the public consultation event of dwellings to the south side of the link road and to the east of Allerton Steel, the EHO considers that those dwellings would be significantly affected by noise from Allerton Steel and so it is considered that it would be inappropriate to grant permission for dwellings in this location. While the indicative Masterplan submitted with the application does not show dwellings in this location, it was raised as a matter of concern by Allerton Steel in their objection to the application, and so for the avoidance of dwellings being proposed at a reserved matters stage, a planning condition should be included on any permission granted specifically prohibiting residential development in the area adjacent to Allerton Steel labelled 'to be determined' on the plan identified as Illustrative Masterplan Scaled, Nod. 0000-0019 Rev A, dated May 2015, and produced by Spawforths.
- 6.7.20 With regard to the suggestion of a care home to the west of Allerton Steel, the EHO considers that it would also be exposed to similar noise levels as dwellings to the east. However, it is considered that there is probably greater scope in the construction of a care home to ensure that internal noise levels and outside amenity areas, protected by the building itself, can be achieved. However, as this would require the submission of a reserved matters application, it is considered that that application would need to be submitted with a revised acoustic assessment and mitigation plan to demonstrate that appropriate ambient noise levels can be achieved. This should form a planning condition on any permission granted.
- 6.7.21 With regard to the outline proposal for a primary school on the north side of the link road, it must be identified that the criteria for acoustic conditions within schools is set out within Building Bulletin 93 (BB93), which requires that ambient noise levels within primary and secondary school classrooms should be no higher than Leq,30min 35dBA, with a further upper limit of 30dBA if hearing-impaired children are being taught. Clearly, this requirement is similar to the requirement for residential properties.
- 6.7.22 The prediction given in the PDA report gave a worst case internal noise level of 13 dB above the lowest required classroom noise level, which means that the required internal noise levels of the classrooms fronting the Link Road are not likely to be achieved with open windows. It is therefore likely that reliance on openable windows to provide ventilation of classrooms and other teaching spaces would not be possible. Therefore, there will be a need for a form of sound attenuated ventilation, which is not uncommon for new schools that are built adjacent to busy roads.

- 6.7.23 With regard to the playground area, BB93 recommends that noise levels in external playgrounds should be below Leq,30min 55dBA wherever possible, though it also recognises that this may not be possible on all sites. Indeed, it states at Section 2.2 that such areas may be used in some cases as buffer zones between high-noise transportation sources and school buildings. However, if the outdoor spaces are intended to be used for outdoor teaching (excepting sports pitches) the BB93 document states that where external noise levels exceed Leq, 30min 50dBA at these spaces, it may be necessary to introduce localised sound attenuating barriers for those portions of the playgrounds.
- 6.7.24 The EHO considers that achievement of the required noise levels is possible but may incur additional build and maintenance costs over that of a school built further into the site and away from the road. However, it is important to highlight that costs for the build can only be determined at the detailed design stage, and as the actual footprint and design of the school is yet to be determined, it is not possible to estimate.
- 6.7.25 The outline part of the application also includes potential for the expansion of Allerton Steel to the north of their existing boundary and the development of B1 (Business) and B2 (general industrial) uses. These uses could have the potential to generate noise that would impact upon the residential dwellings and school to the north of the proposed link road and any sensitive land uses to the south of the link road, although there is also potential for any buildings built between Allerton Steel and the development on the north side of the link road to help remove the 'line of sight' of Allerton Steel's noise source.
- 6.7.26 There has been objections submitted suggesting that the sensitive land uses applied for as part of this application would have the effect of preventing the expansion of Allerton Steel, would therefore result in that business being detrimentally affected, with the potential knock on effect of the loss of the industry and its employment to the town. It is essential to highlight though that at present Allerton Steel does not own or lease the land to its northern boundary and does not have planning permission for expansion, such that any expansions plans are speculative at this time. Furthermore, as the land has been allocated in the Local Plan since 2010, which is after Allerton's relocation to the site from Romanby, the subject proposal cannot be said to have come 'out of the blue'. Lastly, it must be noted that the Council has a statutory obligation to consider the application that is before it and not one that has not yet been made. It would not, therefore, be prudent to recommend refusal of an application based upon a hypothetical situation.
- 6.7.27 If or when Allerton Steel come to make a planning application to expand their premises, or at the time reserved matters applications are lodged for business or general industrial use on land to Allerton's north, they will need to be accompanied by their own noise impact assessments and mitigation plans to minimising their impact on other nearby land uses. This would be the case for any development of its kind whether it is an existing business or new.
- 6.7.28 In conclusion, based upon the advice of the EHO, it is considered that with the adoption of appropriate noise conditions, there is no reason, on the grounds of noise, that this application should not be granted planning permission. It is considered that the requirements of Local Policies CP1 and DP44 and the noise provisions contained within the NPSE, NPPF and PPG to avoid significant adverse impacts on health and quality of life and to mitigate and minimise adverse impacts on health and quality of life, can be achieved.

6.8 Design and Landscaping

Outline Application

6.8.1 In relation to the 'outline' part of the planning application, all matters except access to the site (i.e. appearance, landscaping, layout and scale) are reserved such that the position of particular buildings, internal roadways or proposed planting for example are all matters for later consideration and will be assessed against specific design policies and criteria within the Local Plan at the relevant time. In this regard, the outline part of the application simply has to satisfy one basic test – can the quantum of development proposed, accessed from the location detailed, be adequately accommodated within the constraints of the site?

- 6.8.2 Paragraph 58 of the NPPF states, amongst other things, that planning decisions should aim to ensure that developments will function well and add to the overall quality of the area, establish a strong sense of place, optimise the potential of the site to accommodate development, create and sustain an appropriate mix of uses (including the incorporation of green and other public space as part of developments) and support local facilities and transport networks.
- 6.8.3 Paragraph 61 of the NPPF states that although visual appearance and the architecture of individual buildings are very important factors, securing high quality and inclusive design goes beyond aesthetic considerations. Therefore, planning policies and decisions should address the connections between people and places and the integration of new development into the natural, built and historic environment.
- 6.8.4 The indicative layout plan, parameters plans and associated reports provided with the application illustrate one potential layout for the site. While the proximity of some of the houses and the school in relation to the road and Allerton Steel on Thurston Road has been questioned, the illustrative layout plan is not fixed and design will evolve through more detailed planning, as well as in regard to planning conditions imposed on any permission granted.
- 6.8.5 Notwithstanding, in their most basic form those illustrative plans demonstrate that 900 houses, a primary school, a neighbourhood centre, a sports village and employment uses can be accommodated in line with the original Masterplan vision for the site and would add to the overall quality of the area.
- 6.8.6 It must also be noted that a Landscape and Visual Impact Assessment (LVIA) prepared by Landcare was submitted with the application. Landcare have previously assisted the Council with preparation of Local Plans, including the original Masterplan for this site; in that regard they are well qualified to appraise the landscape impacts that could result from the proposed development. In summary, the LVIA identifies that the bulk of the proposed development area abuts open land of arable and grazing land but on the southern boundary between Darlington Road and Stokesley Road there is the large industrial estate of Thurston Road. The buildings of the industrial estate are visually dominant due to their scale and mass and the light coloured cladding panels, which means that they are very visible from viewpoints in the north west, north and north east including from Darlington Road, rail users, Northallerton Road and Stokesley Road. In this regard, the LVIA concludes that the main feature of the townscape interface is clearly the visual dominance of the industrial estate particularly as bounding hedgerows are of too small a scale, and hedgerow trees too widely spaced, to have any meaningful mitigating effect on the large white or pale toned buildings.
- 6.8.7 That said, the LVIA recognises that the proposed developments, while they would provide a screen to the industrial estate when viewed from the north, would be permanent and thus there will be no reversibility of landscape effects. The magnitude of change is therefore considered to be significant in terms of both the overall area that would be altered and the extent to which the northern built form boundary advances towards the north bringing development much closer both geographically and visually to residents of Brompton.
- 6.8.8 However, this was an anticipated outcome of the site Allocation and is not therefore a suitable justification for considering refusing the outline application, particularly as the 'Brompton Gap' which separates the North Northallerton development from Brompton to the north is protected by the retention and designation of open space areas each side of the railway line, south of Halfway House and west of Northallerton Road.
- 6.8.9 Overall, the conclusion is made that the proposed development will create a change in the land-use and landscape character of the development site. The nature of this change will however be similar in scale and complimentary in character to that existing elsewhere within the local area and the mitigation measures incorporated into the design will help to assimilate the development into the wider landscape and townscape context. Furthermore, and as identified in the LVIA, whilst there is a large scale loss of landscape permanently to the development, many of the most susceptible and sensitive landscape effect receptors can be protected. These include some of the hedges and associated trees, the watercourses and all the existing

- landscapes that lie within the designated open space areas. Further landscape mitigation is recommended in the LVIA which should form a planning condition on any planning permission granted.
- 6.8.10 With that mitigation implemented, it is considered that the outline development proposal can be integrated into its surroundings.

Full Application

- 6.8.11 With regard to the 'full' application the Planning Policies CP16 Protecting and Enhancing Natural and Man Made Assets, CP17 Promoting High Quality Design, DP30 Protecting the Character and Appearance of the Countryside, DP32 General Design and DP33 Landscaping are relevant. In summary all of these policies require in some way that proposals enhance and respect their surroundings, promote a public realm which is rich in identity, attractive and safe, and create opportunities for connected layouts.
- 6.8.12 In addition to Local Planning Policies Section 7 of the NPPF states at Paragraph 56 that good design is a key aspect of sustainable development, is indivisible from good planning, and should contribute positively to making places better for people. Paragraph 57 of the NPPF also states that it is important to plan positively for the achievement of high quality and inclusive design for all development, including individual buildings, public and private spaces and wider area developments schemes.
- 6.8.13 The design of the Phase 1 proposals has been influenced by the original adopted Masterplan for the site, but also the subsequent requirements of NYCC Highways, on the basis that the design of the link road and bridge has evolved from a vision of a 'village street' in the original Masterplan, to a strategic link road designed to accommodate HGV's to provide a viable alternative to their passage through the town. The latter has had the effect of altering the proximity of buildings adjacent to the link road and their level of interaction with the public realm. In addition, through the course of the application further design revisions have been required on the eastern site (Taylor Wimpey proposals) due to the specific drainage requirements for that scheme, and this has resulted in revisions to the internal roads, plot sizes/arrangement.
- 6.8.14 In relation to the internal roads, as noted in Section 5.2.1, NYCC Highways has provided comment on the east and west proposals and have advised that "the layout(s) will provide streets where vehicle speeds are naturally restrained to 20mph and walking routes are available on pedestrian desire lines feeding to the formal crossing points of the link road; this will also 'deliver safe routes to school' to the new school site." In this regard the proposal is considered to meet the aspirations of the original Masterplan movement framework and the movement requirements in Local Plan Policy DP32.
- 6.8.15 With regard to the built form, the Council has had the opportunity to influence the design through a large number of pre-application meetings and subsequent dialogue since the application has been submitted. However, it must be appreciated that with the contributions being sought from the developers in terms of dedication of the land for the school and sports village and dedication of land for (and delivery of) the link road and bridge, which has had considerable implications on the development's viability, the opportunity to demand a bespoke product from the developers is limited due to the increases in development costs and reduced returns that would be incurred.
- 6.8.16 Having said that, the proposals provide a mix of housing sizes and types, varied densities and notably different layouts on the east and west, which will not only appeal to a range of buyers but will also achieve the character change sought by the original Masterplan.
- 6.8.17 With regard to the eastern development and the Taylor Wimpey proposal it is noted that there are 14 two and a half storey units on their layout, 4 of which are located on the roundabout entrance in order to add the required massing to the roundabout gateway area, while on the western gateway two dual aspect Langdale house types have been plotted either side of the link road in order to front the link road yet not turn their back on the view when crossing from the sports village at Halfway Park. At 1,530

- sq ft they are amongst Taylor Wimpey's biggest products and can create the necessary dominant and symmetrical entrance way whilst maintaining a more rural, low density feel.
- 6.8.18 The applicants were asked to provide justification for not included any taller development on the site for greater diversity, and they advised that not only is there a lack of demand for townhouse properties in the current market, this part of the site has been cast by the original Masterplan as the more rural, low density area and so it is not appropriate to increase massing further than 2.5 storey. The rationale for the Taylor Wimpey design is considered acceptable.
- 6.8.19 With regard to the western development and the Persimmon Homes proposal it is important to appreciate the context from which the site would be accessed. When approached from the north the traveller will pass a garden centre, a field, a pair of semi-detached houses, dense hedgerow planting and then the new roundabout at the western end of the link road. Facing the traveller is the roundabout with link road sweeping east and slightly south, together with a significant greensward overlooked by the proposed neighbourhood centre buildings. The gateway roundabout with residential uses and appearances would strike a noticeable change in character from what has preceded it.
- 6.8.20 When approached from the south the traveller passes a variety of commercial and retail buildings including a hotel. These are single and two storey buildings and set back a good distance from the road behind car parking; they have no sense of enclosure. There will be new commercial buildings opposite the existing ones before approaching the flank of the proposed neighbourhood centre buildings leading up to the link road roundabout. Across the roundabout, facing the traveller, are the proposed new houses set behind a boulevard of frontage trees. The new housing is a terrace of three houses facing west, an imposing double fronted detached house looking towards the roundabout, and three other detached houses with different elevations and sizes, reflecting the curve of the link road, as encouraged by the planning officer at the pre-application stage.
- 6.8.21 After the seventh house is a road serving houses to the rear. The houses thereafter thus stand as three single blocks of development reflecting the curve of the highway but are an uneven distance from it to provide interest. The central of the three blocks is largely designed with larger detached houses, while the other two blocks comprise a variety of detached and terraced blocks to provide diversity of form and scale. The maximum height of the houses is 2.5 storeys, which is commensurate with the scale of dwellings in Northallerton.
- 6.8.22 The proposed house types have been influenced by the form and mass of other residential properties in Northallerton. The use of half dormers and bay windows to feature house types helps to break up the massing of the buildings and maintain visual interest. The predominant roof forms are gables although some are hipped, which reflects the general mix in the surrounding area. It is therefore considered that the proposed development will sit comfortably alongside the mix of existing development within the area.
- 6.8.23 Overall, it is considered that the proposed first phase is of appropriate scale, form and layout and will create a definable sense of place and accord with the principles of Building for Life 12. That said, the Police Architectural Liaison Officer did raise concern with the height of the rear garden boundary fences on the Persimmon Homes Phase 1 Scheme, as these are shown on the layout plan at 1200 mm high. In the interest of security, it is recommended that those fences be raised to between 1500 mm 1800 mm. A planning condition seeking this change should be included on any planning permission granted.
- 6.8.24 With regard to landscaping, several landscape plan iterations have been provided to the Council for consideration, with Pegasus preparing the plans for the Persimmon Scheme and Rosetta preparing the plans for the Taylor Wimpey Scheme. The Council asked Landcare to review the plans in light of the fact that they prepared the LVIA and to determine if the species proposed was appropriate for this location and style of development and also to determine if they accord with the landscape mitigation suggested in the LVIA.
- 6.8.25 Landcare provided the following advice on 6th October 2015:

Taylor Wimpey

I have suggested more planting on the Stokesley Road. A change of tree species along the road margins to native species. The hedges to the road frontages to be a native mix to match those in the surrounding landscape. Additional hedging along the link road frontages in line with the approach on the link road generally. Areas of species rich grassland along the link road frontages, on Stokesley Road and especially bordering the suds drainage ditch running through the site.

Persimmon Phase 1

I have suggested there is more opportunity for tree planting within the site than is currently

shown even given NHBC guidelines. I have asked for some larger species along the link road although it has been suggested there may be concerns on the part of the North Yorkshire Highways. I pointed out that the northern boundary of the site is designated as a hedge lined greenway but it seems that this is more likely to be incorporated on the north side of the designated route in a future phase.

Link Road landscape

I have suggested several species amendments to use predominantly native species along the highway verges. Also some species changes to the hedge mix in the same vein.

I have suggested substantial woodland planting on each side of the railway crossing to ensure it is integrated into the landscape and I have suggested substantial areas of species rich grassland behind the hedge lines with only mown grass along the immediate highway edges.

- 6.8.26 The applicants have taken on board the advice and provided new plans on 21st October 2015 with the following changes:
 - Species Rich Grassland Meadow Grass Mixture for Wetlands has been focused around the watercourse in order to provide the contrast recommended.
 - Native Mix tree mix along the spine road has been changed to alternating Tilia cordata and Carpinus betulus as recommended.
 - Additional Planting
 - At the north eastern edge of the site a mixture of new hedgerow, tree planting in rear gardens and trees to provide the mix required has been employed. Where homes back onto the boundary a 1.8m fence is required and therefore access is not available to maintain a hedge in this position. The proposal has therefore tried to introduce trees in suitable gardens to provide a softer edge. Where the boundary is accessible and a hedge is possible the plans have incorporated it along a 1.2m three rung fence in order to tie this in with the rural surroundings. A hedge has been added to the most easterly edge as recommended.
 - To the link road hedges have been included in the locations recommended.
- 6.8.27 Landcare reviewed the revised plans and were happy to endorse them in line with the suggestions in the Landscape and Visual Impact Assessment.
- 6.8.28 From a design, landscape, open space, ecological and permeability perspective it is considered that the proposed Phase 1 plans offer an appropriate and balanced option for the site. A planning condition should, however, be attached to any planning permission granted requiring compliance with the submitted plans.
- 6.8.29 In conclusion, the proposal is considered to be acceptable in design and landscape terms and compliant with the requirements of Policies CP16, Promoting High Quality Design, DP30, DP32 and DP33 of the Local Plan.

6.9 Conclusion

6.9.1 The application comprises a major development for which an Environmental Impact Assessment was carried out and an Environmental Statement was prepared and submitted with the planning application. The ES has been adequately publicised in

- accordance with the Town and Country Planning (Environmental Impact Assessment) (England and Wales) Regulations 2011.
- 6.9.2 Under Regulation 3 of those Regulations planning permission cannot be granted for EIA development unless the environmental information has been taken into account, which it has in this case. While the likely significant effects of a project on the environment should be identified and assessed at the time of the procedure relating to the principal decision, as stated in Planning Practice Guidance, if those effects are not identified or identifiable at the time of the principal decision, an assessment must be undertaken at the subsequent stage, which means either prior to the approval of reserved matters or prior to the discharge of any condition attached to a full planning permission.
- 6.9.3 Officers are satisfied that the ES contained sufficient information to enable them to assess the main or likely significant effects on the environment and to specify appropriate mitigation measures. Where there has been any doubt, planning conditions have been recommended requiring submission of further environmental information (such as further acoustic surveys) for various reserved matters applications.
- 6.9.4 Furthermore, in determining the application Officers have also had regard to the core message in the NPPF that the purpose of the planning system is to contribute to the achievement of sustainable development. The three dimensions of sustainable development consist of economic, social and environmental roles. Each of these dimensions will be discussed in turn.

Economic Role

- 6.9.5 Paragraph 7 of the NPPF explains that the economic role includes the availability of sufficient land of the right type, in the right places and at the right time to support growth; and building a strong, competitive economy is a cornerstone of Government Policy. In this regard the proposal would generate employment arising from construction works and new household expenditure supporting local facilities and services and increased employment arising from that economic activity. As the development would be a phased construction over a 10 year period it will have a beneficial effect on the local economy for a notable period of time.
- 6.9.6 While it might be argued that those benefits could arise in any event from other residential development elsewhere in the District, this site is the largest allocation in the Council's Local Development Framework and is highlighted as a priority in the North Yorkshire Housing Strategy and Local Investment Plan. As such there is no comparable site or sites and there is a pressing need for additional housing, which could be delivered as early as 2017. This is a factor of significant weight.
- 6.9.7 Consideration is also given to the economic benefits of the site in its existing use as agricultural land. However, as identified in the Agricultural Land Classification and Soil Resources Report dated May 2015, prepared by Reading Agricultural Consultants Ltd and submitted with the application, soil wetness, as influenced by the interaction between impeded drainage and clayey soil profiles, limits most of the site (69.1%) to no better than Subgrade 3b, . While this means that 31% of the land to be developed is best and most versatile land, it must be noted that the proposed allotments fall within the area classified as Grade 2 Very Good Quality land, thus the development will still be able to make good use of the fertile soils. Furthermore, loss of agricultural land was seen by the Local Plan Inquiry Inspector as a consideration overridden by housing need; a consideration which also applies in current circumstances.
- 6.9.8 The loss of the remaining best and most versatile land is therefore considered negligible.

Social Role

- 6.9.9 The proposal would provide the homes required to meet the needs of the present and future generations and it would offer every prospect, through good design, of creating a high quality built environment with accessible local primary school and local services at the proposed neighbourhood centre.
- 6.9.10 Of course though, it is noted that the site would deliver less affordable homes than the 40% desired by Policy CP9, which is disappointing when the North Yorkshire

Strategic Housing Market Assessment concluded that affordable housing needs had increased and that there was a need for an additional 320 affordable homes annually (up to 2016) in Hambleton. However, as identified in Section 6.4, the Council would be in a very weak position trying to require more affordable housing than the viability appraisal demonstrates is possible. The Council would, nevertheless, have the opportunity to revisit the level of affordable housing through a review mechanism recommended to be included within a Section 106 Agreement that would accompany the planning permission. If the housing market improves there may be opportunity for increased affordable housing provision at later phases.

6.9.11 The project would also deliver public open space, allotments, and formal children's play space within the site, as well as enhancing the pedestrian/cycling route in to town along Northallerton Road by closing it off to through vehicles. These are substantial benefits to be added to the overall balance.

Environmental Role

- 6.9.12 Although the application site is greenfield land, the bulk of it is allocated for the proposed uses and the expansion of the development limits has been deemed acceptable given the out of date housing figure, the Government's call to boost significantly the supply of housing, and the inevitable need to release some additional greenfield land to meet housing shortfall. On the basis that the green gap between Brompton and Northallerton is to be retained largely as outlined in the Allocations Document, that the landscape strategy would be capable of providing effective definition and containment, that the site has no 'special' designation, and that the site presents limited ecological value, it is considered that the environmental impacts of the development would be minimal.
- 6.9.13 While there are local concerns about the opportunities for walking, cycling and horse riding through and across the site, the development is not bereft of any walking routes or a cycle path and NYCC Highways has advised that the provision is commensurate with both the scale of the development and the evidence of anticipated users. Furthermore, once the development of the neighbourhood centre is established, the need to head in to Northallerton Town Centre for every day services and provisions will be reduced. The proposal would be generally consistent with Policies CP1 and CP2.

Other Considerations

- 6.9.14 With reference to the number of written objections on the lack of housing need, the housing policies in the local plan (particularly CP5 on the amount of housing and CP4 and DP9 on the need to demonstrate exceptional circumstances for proposals outside development limits) are out of date or inconsistent with the provisions of the NPPF, and as a result provide insufficient housing land to meet identified needs. Therefore, pending the preparation of a new plan, allocated sites (which have already undergone considerable evidence gathering and scrutiny by a Planning Inspector) such as the subject site must be encouraged to come forward and their expansion beyond the development limits considered on their own merits.
- 6.9.15 A history of surface water flooding in the locality is well documented but the technical evidence supporting the application, and scrutinized by the Council's Drainage Engineer, the EA, NYCC SUDS Officer indicates that it would be possible to ensure that surface water from the site could be managed so as not to cause or add to any flooding in the area. While the IDB Engineer has expressed concerns about displacement of water as a result of land raising, he suggests a planning permission condition seeking further detail of the exceedance measures, which would need to be approved before any development could commence. This is a sufficient resolution. With regard to existing problems in Brompton, they are not a sufficient basis to preclude the proposed development.
- 6.9.16 The issue regarding the problems at Low Gates level crossing is also well documented, but as detailed in Section 6.3 the proposed development would not create a further burden on that part of the highway network. While the proposed link road and bridge would not be a panacea for the highway problems in and around Northallerton, it will bring some relief to certain key junctions and provide an alternative to Low Gates. Furthermore, it is designed to mitigate the impact of the

- number of vehicles associated with the 900 new dwellings and on that basis the development as a whole.
- 6.9.17 The issue of noise impacts from the operations of Allerton Steel on the proposed new dwellings and aged care facilities has also been raised in some length and the concern raised by Allerton Steel that they might be forced to adapt their working practices in the future to deal with noise complaints is a valid one and is acknowledged. However, the acoustic testing undertaken to date of the existing conditions demonstrates that the dwellings within Phase 1 could be designed and constructed to minimise noise impacts. With regard to development in later phases it is recommended that additional acoustic testing is undertaken for the development of any sensitive land use across the site or in the proximity of sensitive land uses and that appropriate mitigation if required is conditioned and implemented. As the development of future phases, including the expansion of Allerton Steel, is not as yet defined, it would be premature to assume that acoustic impacts could not be overcome.
- 6.9.18 Finally, although it is said that the proposal would place a strain on secondary school places, and on local health services, none of the relevant providers oppose the development. A variety of other concerns have been noted, but none provide a compelling reason to count against the proposal.
- 6.9.19 In conclusion, planning law requires that applications for planning permission must be determined in accordance with the development plan, unless material considerations indicate otherwise. The NPPF is a material consideration in planning decisions.
- 6.9.20 The three roles of sustainable development are mutually dependent. The proposal would deliver economic and social benefits and would be neutral on environmental benefits. The pattern of growth would be in a generally sustainable location, its negative impacts on the countryside would be minimal and capable of being offset by a landscape strategy; and the loss of best and most versatile agricultural land would be of negligible consequence. While the views of local people are an important consideration, the limited adverse impacts arising from the development would not significantly and demonstrably outweigh the very clear benefits of the scheme when assessed against the policies in the NPPF taken as a whole.
- 6.9.21 In terms of the development plan, the housing target in Policy CP5 is out of date, which in turn undermines the restraint on development outside the development limits given the Council's lack of a 5 year housing supply. Notwithstanding, the proposal would be generally consistent with Policies CP1 and CP2 and the underlying spatial principles. The proposal would not wholeheartedly comply with the targets for affordable housing set out in Policy CP9, however, that Policy recognises that the actual provision on individual sites will be determined through negotiations taking into account viability and the economics of the provision. The applicant's viability assessment has been thoroughly appraised and the provision of up to 15% affordable housing has been justified. This will be revisited through an appropriate review mechanism attached to a Section 106 Agreement. It can therefore be said that the proposal is generally consistent with the development plan when read as a whole.
- 6.9.22 Overall, the benefits arising from the development are significant; primarily the provision of 900 dwellings in an area suffering from a shortfall. Furthermore, the significance of other benefits should not be underestimated. The link road and bridge would bring some relief to Northallerton's queues caused by level crossings over key roads into/out of the town. The land for a two form entry primary school will not only serve the new community on the appeal site, but also wider strategic growth in the immediate surroundings, alleviating existing pressures on primary schools in the district. The land for a much needed sports village will allow expansion of existing sports clubs in the area and contribute to the health and wellbeing of all Hambleton's residents. Adverse impacts are limited.
- 6.9.23 It is therefore recommended that subject to the imposition of conditions and the completion of a Section 106 Agreement to secure the necessary and justifiable infrastructure works, contributions and affordable housing, that planning permission be granted for the application as submitted.

7.0 RECOMMENDATION

That Planning Permission is **GRANTED** for both the Outline and Full proposals subject to:

- 1) Authority being delegated to the Executive Director, in consultation with the Chairman of Planning Committee to agree the terms of a Section 106 Agreement that shall include the following key points:
 - Securing the provision of up to 15% affordable housing subject to deductions for any necessary developer contributions.
 - Requiring full, independently assessed, viability assessments undertaken for all residential phases beyond Phase 1, to inform the percentage of affordable housing deliverable as part of that phase.
 - · Securing the tenure of the affordable homes on all phases.
 - Securing the timing of the completion of the link road and bridge in the early stages of the Phase 1 development and its transfer to NYCC Highways.
 - Securing the transfer and timing of the transfer of the provision of land within the site for a primary school.
 - Securing the transfer and timing of the transfer of land for the sports village, allotments, and associated landscape buffers.
 - Securing the provision and implementation of Travel Plans for each of the various uses.
 - Securing a mechanism for the future maintenance of landscaped areas (excepting the sports village and allotments).
- 2) Subject to the following conditions:

Phasing

1. With the exception of Phase 1 (approved in full as part of this permission – 291 dwellings, the link road and bridge, the village green and SUDS), a detailed phasing plan for the development shall be submitted to and approved in writing by the local planning authority prior to the submission of the first reserved matters application and prior to the commencement of any development. The phasing plan shall specify the proposed timing for delivery of the housing and other build elements of the development. The development shall be carried out in accordance with the approved details, unless otherwise agreed in writing by the Local Planning Authority.

Reason: To ensure compliance with Section 92 of the Town and Country Planning Act 1990.

Reserved Matters to be Submitted

2. No development shall commence on any part of the development other than the formation of access roads and the works specified in Phase 1 (approved in full as part of this permission – 291 dwellings, the link road and bridge, the village green and SUDS) until all of the following reserved matters have been submitted for approval by the Local Planning Authority: (1) Appearance (2) Landscaping (3) Layout (4) Scale. An application for approval of reserved matters for all phases of the development must be made to the Local Planning Authority not later than the expiration of five (5) years beginning with the date of the grant of outline planning permission.

Reason: To enable the Local Planning Authority to properly assess these aspects of the proposal against relevant policies within the Development Plan (particularly DP32) before the development commences and to ensure compliance with Section 92 of the Town and Country Planning Act 1990.

Commencement of Development

3. Unless otherwise agreed in writing by the Local Planning Authority, the development of Phase 1 (approved in full as part of this permission – 291 dwellings, the link road and bridge, the village green and SUDS) shall be begun within 12 months of the date of this permission. Each subsequent phase or sub phase of the development hereby permitted shall be begun within two (2) years of the date of approval of the last of the reserved matters to be approved for that phase or sub phase.

Reason: To ensure that housing is delivered in a timely manner to meet the District's Full Objectively Assessed Housing Need and to ensure compliance with Section 92 of the Town and Country Planning Act 1990.

Approved Plans - Outline

- 4. The development of the outline planning permission hereby approved shall not be undertaken other than generally in accordance with the Environmental Statement (and its technical appendices) submitted with the application and the plans and documents detailed below unless required otherwise by another planning condition of this permission or unless otherwise agreed in writing by the Local Planning Authority:
 - Application Boundary Plan, Nod. 0000-001 Rev B, dated Jan 2014, produced by Spawforths;
 - Illustrative Masterplan Scaled, Nod. 0000-0019 Rev A, dated May 2015, produced by Spawforths;
 - Parameters Land Use Plan, Nod. P0-MP-SPA-P3769-5IL-1000-0001 Rev B, dated 22 May 2015, included within the Design and Access Statement Planning Issue 02 June 2015, produced by Spawforths;
 - Parameters Movement Framework Plan, Nod. P0-MP-SPA-P3769-5IL-1000B Rev B, dated May 2015, included within the Design and Access Statement Planning Issue 02 June 2015, produced by Spawforths;
 - Parameters Character Areas Plan, Nod. P0-MP-SPA-P3769-5IL-1000-0001 Rev A, dated May 2015, included within the Design and Access Statement Planning Issue 02 June 2015, produced by Spawforths;
 - Parameters Density and Phase 1 Areas Plan, Nod. P0-MP-SPA-P3769-5IL-1000-0001 Rev A, dated May 2015, included within the Design and Access Statement Planning Issue 02 June 2015, produced by Spawforths;
 - Parameters Open Space and Landscape Framework Plan, Nod. P0-MP-SPA-P3769-5IL-1000-0001 Rev A, dated May 2015, included within the Design and Access Statement Planning Issue 02 June 2015, produced by Spawforths;
 - Proposed Link Road Alignment and Access Arrangements Plan, Nod. 12003/GA/11 Rev G, dated September 2015, produced by Optima Highway Solutions;
 - Bridge Over Railway General Arrangement Plan, Nod. BHB-DRG-1000 Rev 03, dated March 2015, produced by Buro Happold Engineering; and
 - The Design and Access Statement, Planning Issue 02 June 2015, prepared by Spawforths.

Reason: In order that the development is undertaken in a form that is appropriate to the character and appearance of its surroundings and in accordance with the Development Plan Policies noted in the officer's report to committee.

Approved Plans - Full Phase 1

- 5. The development of Phase 1 approved in full as part of this permission 291 dwellings, the link road and bridge, the village green and SUDS shall not be undertaken other than generally in accordance with the Environmental Statement (and its technical appendices) and the plans detailed below unless required otherwise by another planning condition of this permission or otherwise agreed in writing by the Local Planning Authority:
 - Phase 1 Red Line Boundary Plan, Nod. 12003-SK-01, Rev B, dated August 2015, and produced by Optima Highways Solutions;
 - Illustrative Masterplan Scaled, Nod. 0000-0019 Rev A, dated May 2015, and produced by Spawforths;
 - Persimmon Homes Proposed Site Plan, Nod. YOR.2457.004 Rev B, dated 09 Sept 2015, and produced by Pegasus Urban Design;
 - Persimmon Homes Proposed Site and Street Scenes Plan, Nod. YOR.2457.003 Rev B, dated 09 Sept 2015 and produced by Pegasus Urban Design;

- Taylor Wimpey Planning Layout Plan, Nod. NN:00 Rev C, dated 26 Oct 2015, and produced by Taylor Wimpey;
- Proposed Link Road Alignment and Access Arrangements Plan, Nod. 12003/GA/11 Rev G, dated September 2015, and produced by Optima Highways Solutions;
- Bridge Over Railway General Arrangement Plan, Nod. BHB-DRG-1000 Rev 03, dated March 2015, produced by Buro Happold Engineering;
- Taylor Wimpey Planning Layout Plan, Nod. NN:00 Rev C, dated 26 Oct 2015, and produced by Taylor Wimpey
- Taylor Wimpey Landscape Plan Sheet 1 of 3, Nod. 2585/1 Rev H, dated 02 Nov 2015, and produced by Rosetta Landscape Design.
- Taylor Wimpey Landscape Plan Sheet 2 of 3, Nod. 2585/2 Rev H, dated 02 Nov 2015, and produced by Rosetta Landscape Design.
- Taylor Wimpey Landscape Plan Sheet 3 of 3, Nod. 2585/3 Rev H, dated 02 Nov 2015, and produced by Rosetta Landscape Design.
- Landscape Masterplan for Spine Road, Nod. YOR.2457.010C Sheet 1 of 5, dated Oct 2015, and produced by Pegasus Urban Design;
- Landscape Masterplan for Spine Road, Nod. YOR.2457.006C Sheet 2 of 5, dated Oct 2015, and produced by Pegasus Urban Design;
- Landscape Masterplan for Spine Road, Nod. YOR.2457.007C Sheet 3 of 5, dated Oct 2015, and produced by Pegasus Urban Design;
- Landscape Masterplan for Spine Road, Nod. YOR.2457.008C Sheet 4 of 5, dated Oct 2015, and produced by Pegasus Urban Design;
- Landscape Masterplan for Spine Road, Nod. YOR.2457.009E Sheet 5 of 5, dated Oct 2015, and produced by Pegasus Urban Design;
- The following Persimmon House Type Elevations and Floor Layout Plans:
 - Bickleigh Village (BK-WD16) Rev A;
 - Chedworth (CD-WD10);
 - Clayton Village (CA-WD16) Rev C;
 - Corfe Village (CF-WD16) Rev F;
 - Hanbury Village (HB-WD16) Rev P;
 - Hatfield Village (HT-WD16) Rev P;
 - Henley (HE/2 Bed);
 - Morden Village (MR-WD16) Rev J;
 - Moseley Village (MS-WD06) Rev L;
 - Penshaw (PW/3 Bed);
 - Pickering and Folkestone (PIC & FO);
 - Roseberry Village (RS-WD16) Rev S;
 - Rufford Village (RF-WD16) Rev P;
 - Winster Village (WS-WD16) Rev S.
- The following Taylor Wimpey House Type Elevations and Floor Layout Plans:
 - Aldenham Issue 5 (PD32/5/PL1 & PD32/5/PL2);
 - Alton Issue 5 (PB35/5/PL1 & PB35/5/PL2);
 - Bellerby Issue 5 (ZA34/5/PL1 & ZA34/5/PL2);
 - Bradenham Issue 5 (PD48/5/PL1 & PD48/5/PL2);
 - Cotterdale Issue 5 (ZA25/5/PL1 & ZA25/5/PL2);
 - Downham Issue 5 (PD49/5/PL1 & PD49/5/PL2);

- Gosford Issue 5 (PA34/5/PL1 & PA34/5/PL2);
- Haddenham Issue 5 (PD411/5/PL1 & PD411/5/PL2);
- Holywell 722 (722/5/PL3 & 722/5/PL1);
- Langdale Issue 5 (PT43/5/PL1 & PT43/5/PL2);
- Lavenham Issue 5 (PD51/5/PL1 & PD51/5/PL2);
- Midford Issue 5 (PA44/5/PL1 Rev A & PA44/5/PL2 Rev A):
- Shelford Issue 5 (PA48/5/PL1 & PA48/5/PL2);
- Woodruff Issue 4 (B881/4/PL1 & B881/4/PL2).

Reason: In order that the development is undertaken in a form that is appropriate to the character and appearance of its surroundings and in accordance with the Development Plan Policies noted in the officer's report to committee.

Boundary Treatment – Details

6. The rear garden side and rear walls or fences on the Phase 1 residential plots, as shown on Persimmon Homes Proposed Site Plan, Nod. YOR.2457.004 Rev B, dated 09 Sept 2015, and produced by Pegasus Urban Design shall be increased in height from 1200 mm to a minimum of 1500 mm and a maximum of 1800 mm.

Reason: To provide adequate privacy and security to individual occupiers of those dwellings in accordance with Development Plan Policy DP32 and advice of the Police Architectural Liaison Officer.

Landscaping

- 7. The reserved matters application(s) for landscaping referred to in Condition 2 of this permission must include a masterplan for the entire site encompassing all areas identified on the "Open Space and Landscape Framework" Nod. P0-MP-SPA-P3769-5IL-1000-0001 Rev A, dated May 2015, including the following:
 - a) The mitigation measures proposed in the Landscape and Visual Impact Assessment of Appendix 7, Volume 2 of the Environmental Statement submitted with the application;
 - b) The landscape measures approved as part of the Phase 1 development;
 - The ecological enhancement measures required by Condition 27 of this permission;
 - d) A detailed landscape scheme for watercourses;
 - e) A detailed scheme for buffer area between Northallerton and Brompton and the northern boundary generally;
 - f) A detailed scheme for "the Meander at Hawthorne" including suitable integrated fencing and/or planting on the western side of the railway line to prevent public access to the railway line;
 - Individual planting plans for each phase of the development (including schedules of plant species, sizes, numbers or densities, and in the case of trees, planting, staking, mulching, protection, soil protection and after care methods); and
 - h) An implementation and management programme.

The Masterplan and individual planting plans must be approved in writing by the Local Planning Authority. Landscaping shall be carried out in accordance with the approved design, implementation and management programme unless otherwise agreed in writing by the Local Planning Authority.

Reason: To enable the Local Planning Authority, to determine that the proposals will maintain and improve the amenity of the area, and to monitor compliance.

8. If within a period of five years, from the completion of any phase (or sub phase) of the development, any trees, plants, or hedgerows planted as part of this development either die, are removed, become seriously damaged or diseased, or become (in the opinion of the Local Planning Authority) otherwise defective, they shall be replaced with the same species in the current planting season or the first two months of the next planting season, whichever is the sooner, at the developers expense, unless the Local Planning Authority gives written consent

to any variation.

Reason: To maintain and improve the amenity of the area.

9. Prior to the commencement of Phase 1 of the development (approved in full as part of this permission – 291 dwellings, the link road and bridge, the village green and SUDS), a revised Planting Plan for the Persimmon Homes Scheme to the west of the site to accord with the Landscape Masterplan for the Spine Road, Nod. YOR.2457.006C Sheet 2 of 5, dated October 2015 and produced by Pegasus Urban Design, must be submitted to and approved in writing by the Local Planning Authority.

Reason: To ensure consistency between approved plans.

10. Trees identified as being retained on the site in the amended Arboricultural Report submitted to the Local Planning Authority on 6th October 2015, must be protected in line with BS 5837:2012 (or any subsequent guidance). If during the construction of the development a tree identified as being retained in the Arboricultural Report is required to be removed, the written consent of the Local Planning Authority must be obtained prior to the removal of the said tree.

Reason: To maintain and improve the amenity and biodiversity of the area.

Hours of Construction

11. No building works including excavation, demolition works, piling operations, or other external construction works in general shall be carried out within 200 metres of the boundary of any residential property except between the hours of 07:30 and 18:00 Monday to Friday and 08:00 and 13:00 Saturday, with no construction taking place on Sunday or any public holiday unless by prior written consent of the Local Planning Authority.

Reason: In order to protect the amenity of neighbouring occupiers in accordance with the Development Plan Policies CP1 and DP1.

Construction Method Statement

- 12. No development for any separate phase or sub-phase of the development shall take place until a Construction Method Statement and Plan for that separate phase or sub-phase has been submitted to, and approved in writing by, the Local Planning Authority in consultation with the Local Highway Authority. The approved Phase Construction Method Statement shall be adhered to throughout the construction period for the phase. The statement shall provide for the following in respect of the phase:
 - i) Working hours on site (having regard to Condition 11 above);
 - ii) The parking of vehicles of site operatives and visitors;
 - iii) Loading and unloading of plant and materials;
 - iv) Construction traffic management;
 - The routing of HGV and other Construction related traffic to and from the site;
 - vi) Timing of deliveries and other construction related traffic to and from the site;
 - vii) A programme for the delivery of the works;
 - viii) Details and programme of all traffic management (temporary signs and controls) proposed;
 - ix) Details of construction accesses to be used during the building of roads, dwellings and other infrastructure.
 - Storage of plant and materials used in the construction of the development to avoid any potential ground contamination;
 - xi) The erection and maintenance of security hoarding where appropriate:
 - xii) Wheel washing facilities;
 - xiii) Measures to control the emission of dust and dirt during construction;
 - xiv) Measures to control noise and vibration during construction;
 - xv) Means of protection of trees and hedgerows during site preparation and construction;
 - xvi) A scheme for recycling/disposing of waste resulting from demolition, excavation and any other construction works.

Reason: To maintain and improve the amenity of the area and in the interests of the safety and convenience of highway users.

<u>Levels</u>

13. No phase of the development (including Phase 1 approved in full as part of this permission –

291 dwellings, the link road and bridge, the village green and SUDS) shall commence until detailed cross sections showing the existing ground levels (of the site and the immediately adjacent land) in relation to the proposed ground and finished floor levels for that phase of the development have been submitted to and approved in writing by the Local Planning Authority. The levels shall relate to a fixed Ordnance Datum. The development shall be constructed in accordance with the approved details.

Reason: To ensure that the development is appropriate in terms of amenity and meeting sustainability objectives in accordance with Development Plan Policies CP1 and DP1.

Archaeology

- 14. No development (which includes site clearance works and excavation) shall take place/commence on any separate phase or sub phase until a Written Scheme of Investigation has been submitted to and approved by the Local Planning Authority in writing in respect of each phase of the development. The Scheme shall include an assessment of significance and research questions, and:
 - a) The programme and methodology of site investigation and recording;
 - b) Community involvement and/or outreach proposals;
 - c) The programme for post investigation assessment;
 - d) Provision to be made for analysis of the site investigation and recording;
 - e) Provision to be made for publication and dissemination of the analysis and records of the site investigation;
 - Provision to be made for archive deposition of the analysis and records of the site investigation;
 - g) Nomination of a competent person or persons/organisation to undertake the works set out within the Written Scheme of Investigation.

Reason: This condition is imposed in accordance with Section 12 of the NPPF as the site is of archaeological interest and on the advice of North Yorkshire County Council Archaeological Officer.

15. No demolition/development shall take place other than in accordance with the Written Scheme of Investigation approved under Condition 14 of this permission.

Reason: This condition is imposed in accordance with Section 12 of the NPPF as the site is of archaeological interest and on the advice of North Yorkshire County Council Archaeological Officer.

16. No part of the development on any separate phase or sub phase shall be occupied or the link road operational until the site investigation and post investigation assessment has been completed in accordance with the programme set out in the Written Scheme of Investigation approved under Condition 14 of this permission and the provision made for analysis, publication and dissemination of results and that archive deposition has been secured.

Reason: This condition is imposed in accordance with Section 12 of the NPPF as the site is of archaeological interest and on the advice of North Yorkshire County Council Archaeological Officer.

- 17. The details submitted in pursuance of Condition 14 above shall be preceded by the submission to the Local Planning Authority for approval in writing, and subsequent implementation, of a Scheme of Archaeological Investigation to provide for:
 - a) The proper identification and evaluation of the extent, character and significance of archaeological remains within the application area; and
 - b) An assessment of the impact of the proposed development on the archaeological significance of the remains.

Reason: This condition is imposed in accordance with Section 12 of the NPPF as the site is of archaeological interest and on the advice of North Yorkshire County Council Archaeological Officer.

Flood Risk

18. No development (which includes site clearance works and excavation) shall take place for any phase until a Detailed Design and associated Management and Maintenance Plan of surface

water drainage for the site based on sustainable drainage principles and an assessment of the hydrological and hydrogeological context of the development has been submitted to and approved in writing by the local planning authority. The surface water drainage design should demonstrate that the surface water runoff generated during rainfall events up to and including the 1 in 100 years rainfall event, to include for climate change and urban creep, will not exceed 1.4 l/s/ha following the corresponding rainfall event. The approved drainage design shall be implemented in accordance with the approved detailed design prior to completion of the development.

Reason: To prevent the increased risk of flooding; to ensure the future maintenance of the sustainable drainage system; to improve and protect water quality; and in accordance with Development Plan Policies CP21 and DP43.

19. The Detailed Drainage Design to be submitted in line with Condition 18 of this permission shall demonstrate that the surface water drainage system(s) are designed in accordance with the standards detailed in North Yorkshire County Council's SUDs Design Guidance which is in place at the time of submission.

Reason: To prevent the increased risk of flooding; to ensure the future maintenance of the sustainable drainage system; to improve and protect water quality; and in accordance with Development Plan Policies CP21 and DP43.

20. The Management and Maintenance Plan to be submitted in line with Condition 18 of this permission must provide for the lifetime of the development, and the details shall include, but not be exclusive to, the arrangements for adoption by any public authority undertaker and any other arrangements to secure the operation of the scheme throughout its lifetime.

Reason: To prevent the increased risk of flooding; to ensure the future maintenance of the sustainable drainage system; to improve and protect water quality; and in accordance with Development Plan Policies CP21 and DP43.

- 21. The development shall be carried out generally in accordance with the submitted Flood Risk Assessment and the following mitigation measures outlined below unless otherwise agreed in writing by the Local Planning Authority. These measures shall be fully implemented prior to occupation of any phase of the development, and according to the scheme's phasing arrangements as approved by the local planning authority:
 - a) Surface water run-off will be limited to 1.4l/s/ha so that it does not increase the risk of flooding off site.
 - b) Provision of level for level compensatory flood storage to mitigate for the proposed link road as described in Section 8 of the submitted Flood Risk Assessment and Drawing No. 4033 FRA05 Rev 0. The compensatory storage shall be located outside of Flood Zone 3, and must be approved in writing by the local planning authority in consultation with the Environment Agency and provided before any construction of the link road within Flood Zone 3 commences. This compensatory flood storage submission to the local planning authority should also include a report providing detailed calculations demonstrating the loss of storage as a result of the development, and, that gained by the provision of compensatory storage. The applicant must also include detailed design drawings for the proposed compensatory storage detention basins.
 - c) The applicant shall provide large box section culverts in the embankment for the link road as detailed in Section 8.13 of the submitted Flood Risk Assessment and Drawing No. 4033 FRA05 Rev 0. These should be sized appropriately to maintain existing overland flows.

Reason: To prevent flooding by ensuring the satisfactory disposal of surface water from the site; to prevent flooding elsewhere by ensuring compensatory storage of flood water is provided; to ensure that existing flood flow routes are maintained, in accordance with Development Plan Policies CP21 and DP43.

22. With the exception of the proposed link road, for which level for level compensatory storage shall be provided, there must be no raising of ground levels within the floodplain, and all excess spoil shall be removed from the floodplain (prior to the occupation of the development).

Reason: To ensure that there is no loss of storage within the floodplain, and to ensure that possible future flood flows are not pushed on to others in accordance with Development Plan Policies CP21 and DP43.

23. Prior to the commencement of the development, the construction details of the scheme to raise ground levels and deepen the watercourse through the east of the site shall be submitted to and approved by the Local Planning Authority in consultation with the Internal Drainage Board. The effect of raised ground levels should be investigated with regard to loss of flood plain to the site's easternmost watercourse, surface water flood risk, exceedance flow paths through the development and consequent effect on downstream flood risk; furthermore, a scheme to mitigate any residual flood risk shall be submitted to and approved by the Local Planning Authority prior to the commencement of the development.

Reason: To ensure that there is no loss of storage within the floodplain, and to ensure that possible future flood flows are not pushed on to others.

Drainage and Sewerage

24. The development shall be constructed in accordance with the phasing plan submitted to and approved by the Local Planning Authority. Prior to commencement of each phase of the development evidence shall be submitted to and approved by the local planning authority that the necessary waste water infrastructure (including off-site works and the point of connection [s] into the existing public sewer) and whether permanent or temporary solutions, will be provided to serve that phase of the development. No occupation of any part of the development shall commence until it has been proven to the satisfaction of the Local Planning Authority that the required foul water drainage system has been implemented. Furthermore, the site shall be developed with separate systems of drainage for foul and surface water on and off site and no additional surface water shall discharge to the existing local public sewer network.

Reason: In order to ensure that provision of waste water infrastructure is provided in time in accordance with Development Plan Policy DP6

25. Notwithstanding the above, foul water drainage associated with phase 1 of the development shall be constructed and operated in accordance with Sustainable Drainage Assessment and Detailed Foul and Surface Water Drainage Report no. 437/SDD1(B) and associated drawing no. 4377-C--D7-)1 both dated May 2015 and the Sustainable Drainage Assessment and Detailed Foul and Surface Water Drainage Report 3525/SDD1(A) all prepared by iD Civils Design Ltd. Furthermore, unless otherwise approved in writing by the local Planning Authority, the pumped foul water discharge from any part of the Phase 1 development to public sewer shall not exceed 4 litres a second.

Reason: In the interest of satisfactory drainage and Development Plan Policies DP6 and DP43.

Public Rights of Way

26. No works are to be undertaken which will create an obstruction, either permanent or temporary to the Public Right of Way adjacent to the proposed development.

Reason: To continue to allow users of the PROW to pass unhindered and safely.

Ecology

27. No development (which includes site clearance works and excavation) shall take place/commence on any separate phase or sub phase until an Ecological Enhancement Management Plan has been submitted to the Council for approval for each separate phase or sub phase. The Plan will set out all necessary ecological mitigation and enhancements measures, including those specified in Chapter 8: Biodiversity of the ES, to cover the construction of the development and not less than 10 years following its completion. The Plan will make provisions for, but is not limited to, habitats, terrestrial mammals, bats, birds, and invertebrates. The biodiversity mitigation and compensation for Phase 1 (approved in full as part of this permission – 291 dwellings, the link road and bridge, the village green and SUDS) must be fully self contained and must not rely on mitigation schemes from the Outline Phases to secure net gains for biodiversity.

Reason: In accordance with the objectives of mitigating impacts on ecological interests and enhancing biodiversity, and Development Plan Policies CP16 and DP31.

28. A badger survey must be carried out not less than 12 weeks before the commencement of construction on each separate phase or sub phase. Results and appropriate mitigation

measures (where required) must be submitted to the Local Planning Authority and Natural England for approval in writing and thereafter implemented in full.

Reason: In accordance with the objectives of mitigating impacts on ecological interests and enhancing biodiversity, and Development Plan Policies CP16 and DP31.

29. No development shall commence until a scheme providing for pre-construction surveys of and mitigation impacts for breeding birds including any nests on the site of each separate phase or sub phase has been submitted to and approved in writing by the Local Planning Authority. Surveys should also be suitable to record Schedule 1 barn owl, if present.

Reason: To comply with the Wildlife and Countryside Act 1981 to protect all wild birds whilst they are actively nesting or roosting and in accordance with the objectives of mitigating impacts on ecological interests in Development Plan Policies CP16 and DP31.

30. To ensure that works are legally compliant, vegetation clearance should be undertaken between September and late February to avoid the bird nesting season (March–August) unless otherwise agreed in writing by the Local Planning Authority. Any nests identified during vegetation clearance should be protected until the young have fledged. Where a Wildlife and Countryside Act Schedule 1 species is involved, mitigation for impacts, e.g. loss of nesting site, must be devised, approved by the Local Planning Authority and implemented. If vegetation clearance and/or construction work is approved in writing by the Local Planning Authority during or extending into the bird nesting season, further assessments and mitigation measures, such as ecological supervision at the developers expense, may be required by the Local Planning Authority.

Reason: To comply with the Wildlife and Countryside Act 1981 to protect all wild birds whilst they are actively nesting or roosting and in accordance with the objectives of mitigating impacts on ecological interests in Development Plan Policies CP16 and DP31.

31. To reduce the risk of bats roosting in the property known as Hartington House on the site, the buildings should be demolished as soon as practicable. However, in the event that the buildings are not removed before September 2016, bat dusk emergence and dawn re-entry surveys must be repeated in September 2016 and annually thereafter if the buildings continue to remain until the development of that part of the site is undertaken. The results of the surveys and any required mitigation measures must be submitted to the Local Planning Authority for written approval prior to the removal of those buildings after September 2016.

Reason: In accordance with the objectives of protecting Protected Species, mitigating impacts on ecological interests and enhancing biodiversity, and Development Plan Policies CP16 and DP31.

32. The Phase (or Phases) of development which includes the buildings at Sheepcote Close identified on Drawing Number MMD-326684-L-DR-XX-2008, Rev P1, dated 31/03/2014 and included in the Mott MacDonald Initial Bat Assessment Report, dated May 2015 and forming part of Appendix 8.1 of Volume 2 of the submitted Environmental Statement, shall not commence until all the buildings are reassessed for the presence of bats. The results of the initial assessment must be submitted to the local planning authority for approval. If after this initial assessment further emergence and activity surveys are deemed to be necessary, the Phase (or Phases) of development shall not take place until these further surveys have taken place, mitigation measures identified and reports prepared and submitted for approval in writing by the Local Planning Authority. The mitigation measures identified must then be undertaken in accordance with the approved mitigation scheme.

Reason: In accordance with the objectives of protecting Protected Species, mitigating impacts on ecological interests and enhancing biodiversity, and Development Plan Policies CP16 and DP31.

Noise

33. The dwellings hereby permitted shall be designed to ensure that the following internal noise levels (as recommended by BS8233:2014 – Guidance on Sound Insulation and Noise Reduction for Buildings) are not exceeded due to environmental noise, as follows:

Bedrooms - 30dB L_{Aeq},* and 35dB L_{Aeq}, T

Living rooms- 35dB L_{Aeq, T}[†] Dining Room – 40dB L_{Aeq, T}[†]

- *- Night-time 8 hours between 23:00-07:00
- † Daytime 16 hours between 07:00-23:00.

A validation test shall be carried out on a relevant sample of premises (to be agreed with the Local Planning Authority) during daytime and night time hours following completion of each phase of the development but prior to occupation. The results shall be submitted to the LPA for approval in writing. No dwelling shall be occupied until it can be demonstrated that the noise levels noted above at the sample properties have been met.

Reason: To ensure that the occupiers and users of the development do not suffer a loss of amenity by reason of excess noise from neighbouring land uses and transportation sources in accordance with Policy CP1 Sustainable Development.

34. The land immediately adjacent to Allerton Steel on the south side of the link road labelled 'to be determined' on the plan identified as Illustrative Masterplan Scaled, Nod. 0000-0019 Rev A, dated May 2015, and produced by Spawforths, must not be developed or used for Use Class C3: Dwellinghouses or Use Class C4: Houses in Multiple Occupation of the Town and Country Planning (Use Classes) Order 1987 (as amended).

Reason: Occupiers of dwellings in this location would suffer a significant loss of amenity by reason of excess noise from the adjacent industrial use of the land, contrary to the provisions of Policy CP1 Sustainable Development of the Local Plan and the NPPF.

35. Any reserved matters application for sensitive land uses (such as dwellinghouses, sheltered accommodation, and care homes) to the south of the link road (having regard to the terms of Condition 34 of this permission) must be accompanied by a Noise Impact Assessment and Mitigation Plan to demonstrate compliance with internal noise levels identified in BS 8233:2014 - Guidance on Sound Insulation and Noise Reduction for Buildings (or any replacement guidance). The Assessment and Plan must be approved in writing by the Local Planning Authority prior to the commencement of any development of the said land and the mitigation measures approved must be implemented prior to the occupation of any buildings on the said land.

Reason: To ensure that the occupiers and users of the development do not suffer a loss of amenity by reason of excess noise from environmental and transportation sources in accordance with Policy CP1 Sustainable Development and the NPPF.

36. Any reserved matters application for any land use not included in Conditions 34 and 35 of this permission to the south of the link road must be accompanied by a Noise Impact Assessment and Mitigation Plan to demonstrate that the use(s) itself and any associated Heavy Goods Vehicle movements would not increase the internal or external noise levels of sensitive land uses approved across the wider development site above the levels recommended in BS 8233:2014 - Guidance on Sound Insulation and Noise Reduction for Buildings (or any replacement guidance) or BB93 – Acoustic Design of Schools (or any replacement guidance) as relevant. The Assessment and Plan must be approved in writing by the Local Planning Authority prior to the commencement of any development of the said land and the mitigation measures approved must be implemented prior to the commencement of use of any buildings on the said land.

Reason: To ensure that the occupiers and users of the development do not suffer a loss of amenity by reason of excess noise from environmental and transportation sources in accordance with Policy CP1 Sustainable Development and the NPPF.

<u>Retail</u>

37. The floorspace of the individual A1 (Shops) units within the neighbourhood centre must not exceed 500 m² gross unless otherwise approved in writing by the local planning authority.

Reason: To ensure that the neighbourhood centre primarily serves local needs and to maintain major retail provision in the primary retail area boundary of Northallerton Town Centre in line with Local Plan Policies DP21 and DP24, and the provisions of the NPPF.

Highways for any Phase of the Development

38. Detailed Plans of Link Road, Roundabouts and Bridge

Prior to the commencement of construction of any section of the link road, detailed design and construction drawings for the link road must be submitted to and approved in writing by the Local Planning Authority in consultation with the Local Highways Authority. As part of the detailed design process, a formal Road Safety and Non-Motorised User Audit must be undertaken; in addition, the shared footway/cycleway should be relocated to the north side of the link road and the single footway relocated to the south side of the link road unless evidence is provided to demonstrate that this cannot be safely achieved or that the cost in changing the specification of the road reduces the viability of the development. The construction of the link road must be in accordance with the approved details (which are to include any amendments required by the audits which have been undertaken) or such amendments as may be agreed in writing.

Reason: To secure an appropriate highway constructed to an adoptable standard in the interests of highway safety and the amenity and convenience of highway users and to improve the safety of non-motorised users.

39. Prior to the commencement of any development on the site, detailed design and construction drawings for the new roundabouts at Darlington Road and Stokesley Road, must be submitted to and approved in writing by the Local Planning Authority in consultation with the Local Highways Authority. The construction of the roundabouts must be in accordance with the approved details or such amendments as may be agreed in writing.

Reason: To secure an appropriate highway constructed to an adoptable standard in the interests of highway safety and the amenity and convenience of highway users.

40. Prior to the commencement of any work to the bridge or its embankments, detailed design and construction details for the bridge and embankments must be submitted to and approved in writing by the Local Planning Authority in consultation with the Local Highways Authority and Network Rail. The construction of the bridge and embankments must be in accordance with the approved details or such amendments as may be agreed in writing.

Reason: To secure an appropriate highway constructed to an adoptable standard in the interests of highway safety and the amenity and convenience of highway users.

41. Detailed plans of road and footway layout (outline all types)

Unless otherwise approved in writing by the Local Planning Authority, there shall be no excavation or other groundworks, except for investigative works or the depositing of material on the site for each separate phase or sub-phase of the development, until the following drawings and details have been submitted to and approved in writing by the Local Planning Authority in consultation with the Highway Authority:

- a. Detailed engineering drawings to a scale of not less than 1:500 and based upon an accurate survey showing:
 - The proposed highway layout including the highway boundary
 - Dimensions of any carriageway, cycleway, footway, and verges
 - Visibility splays
 - The proposed buildings and site layout, including levels
 - Accesses and driveways and parking
 - Drainage and sewerage system
 - Lining and signing
 - Traffic calming measures
 - All types of surfacing (including tactiles), kerbing and edging
 - Links to the pedestrian and cycle crossing points and bus stops on the link road.
- b. Longitudinal sections to a scale of not less than 1:500 horizontal and not less than 1:50 vertical along the centre line of each proposed road showing:
 - The existing ground level
 - The proposed road channel and centre line levels
 - Full details of surface water drainage proposals.

- c. Full highway construction details including:
 - Typical highway cross-sections to scale of not less than 1:50 showing a specification for all the types of construction proposed for carriageways, cycleways and footways/footpaths
 - When requested cross sections at regular intervals along the proposed roads showing the existing and proposed ground levels
 - Kerb and edging construction details
 - Typical drainage construction details.
- d. Details of the method and means of surface water disposal.
- e. Details of all proposed street lighting.
- f. Drawings for the proposed new roads and footways/footpaths giving all relevant dimensions for their setting out including reference dimensions to existing features.
- g. Full working drawings for any structures which affect or form part of the highway network.
- h. A programme for completing the works.

The development shall only be carried out in full compliance with the approved drawings and details unless agreed otherwise in writing by the Local Planning Authority with the Local Planning Authority in consultation with the Highway Authority.

Reason: To secure an appropriate highway constructed to an adoptable standard in the interests of highway safety and the amenity and convenience of highway users.

42. B8 Uses – Revised Transport Assessment

No part of the site is to be developed for or occupied by B8 (Storage and Distribution) Uses until a revised Transport Assessment including plans illustrating turning circles and visibility splays has been provided to the Council's satisfaction to demonstrate that the link road and any side access roads can safely accommodate the necessary vehicles movements/manoeuvres (specifically any Heavy Goods Vehicles).

Reason: In the interests of the safety and convenience of highway users.

43. Travel Plan

No separate phase or sub-phase of the development, including the school, shall take place until a Phase Travel Plan for that separate phase or sub-phase of the development, based upon the agreed Framework Travel Plan, has been submitted to, and approved in writing by, the Local Planning Authority in consultation with the Local Highway Authority. No separate phase or sub-phase of the development shall be occupied prior to implementation of the Approved Phase Travel Plan (or implementation of those parts identified in the Approved Phase Travel Plan as capable of being implemented prior to occupation). Those parts of the Approved Phase Travel Plan that are identified therein as being capable of implementation after occupation shall be implemented in accordance with the timetable contained therein and shall continue to be implemented as long as any part of the development is occupied.

Reason: In accordance with Development Plan Policy CP2 and to establish measures to encourage more sustainable non-car modes of transport.

<u>Highways – Off-Site Highway Works</u>

44. Approval of details for site works in the highway –

Unless otherwise approved in writing by the Local Planning Authority, there shall be no excavation or other groundworks, except for investigative works, or the depositing of material on the site in connection with the construction of the access road or building(s) or other works until:

- (i) The details of the following off site required highway improvement works, works listed below have been submitted to and approved in writing by the Local Planning Authority in consultation with the Local Highway Authority:
 - a. Amendments to the existing Stonecross mini-roundabout to improve capacity within the existing highway boundary
 - b. Amendments to the existing mini-roundabout at the western end of Quaker Lane to improve capacity within the existing highway boundary

c. Amendments to the existing mini-roundabout at the eastern end of Quaker Lane to improve pedestrian facilities following the opening of the link road to through traffic.

The aforementioned amendments should be generally in accordance with the plans numbered 12003/GA/01 Rev B, 12003/GA/02 Rev A and 12003/GA/03 Rev B, prepared by Optima Highways and submitted with the application or subsequent plans approved by the Local Planning Authority after consultation with the Local Highway Authority.

- (ii) An independent Stage 2 Road Safety Audit for the agreed off site highway works has been carried out in accordance with HD19/15 - Road Safety Audit or any superseding regulations and the recommendations of the Audit have been addressed in the proposed works.
- (iii) A programme for the completion of the proposed works has been submitted to and approved writing by the Local Planning Authority in consultation with the Local Highway Authority.

Reason: In the interests of the safety and convenience of highway users.

45. Completion of works in the highway before occupation of the 51st dwelling

Unless otherwise approved in writing by the Local Planning Authority in consultation with the Highway Authority, the following off site highway works shall have been constructed in accordance with the details approved in writing by the Local Planning Authority under Condition Numbers 44(i)(a) and 44(i)(b) prior to the occupation of the 51st dwelling:-

- a. Amendments to the existing Stonecross mini-roundabout to improve capacity within the existing highway boundary
- b. Amendments to the existing mini-roundabout at the western end of Quaker Lane to improve capacity within the existing highway boundary

Reason: In the interests of the safety and convenience of highway users.

46. Completion of works in the highway within six months of opening of the Link Road to through traffic

Unless otherwise approved in writing by the Local Planning Authority in consultation with the Highway Authority, the amendments to the existing mini-roundabout at the eastern end of Quaker Lane to improve pedestrian facilities shall have been constructed in accordance with the details approved in writing by the Local Planning Authority under Condition Number 44(i)(c) of this permission within six months of opening of the Link Road to through traffic.

Reason: In the interests of the safety and convenience of highway users.

<u>Highways – For the Phase 1 Residential East of Darlington Road (A167)</u>

47. Construction of roads and footways prior to occupation of dwellings

No dwelling in this Phase or sub-phase to which this planning permission relates shall be occupied until the carriageway and any footway/footpath from which it gains access and all roads linking it back to the existing adopted highway are constructed to basecourse macadam level and/or block paved and kerbed with street lighting installed and in operation.

The completion of all road works, including any phasing, shall be in accordance with a programme approved in writing by the Local Planning Authority in consultation with the Highway Authority before the first dwelling of this phase is occupied, and in accordance with Condition 41(h).

Reason: To ensure safe and appropriate access and egress to the dwellings, in the interests of highway safety and the convenience of prospective residents.

48. Parking for dwellings

No dwelling shall be occupied until the related parking facilities for that dwelling approved under Condition 41 have been constructed in accordance with the approved drawing. Once created these parking areas shall be maintained clear of any obstruction and retained for their intended purpose at all times.

Reason: To provide for adequate and satisfactory provision of off-street accommodation for vehicles in the interest of safety and the general amenity of the development.

49. Garage conversion to habitable room

Notwithstanding the provisions of the Town and Country Planning General Permitted Development Order 1995 or any subsequent Order, the garage(s) shall not be converted into domestic accommodation without the granting of an appropriate planning permission.

Reason: To ensure the retention of adequate and satisfactory provision of off-street accommodation for vehicles generated by occupiers of the dwelling and visitors to it, in the interest of safety and the general amenity the development.

Highways - For the Phase 1 Residential West of Stokesley Road (A684)

50. Construction of roads and footways prior to occupation of dwellings

No dwelling in this phase or sub-phase to which this planning permission relates shall be occupied until the carriageway and any footway/footpath from which it gains access and all roads linking it back to the existing adopted highway are constructed to basecourse macadam level and/or block paved and kerbed with street lighting installed and in operation.

The completion of all road works, including any phasing, shall be in accordance with a programme approved in writing by the Local Planning Authority in consultation with the Highway Authority before the first dwelling of this phase is occupied, and in accordance with Condition 41(h).

Reason: To ensure safe and appropriate access and egress to the dwellings, in the interests of highway safety and the convenience of prospective residents.

51. Closing Brompton Road to Through Traffic

No dwelling in **in this phase or sub-phase** to which this planning permission relates shall be occupied until a programme for the closing of Brompton Road to through vehicular traffic has been submitted to and approved in writing with the Local Planning Authority in consultation with the Highway Authority. The programme shall include the provision of an alternative route linking Brompton Road north to Stokesley Road via the link road and roundabout and the maintaining of direct pedestrian and cycling routes. Brompton Road shall be closed to through vehicular traffic in accordance with this programme.

Reason: To ensure safe and appropriate access and egress to the dwellings, in the interests of highway safety and the convenience of prospective residents.

52. Parking for dwellings

No dwelling shall be occupied until the related parking facilities for that dwelling approved under Condition 41 have been constructed in accordance with the approved drawing Once created these parking areas shall be maintained clear of any obstruction and retained for their intended purpose at all times.

Reason: To provide for adequate and satisfactory provision of off-street accommodation for vehicles in the interest of safety and the general amenity of the development.

53. Garage conversion to habitable room

Notwithstanding the provisions of the Town and Country Planning General Permitted Development Order 1995 or any subsequent Order, the garage(s) shall not be converted into domestic accommodation without the granting of an appropriate planning permission.

Reason: To ensure the retention of adequate and satisfactory provision of off-street accommodation for vehicles generated by occupiers of the dwelling and visitors to it, in the interest of safety and the general amenity the development.

Highways - Outline areas of the site (Residential)

54. Construction of roads and footways prior to occupation of dwellings

No dwelling in this separate phase or sub-phase to which this planning permission relates shall be occupied until the carriageway and any footway/footpath from which it gains access and all roads linking it back to the existing adopted highway are constructed to basecourse macadam level and/or block paved and kerbed with street lighting installed and in operation.

The completion of all road works, including any phasing, shall be in accordance with a programme approved in writing by the Local Planning Authority in consultation with the Highway Authority before the first dwelling of this phase is occupied in accordance with Condition 41(h).

Reason: To ensure safe and appropriate access and egress to the dwellings, in the interests of highway safety and the convenience of prospective residents.

55. Parking for dwellings

No dwelling shall be occupied until the related parking facilities approved under Condition 41 have been constructed in accordance with the approved drawing. Once created these parking areas shall be maintained clear of any obstruction and retained for their intended purpose at all times.

Reason: To provide for adequate and satisfactory provision of off-street accommodation for vehicles in the interest of safety and the general amenity of the development.

56. Garage conversion to habitable room

Notwithstanding the provisions of the Town and Country Planning General Permitted Development Order 1995 or any subsequent Order, the garage(s) shall not be converted into domestic accommodation without the granting of an appropriate planning permission.

Reason: To ensure the retention of adequate and satisfactory provision of off-street accommodation for vehicles generated by occupiers of the dwelling.

57. <u>Doors and windows opening over the highway</u>

All doors and windows on elevations of the building(s) adjacent to the existing and/or proposed highway shall be constructed and installed such that from the level of the adjacent highway for a height of 2.4 metres they do not open over the public highway and above 2.4 metres no part of an open door or window shall come within 0.5 metres of the carriageway. Any future replacement doors and windows shall also comply with this requirement.

Reason: To protect pedestrians and other highway users.

Highways - Outline areas of the site (non-Residential)

58. Construction of roads and footways (non-residential)

No part of any separate phase or sub-phase of the development to which this permission relates shall be brought into use until the carriageway and any footway/footpath from which it gains access shall be constructed to basecourse macadam level and/or block paved and kerbed and connected to the existing highway network with street lighting installed and in operation.

The completion of all road works, including any phasing, shall be in accordance with a programme approved in writing by the Local Planning Authority in consultation with the Highway Authority before the first dwelling of this phase is occupied, and in accordance with Condition 41(h).

Reason: To ensure safe and appropriate access and egress to the premises, in the interests of highway safety and the convenience of prospective users of the highway.

59. Details of access, turning and parking

Unless otherwise approved in writing by the Local Planning Authority, there shall be no excavation or other groundworks, except for investigative works, or the depositing of material on any non-residential phase of site in connection with the construction of the access road or building(s) or other works hereby permitted until full details of the following have been submitted to and approved in writing by the Local Planning Authority in consultation with the Highway Authority:

- Tactile paving
- b. Vehicular, cycle, and pedestrian accesses
- vehicular and cycle parking
- d. Vehicular turning arrangements

- e. Manoeuvring arrangements
- f. Loading and unloading arrangements.

Reason: To ensure safe and appropriate access and egress to the premises, in the interests of highway safety and the convenience of prospective users of the highway.

60. Provision of approved access, turning and parking areas

No part of any non-residential phase of the development shall be brought into use until the approved vehicle access, parking, manoeuvring and turning areas approved under Condition 41 have been constructed in accordance with the approved drawings. Once created these areas shall be maintained clear of any obstruction and retained for their intended purpose at all times

Reason: To ensure appropriate on-site facilities in the interests of highway safety and the general amenity of the development.

61. Doors and windows opening over the highway

All doors and windows on elevations of the building(s) adjacent to the existing and/or proposed highway shall be constructed and installed such that from the level of the adjacent highway for a height of 2.4 metres they do not open over the public highway and above 2.4 metres no part of an open door or window shall come within 0.5 metres of the carriageway. Any future replacement doors and windows shall also comply with this requirement.

Reason: To protect pedestrians and other highway users.

Informatives

Conditions 21 - 22: Main River Consent

Formal consent from the Environment Agency will be required for any works in, over, under or within 8m of a main river and/or flood defence. The consent must be obtained before any works commence.

Conditions 21 - 22: Surface Water

All surface water drainage details must be agreed with North Yorkshire County Council and the Internal Drainage Board before development commences. Where there are known issues of either surface water flooding or localised flooding from watercourses classified as non main river (that are outside Flood Zones 2 and 3) the applicant must agree any necessary flood compensation with the IDB and North Yorkshire County Council in their role as lead local flood authority.

Conditions 21 - 22: Flooding/Erosion

It is the applicant's responsibility to ensure that their operations do not cause or exacerbate flooding problems and/or erosion problems for others as a result of their works.

Condition 26: Public Rights of Way

Applicants are advised to contact the County Council's Access and Public Rights of Way team at County Hall, Northallerton via paths@northyorks.gov.uk to obtain up to date information regarding the line of the route of the way. The applicant should discuss with the Highway Authority any proposals for altering the route.

Condition 41: Detailed plans of road and footway layout (outline all types)

In imposing condition number 41 it is recommended that before a detailed planning submission is made a draft layout is produced for discussion between the applicant, the Local Planning Authority and the Highway Authority in order to avoid abortive work. The agreed drawings must be approved in writing by the Local Planning Authority for the purpose of discharging this condition.

Condition 59: Provision of approved access, turning and parking areas

The proposals shall cater for all types of vehicles that will use the site. The parking standards are set out in the North Yorkshire County Council publication 'Transport Issues and Development - A Guide' available at www.northyorks.gov.uk

Parish: Dalton Committee Date: 12 November 2015

Ward: Sowerby & Topcliffe Officer dealing: Mr T J Wood

2

15/01324/FUL

Construction of dwellinghouse at Dalton Lane, Dalton, North Yorkshire for Mr J Binks

1.0 SITE DESCRIPTION AND PROPOSAL

- 1.1 This scheme is a re-submission of a withdrawn application 15/00259/FUL which was brought before Members earlier this year. The scheme was withdrawn to allow the applicant to consider the justification for the principle of the dwelling, and to allow them opportunity to submit a concurrent application for the change of use of adjacent land for the storage of plant and machinery (15/01323/FUL). The application was further deferred from consideration for the submission of further information by the applicant and others in support of the case.
- 1.2 This application refers to the construction of a replacement dwellinghouse. The applicant has presented evidence as part of their application submission to suggest that a previous permission on the site 2/37/56B/PA was commenced (the foundations were set) but was not completed. Application 2/37/56B/PA related to a 3 bedroom agricultural workers dwelling. The proposed dwelling would sit partly on the foundations in situ. It is acknowledged that the previously approved scheme has been commenced and whilst substantially smaller than the proposed dwelling by using 'permitted development' rights under the General Permitted Development Order there is scope to extended the previously approved dwelling, once it has been completed and occupied, to form an extended dwelling that may be suitable to meet the stated needs of the applicant.
- 1.3 The application site is to the south-west of the periphery of Dalton and sits to the south-west of an access track which previously formed the perimeter road to the former airfield. Via the public highway the site is approximately 0.5miles away from the centre of Dalton although it is acknowledged that a green lane may permit a more direct access to the settlement.
- 1.4 The topography of the land is flat. Disused agricultural and former aerodrome structures sit immediately to the north and west of the application site. Agricultural land extends to the south.
- 1.5 The occupier of the proposed dwelling is a plant and machinery contractor which includes agricultural contracting. The applicant's agent has provided to the Council on 11 May 2015 an indication of the applicant's local client base. The applicant currently resides in Topcliffe.
- 1.6 The dwelling would be formed in an 'L' shape. Amended plans have been received on 8 May 2015 reducing the width of the two storey projection to the northern elevation. The dwelling would provide for a cloakroom, kitchen, living/dining area, playroom, hall, office, sitting room, WC/shower room and a utility area at ground floor level. At first floor the dwelling would comprise 6 bedrooms, a bathroom and a WC/shower room.
- 1.7 The dwelling, as amended, would have maximum dimensions of 18.3m x 16.5m, with

- a total height of approximately 8.5m. Materials for the proposed structure would comprise handmade bricks and clay pantiles.
- 1.8 Access to the proposed dwelling would be taken via a proposed track to the north which would connect to Dalton Lane. The track would be surfaced in gravel on hardcore and would be bounded by a post and rail fence and landscaping.

2.0 RELEVANT PLANNING AND ENFORCEMENT HISTORY

- 2.1 2/37/56B/PA Revised application for the siting and details of the construction of an agricultural workers bungalow with domestic garage at Part OS 7900; Granted 1985.
- 2.2 15/00259/FUL Construction of replacement dwellinghouse; Withdrawn 8 June 2015.
- 2.3 15/01323/FUL Change of use of agriculture land to use as storage area for implements and machinery in association with groundworks and agricultural contracting business; Pending Consideration.

3.0 NATIONAL AND LOCAL POLICY

3.1 The relevant policies are:

Core Strategy Policy CP1 - Sustainable development

Core Strategy Policy CP2 - Access

Core Strategy Policy CP4 - Settlement hierarchy

Core Strategy Policy CP15 - Rural Regeneration

Core Strategy Policy CP16 - Protecting and enhancing natural and man-made assets

Core Strategy Policy CP17 - Promoting high quality design

Core Strategy Policy CP19 - Recreational facilities and amenity open space

Core Strategy Policy CP21 - Safe response to natural and other forces

Development Policies DP1 - Protecting amenity

Development Policies DP2 - Securing developer contributions

Development Policies DP3 - Site accessibility

Development Policies DP4 - Access for all

Development Policies DP6 - Utilities and infrastructure

Development Policies DP10 - Form and character of settlements

Development Policies DP26 - Agricultural issues

Development Policies DP30 - Protecting the character and appearance of the countryside

Development Policies DP32 - General design

Development Policies DP37 - Open space, sport and recreation

Development Policies DP43 - Flooding and floodplains

Interim Guidance Note - adopted by Council on 7th April 2015

National Planning Policy Framework

4.0 CONSULTATIONS

- 4.1 Parish Council no objections but wish to see Agricultural Occupancy Restrictions added to the conditions should permission be granted.
- 4.2 Highway Authority no objection.
- 4.3 Environmental Health Officer no objection; conditions are recommended regarding septic tank installation and maintenance.
- 4.4 Environmental Health Scientific Officer The above application is for a dwelling on the site of a former World War Two airfield and therefore there is the potential for

contamination and/or unexploded ordnance to exist. I order to determine the risk from contamination and/or ordnance I would recommend conditions, relating to contaminated land and unexploded ordnance should planning approval be granted.

- 4.5 Ministry of Defence no response
- 4.6 Yorkshire Water no response (Previous response: Based on the information submitted, no comments required).
- 4.7 Neighbours notified and site notice posted no responses.

5.0 OBSERVATIONS

- 5.1 The main issues for consideration in this case relate to (i) the principle of new dwellings in this location, outside Development Limits, together with an assessment of the likely impact of the proposed dwelling on the character and appearance of the village, (ii) drainage, (iii) highway safety, (iv) neighbour amenity and (v) developer contributions.
- 5.2 There are four factors to be balanced in this case relating to the principle of the proposal: 1) the fall-back position of the commenced dwelling on site and whether it is a feasible alternative to the dwelling proposed; 2) whether the employment of the applicant as mainly an agricultural contractor is a justifying factor for the dwelling outside of settlement limits; 3) whether the implications of the Interim Policy Guidance for dwellings outside of development limits; and 4) whether the development constitutes sustainable development.
- 5.3 The commenced dwelling is a three bedroom property which is smaller in scale than the proposed dwelling and would not provide the level of accommodation which the applicant is seeking. The feasibility of it being completed has been considered to be relatively low, particularly as it would need to be heavily extended to meet the needs of the applicant. Evidence has been since been supplied from a local surveyor (agricultural specialist) that explains that there would be demand for an agricultural workers dwelling as shown in the approved plans. The likelihood of the dwelling being built, if the proposed scheme does not go ahead, is now considered to be high. The approved bungalow was approved subject to an agricultural occupancy condition. The evidence supplied by the applicant, their agent and accountant shows that the applicant is mainly employed in agricultural contracting. It is reasonable to conclude that the applicant would meet the occupancy restriction that applied to the approved bungalow. The additional information shows that considerable weight should be given to the fall back position in the planning balance.
- 5.4 The applicant has provided supporting information regarding their contracting enterprise, the extent of agriculture in this enterprise, and their client base in relation to the application site. The application site is relatively central to their contracting operations locally. The applicant currently lives at Topcliffe and keeps a limited amount of machinery there. Some machinery is currently stored at a relation's property in Dalton with the remainder moved from site to site. The application under reference 15/01323/FUL has been approved for storage on adjacent land. There is evidence to suggest that a new dwelling adjacent to the storage site would aid the efficient operation of the business. None of the details would amount to an essential requirement to locate in the countryside. It is, however, not doubted that the applicant's business would continue to support the sustainable rural economy. The applicant's business also features low in the planning balance.
- 5.5 The third issue relates to the Interim Policy Guidance (IPG) which, as of 7 April 2015 ensures appropriate consistent interpretation of the NPPF alongside LDF policies

CP4 and DP9 for housing development outside of settlement limits. It is considered that the location of the approved and proposed dwelling would not respect the form of the dwelling. Either the previously approved dwelling or the proposed dwelling would support the few local services in the nearby villages.

- 5.6 The proposed dwelling is larger and would have greater visual impact in the landscape than the approved dwelling. The agricultural worker's dwelling was a three-bedroom bungalow, whereas the proposed dwelling would comprise six bedrooms over two floors, with a footprint approximately double that of the bungalow. the proposal would not harm the natural, built or historic environment. The dwelling would have an impact, but not a detrimental one, on the open character and appearance of the surrounding countryside and it would not lead to a coalescence of settlements. The dwelling could be accommodated within the capacity of the existing infrastructure. Therefore, if acceptable in principle, the development has the capacity to conform to the remaining LDF policies. However, it is clear that the development is not compliant with the IPG due to its distance from Dalton's existing built form. Accordingly it is considered that the proposed dwelling can take no support from the IPG in the planning balance.
- 5.7 The consideration of the scheme in terms of whether it represents sustainable development in terms of the NPPF definition is a further material consideration. The future occupants of the dwelling would perform a limited economic role. The scheme would perform a social role in supporting the community of Dalton and helping to create a high quality built environment.
- 5.8 Turning to issues of detail, the dwelling would not harm local visual amenity and is thus acceptable in terms of policy DP30. It would not erode the amenities of occupiers of adjacent property and is acceptable on that ground. Furthermore, it would not raise any highway safety issues.
- 5.9 It is considered on balance the demonstration of the strength of the fall back position presented by the existing consent for the dwelling, the evidence that the occupation would comply with the agricultural occupancy condition presented by the applicant and his agent and that the scheme does not breach the NPPF definition of sustainable development. These factors are of such weight that the fact that the scheme cannot take support from the IPG is inconsequential.

6.0 RECOMMENDATION:

- 6.1 That the application is **GRANTED** subject to the following conditions:
 - 1. The development hereby permitted shall be begun within three years of the date of this permission.
 - 2. The permission hereby granted shall not be undertaken other than in complete accordance with the drawings and details received by Hambleton District Council on 10 June 2015 as amended by the drawings and details received by Hambleton District Council on 8 May 2015 unless otherwise approved in writing by the Local Planning Authority.
 - 3. The development hereby approved shall not be constructed other than of the materials detailed on the application form received by Hambleton District Council on 10 June 2015 unless otherwise agreed in writing by the Local Planning Authority.
 - 4. The development shall not be commenced until a detailed landscaping scheme indicating the type, height, species and location of all new trees and

shrubs, has been submitted to and approved by the Local Planning Authority. No part of the development shall be used after the end of the first planting and seeding seasons following the approval of the landscaping scheme, unless the approved scheme has been completed. Any trees or plants which within a period of 5 years of planting die, are removed, or become seriously damaged or diseased, shall be replaced with others of similar size and species.

- 5. The site shall be developed with separate systems of drainage for foul and surface water.
- 6. The development hereby approved shall not be commenced until details of the foul sewage and surface water disposal facilities have been submitted and approved in writing by the Local Planning Authority.
- 7. The use of the development hereby approved shall not be commenced until the foul sewage and surface water disposal facilities have been constructed and brought into use in accordance with the details approved under condition 7 above.
- 8. No development shall be commenced until an assessment of the risks posed by contamination, carried out in line with the Environment Agency's Model Procedures for the Management of Land Contamination CLR11, has been submitted to and approved by the Local Planning Authority. A scheme for the remediation of any contamination shall be submitted and approved by the Local Planning Authority before any development occurs. The development shall not be occupied until the approved remediation scheme has been implemented and a verification report detailing all works carried out has been submitted to and approved in writing by the Local Planning Authority.
- 9. If unexploded ordnance (UXO) is found or suspected at any time during development all works shall cease and the Local Planning Authority shall be notified in writing immediately. No further works shall be undertaken or the development occupied until a risk assessment carried out by an unexploded ordnance (UXO) risk assessment specialist has been submitted to and approved in writing by the Local Planning Authority. Where risk mitigation is necessary a scheme for the mitigation of the UXO shall be submitted and approved by the Local Planning Authority before any further development occurs. The development shall not be occupied until the approved risk mitigation scheme has been implemented and a verification report and clearance certificate has been submitted to and approved in writing by the Local Planning Authority.
- 10. The development shall not be commenced until details relating to boundary walls, fences and other means of enclosure for all parts of the development have been submitted to and approved in writing by the Local Planning Authority.
- 11. No dwelling shall be occupied until the boundary walls, fences and other means of enclosure have been constructed in accordance with the details approved in accordance with condition 11 above. All boundary walls, fences and other means of enclosure shall be retained and no part thereof shall be removed without the prior consent of the Local Planning Authority.
- 12. Prior to development commencing detailed cross sections shall be submitted to and approved in writing by the Local Planning Authority, showing

the existing ground levels in relation to the proposed ground and finished floor levels for the development. The levels shall relate to a fixed Ordnance Datum. The development shall be constructed in accordance with the approved details and thereafter be retained in the approved form.

The reasons are:-

- 1. To ensure compliance with Sections 91 and 92 of the Town and Country Planning Act 1990 and where appropriate as amended by Section 51 of the Planning and Compulsory Purchase Act 2004.
- 2. In order that the development is undertaken in a form that is appropriate to the character and appearance of its surroundings and in accordance with the Development Plan Policy(ies) CP1, CP16, CP17, DP1, DP30 and DP32.
- 3. To ensure that the external appearance of the development is compatible with the immediate surroundings of the site and the area as a whole in accordance with Hambleton Local Development Framework Policy CP17.
- 4. In order to soften the visual appearance of the development and provide any appropriate screening to adjoining properties in accordance with Local Development Framework Policy DP30.
- 5. In the interest of satisfactory drainage and to avoid pollution of the water environment.
- 6. In order to avoid the pollution of watercourses and land in accordance with Local Development Framework CP21 and DP43.
- 7. In order to avoid the pollution of watercourses and land in accordance with Local Development Framework CP21 and DP43.
- 8. To assess and mitigate against the risks posed by contamination.
- 9. To assess and mitigate against the risks posed by unexploded ordnance.
- 10. To protect the amenity of the neighbouring residents and to ensure that the development is appropriate to the character and appearance of its surroundings.
- 11. To protect the amenity of the neighbouring residents and to ensure that the development is appropriate to the character and appearance of its surroundings.
- 12. To ensure that the development is appropriate in terms of amenity in accordance with Local Development Framework Policies CP1 and DP1.

Parish: Easingwold

Ward: Easingwold

3

Committee Date: 12 November 2015 Officer dealing: Mr T J Wood Target Date: 24 February 2015

14/02285/FUL

Construction of 116 dwellings with associated access, open spaces and landscaping. at Land to the East of Kellbalk Lane, Easingwold for Persimmon Homes (Yorkshire) Ltd.

1.0 SITE DESCRIPTION AND PROPOSAL

- 1.1 Members resolved to defer this application at the meeting on 23 July 2015 in order to receive further information on drainage and to enable the applicant to consider the request of the Committee to include more bungalows in the scheme and for consideration to be given to the hours of working on the site. Since the deferral a flood event in Easingwold has raised further questions relating to the adequacy of the drainage systems and a further response of Yorkshire Water has been received.
- 1.2 The applicant has provided additional drainage details and details of the recently recorded flooding events compiled by those in Easingwold have been supplied to Yorkshire Water and comment has been received from Yorkshire Water. These are reported with comment in paragraphs 4.3 and 5.9 below concluding that a conditional approval is appropriate.
- 1.3 Since the previous consideration of this application additional bungalows have been proposed, 10% of the new dwellings proposed are bungalows. The scheme continues to provide 50% of the dwellings as affordable dwellings in perpetuity controlled by a Section 106 Agreement. These issues are considered in paragraphs 5.13 5.18.
- 1.4 Additional consultation has been undertaken following the receipt of amended plans and the responses of Easingwold Town Council and neighbours are reported at paragraphs 4.1 and 4.11. Additional comment is made in paragraph 5.44 and 5.45 in respect of neighbour amenity. An hours of work condition can be recommended, a condition relating to the management of the deliveries to the site is part of condition 3.
- 1.5 Other than as set out in the preceding paragraphs the remainder of this report is unchanged from that presented to the Planning Committee on 23 July 2015.
- 1.6 The proposal seeks to develop two sites that are allocated in the LDF for housing, the southern part of EH2 and all of EH3. EH2 and EH3 are greenfield sites, neither is active agricultural production. EH2 is an overgrown tract of land between housing estates. EH3 is a mix of grassland, scrub and unmaintained woodland that lies between housing on Kellbalk Lane and actively farmed land to the east of Easingwold.
- 1.7 The land is higher at the northern end, falling more gently at the southern end, but is however generally flat throughout. Many trees and hedgerows enclose and subdivide the land in to a series of 4 main parcels.
- 1.8 There are public rights of way to the southern edge of EH2, continuing across the northern edge of the site EH3. There are many informal walked routes through the body of EH3 and a route that breaches the eastern hedgerow to run parallel with the site boundary finally emerging on to Crabmill Lane through a narrow gap in the

- hedgerow. The line of Kellbalk Lane is grassed; a parallel tarmac path runs the length of EH2 and EH3 and links Crabmill Lane with Back Lane.
- 1.9 The application originally sought permission for 134 units but following revisions to address concerns about the layout, to protect trees on the boundaries, to provide parking close to dwellings and improve amenity the number of units has been reduced to 116. This results in a density of 33.4 dwellings per hectare.
- 1.10 Three areas of public open spaces would be provided within the site giving a combined area of 5,700 sq. m (0.57 hectares, 1.4 acres). An emergency link road is shown to be proposed to connect between the development of Meadow Springs Way across the Kellbalk Lane public footpath to Oxenby Place. There is currently a public footpath connection from Kellbalk Lane to the highway and footways in Oxenby Place.
- 1.11 The proposal would obstruct informal walking routes within the land to the east of Kellbalk Lane. Footpath connections are proposed to be retained to Kellbalk Lane and to the field edge path to the east of the development site. The definitive rights of way are not obstructed and no pubic rights of way need to be diverted.
- 1.12 A Tree Preservation Order 15/00009/TPO has been made to protect important 3 groups and 11 individual trees that have been identified as important to the site (and as defined within the submitted tree report) where these trees are within and close to the boundaries of the application site.

2.0 RELEVANT PLANNING AND ENFORCEMENT HISTORY

2.1 There is no relevant planning or enforcement history relating to the application site. To the north of the Crabmill Lane part of the application site a full planning application for the construction of 22 dwellings, associated access and provision of public open space was approved on 31 July 2014. (14/00406/FUL) The scheme provided 11 affordable dwellings (50% of the total) and 2 bungalows. The construction works are underway.

3.0 RELEVANT PLANNING POLICIES

3.1 The relevant policies are:

Core Strategy Policy CP1 - Sustainable development

Core Strategy Policy CP2 - Access

Core Strategy Policy CP8 - Type, size and tenure of housing

Core Strategy Policy CP9 - Affordable housing

Core Strategy Policy CP9A - Affordable housing exceptions

Core Strategy Policy CP16 - Protecting and enhancing natural and man-made assets

Core Strategy Policy CP17 - Promoting high quality design

Core Strategy Policy CP18 - Prudent use of natural resources

Core Strategy Policy CP19 - Recreational facilities and amenity open space

Core Strategy Policy CP21 - Safe response to natural and other forces

Development Policy DP1 - Protecting amenity

Development Policy DP2 - Securing developer contributions

Development Policy DP3 - Site accessibility

Development Policy DP4 - Access for all

Development Policy DP6 - Utilities and infrastructure

Development Policy DP8 - Development Limits

Development Policy DP13 - Achieving and maintaining the right mix of housing

Development Policy DP15 - Promoting and maintaining affordable housing

Development Policy DP29 - Archaeology

Development Policy DP30 - Protecting the character and appearance of the countryside

Development Policy DP31 - Protecting natural resources: biodiversity/nature conservation

Development Policy DP32 - General design

Development Policy DP33 - Landscaping

Development Policy DP34 - Sustainable energy

Development Policy DP37 - Open space, sport and recreation

Development Policy DP39 - Recreational links

Development Policy DP43 - Flooding and floodplains

Affordable Housing - Supplementary Planning Document - Adopted 7 April 2015

Sustainable Development - Supplementary Planning Document - Adopted 7 April 2015

Allocations Document Policy EH2 - Kellbalk Lane and East of Oxenby Place, Easingwold - adopted 21 December 2010

Allocations Document Policy EH3 - North of Meadow Spring Way, Easingwold - adopted 21 December 2010

National Planning Policy Framework

National Planning Practice Guidance

4.0 CONSULTATION

4.1 Easingwold Town Council – Supports the principle of development of the site but notes great concern regarding the infrastructure and services to foul and surface water from this site. Support 50% policy for delivery of affordable houses for local people and request that at least 10% of all dwellings are bungalows.

The Town Council wish to make further representations when we know the reports from the statutory consultations and any amendments made to the application. Given the current development we have concerns about the capacity of the school to accommodate these extra numbers and wish to see highway improvements at the junction on Stillington Road, Long Street and Crabmill Lane. There are 2 public right of ways 10.40/14 and 10.40/16 please retain these as public right of ways.

Following re-consultation after the 23 July Planning Committee the following response has been received:

Wish to see **REFUSED** until the issue of the disposal of surface water is addressed to the satisfaction of Easingwold Town Council. Wish to ensure safe pedestrian access through footpaths.

Further response received

Wish to see **REFUSED** for the following reasons:

- The Social Infrastructure of Easingwold is not in a position to accommodate further large population increases at this time. Bringing forward this development is detrimental to the Town and its services. A requirement of Sustainable development is that it should maintain or enhance the vitality of the local community. This application, at this time, will not do that.
- 2. The recent heavy storms showed up some of the deficiencies of the surface water rainfall disposal for Easingwold. Much of the water from North of Stillington Road outfalls across Stillington Road into Leasmire Beck which runs in-between the current housing development and the Business Park. This is in open ditch for some way until the route turns sharply in 2 directions

- and then enters a pipe across private land causing a restriction in its capacity & the potential for flooding.
- 3. The Kyle & Ouse Drainage Board require that developers do not discharge "more than normal agricultural flows" into their drainage ditches. This can be accommodated by the developers by building-in storage facilities and discharging slowly over time, but in storm conditions when the system overloads quickly this does not work.
- 4. The drainage infrastructure for Easingwold cannot currently accommodate the level of development proposed without significant upgrade.
- 5. During the Neighbourhood Plan consultations earlier this year discussions were held with the Doctors from Millfield Surgery, the Dentists and the Pharmacy. It was clear from these service providers that Easingwold has a serious problem as new housing grows faster than the availability of local Social Infrastructure. Millfield Surgery was stretched, Red Lea Dentists has adopted a no more NHS patients policy and has now discarded their waiting lists and the Pharmacy is too small.
- 6. The Doctors from Millfield Surgery have recently written to the Planning Inspectorate with regards to the new Gladman planning appeal. They explain clearly that "Easingwold has seen a significant increase in its population in recent years as a result of a number of large scale developments" "At the present time health care provision is working at or close to capacity". The comments contained in their letter equally apply to the Kellbalk Lane application. The applicants for this planning application have not held consultation with the Doctors about their plan to introduce a further 2 -300 people to the Town.
- 7. The HDC Local Development Framework document (LDF) Adopted 3 April 2007 set out Settlement Proposals for Easingwold. This was reviewed in 2010 and the Allocations for Sites for Housing & Mixed Use was Adopted 21 Dec 2010. This included housing in 3 phases to maintain the on-going HDC 5 year supply of housing.
- 8. Phase 2 was 2016-21 & Phase 3 was defined as "long term" for the period 2021 2026 which included land East of Oxenby Place and East of Kellbalk Lane, which is the site in this planning application.
- This application brings forward the development, as did Oxenby Place and represents the last allocation contained within the HDC allocation of sites for Easingwold.
- 10. During the NP Public Consultation Phase 2 in March 2015 the public view was that future housing development should be "responsible & planned" and growth should be "matched by facilities".
- 11. The Town Council is asked to oppose this Kellbalk Lane application at this time on the grounds that it is premature in the Local Plan and its timing detrimental to the well-being of the Town.
- 4.2 Highway Authority Advises that the design standard applied is Manual for Streets and that the required visibility splays of 45 x 2.4 metres are available. A request has been made to provide an emergency access to the Crabmill Lane site or alternatively a widening of the carriageway to facilitate access from Crabmill Lane for emergency

vehicles. In the absence of detailed highway designs for new estate roads it is anticipated that conditions are recommended.

4.3 Yorkshire Water – Advise that, further to recent communications, the submitted Geoenvironmental Appraisal (prepared by Lithos Consulting dated March 2015) confirms sub-soil conditions do not support the use of soakaways because the water table was encountered at a shallow depth. Yorkshire Water asks the developer to give consideration to use of a watercourse which is understood to be located some 100m to the south-east of the site. If the watercourse can be proven not reasonably practical, then Yorkshire Water will have to consider disposal to sewer (at a restricted rate) in line with surface water disposal hierarchy.

Advise that following receipt of additional information from Alan Wood and Partners (contained in an email submissions of 20 April 2015 and 29 June 2015) that the proposals are acceptable. They advise that the following statement supersedes the response of 2 January 2015:

It is noted that there are no watercourses within the vicinity of the development. Curtilage surface water may therefore discharge to public 225 mm diameter surface water sewer located in Kell Balk Lane adjacent to the site, at a restricted rate so as not to exceed 5 (five) litres per second.

A condition is recommended to achieve this specification. Yorkshire Water had also previously sought a condition to protect a water main, however they have now confirmed that the pipe is outside the application site and no planning condition is sought to protect the asset.

Yorkshire Water have, following requests for additional advice following the flooding events in the town during August and September 2015, confirmed that conditions are required relating to the approval of the drainage schemes and that the reports and documents relating to flood events has not changed that advice.

- 4.4 Natural England Provide advice and no objection to the proposal.
- 4.5 Historic England (Formerly English Heritage) no objection.
- 4.6 NYCC Archaeology identifies the need for additional survey work as there is potential for archaeology on the site noting the excavations for the Easingwold bypass found previously unidentified late Iron Age settlement and Romano-British field system.
- 4.7 North Yorkshire Police Advise that rear parking courts should be removed and the areas should be overlooked. (Note: amendments to the layout have addressed many of these areas of concern.) Areas of open space close to homes provide a venue for anti-social behaviour and the distinction between private and public spaces is uncertain in some areas and can result in a loss of privacy (such as in the areas around plots 1 and 3). Recommends conditions on aspect such as secure cycle stores, lighting over external doors.
- 4.8 Environment Agency Acknowledge proposal to direct foul and surface water to the public sewer and raise no objection to the approach due to the likely impermeability of the ground.
- 4.9 Forest of Galtres Society Raise concern regarding overflowing sewers, flooding, path network, landscaping and site layout.

- 4.10 Environmental Health This service has considered the potential impact on amenity and likelihood of the development to cause a nuisance and consider that there will be no negative impact. Therefore the Environmental Health Service has no objections
- 4.11 Public consultation. Comments have been received raising a range of issues as summarised below:
 - Potential obstruction of public rights of way
 - Concern about the increasing number of homes and rate of development in Easingwold
 - Infrastructure has not increased in pace with development no additional shops, garages, schools, doctors, dentists, no upgrades to roads.
 - Wet and unstable ground conditions exist and query the need for large amounts of material to be brought in to provide suitable foundations
 - Land is liable to flooding and the development will increase the risk of flooding
 - Surface water drainage must be provided that can cope with the flows noting that the existing system results in flooding during heavy rain storms
 - The increase in discharge rates from 4.9 litres per second to 176.1 litres per second is significant and more should be done to control the disposal rates.
 More needs to be done to control the overland flows from Back Lane and Highland Court before they result in flooding of Meadowsprings.
 - Building will impact on the habitat of the diverse wildlife found on the site, hedgerows should be retained to protect wildlife habitat
 - Traffic increases through Meadow Spring Way for the safety of existing residents very young and the old
 - No vehicles should be allowed to use Kellbalk Lane track
 - Trees particularly the large Oak should be protected
 - Consideration should be given to the increasing use of Back Lane by pedestrians and the need to improve pedestrian safety
 - Claims of an 90 year lease to provide land for dog walkers due to the rights of way and access that is afforded.
 - Additional development will be likely to give rise to more dog fouling on neighbours gardens on Kellbalk Lane
 - Boundary fencing to Highland Court and Orchard Close should be required to maintain security to those areas
 - Close proximity of new dwellings to Orchard Close would compromise the boundary planting that is a characteristic of the site
 - Side (west facing) windows would result in overlooking and a loss of privacy to dwellings on Orchard Close and should be obscurely glazed
 - Suggest the inclusion of dormer bungalows with a bedroom at ground level

Additionally 6 responses received following re-notification after the 23 July 2015 Planning Committee.

- These responses restated concerns regarding the rate of growth of the town and additionally that the parking available in the town centre is unable to provide for the growing elderly and disabled population who seek (or need) to park close to the town sent. Similarly that the junctions of Crabmill Lane Leasmires Avenue and Tylers Walk are overstretched.
- A small park and ride to cater for York commuters who park in Easingwold should be provided.
- The making of a TPO is supported but more trees should be protected and the hedges retained and open space retained. Loss of open space and new dwellings will intensify the amount of dog fouling around the paths of Kellbalk Lane.

- Research has shown that Kellbalk Lane is a Roman Road and should be protected.
- The new housing will block out light to the dwellings on Kellbalk Lane.
- Objection to the formation of an emergency link on to Oxenby Place as it has not been requested by the emergency services and will cut through the ancient historic path on Kell Balk.

5.0 OBSERVATIONS

- 5.1 The site is allocated for housing development and as noted previously the proposal straddles two areas of allocation EH2 and EH3. The phasing requirements of the LDF at CP7 were relaxed by resolution of Council in December 2013. The proposals for the site are not premature. The concerns of the Easingwold Town Council relating, amongst other things, to the capacity of the social infrastructure, are matters for the infrastructure providers, these are not matters that should delay a development of a site that has been allocated for residential development since 2010 for development in the period 2016 to 2026, noting that no completions of new dwellings will take place before 2016 there would be no breach of the policy. There are a series of requirements of the allocation and these are discussed separately below. Therefore, the main issues in the consideration of this application are:
 - Drainage;
 - The density of development;
 - Housing need, affordable housing and housing type;
 - Access and traffic issues;
 - Provision of open space within the site, suitably linked to existing and proposed footpath routes;
 - Contributions towards off-site infrastructure, including cycle and footpath links and if required drainage and sewerage infrastructure, additional school places and local health care;
 - Trees, hedgerows and ecology;
 - Design; and
 - Landscape impact.

<u>Drainage</u>

- 5.2 As a result of consultation, and in common with other development proposals in the south and east of Easingwold, drainage is a very substantial issue of concern to all involved. The provision of a drainage system that is capable of accommodating foul water and surface water flows from the site whilst also dealing with issues of flooding is critical to the acceptability of this proposal.
- 5.3 Reports have been supplied by the developer to explain the means of providing foul sewage disposal and the measures proposed have not resulted in any objection from Yorkshire Water or the Environment Agency.
- 5.4 The surface water disposal arrangements propose to retain water on site in a piped system during periods of high flow rates for release into the system when flow rates are reduced, an approach indicated in Yorkshire water's consultation response.
- 5.5 Overland flow of water during high intensity rainfall is reported by neighbours to the site. The water is recorded to flow from Back Lane on to Highland Court where the existing systems appear incapable of accommodating the flows such that water then runs between properties and onto the open ground to the north of Meadow Springs Way. Concern has been raised that if no improvements are made to the off-site surface water drainage system this could result in flooding of the new homes to the

north of Meadow Springs Way and by changing the conditions of the land that overland flow could then affect properties on Meadow Springs Way and Hopwith Close.

- 5.6 Whilst dealing with pre-existing drainage problems is not the responsibility of the developer there is a requirement to ensure that the development does not give rise to additional flooding problems or transfer the impact of flood events from the open ground of the application site to the neighbouring dwellings.
- 5.7 The application includes a report that has considered the impact and advice by the agents (Alan Wood and Partners) states as follow:

"The new development will be protected from overland flow by the fact that the finished ground floor level of the properties will be a minimum of 150mm above the surrounding ground.

Should water run across the ground from off-site it will run to the lowest points first, and therefore the flow will follow the roads and footpaths which will be lower than the houses adjacent.

The site will have a surface water drainage system that is designed to a standard which Yorkshire Water will accept; Yorkshire Water will take on the responsibility for its operation, maintenance and management and the system will be designed and constructed to manage rainfall events up to the 1 in 100 year chance of occurrence, with an allowance for a 30% increase in intensity due to the impact of climate change over the next 100 years.

This makes the statistical chance of flooding less than 1%, and it is the highest standard that is expected of any new drainage system from a legal perspective.

Whilst the sewers will not be designed specifically to take overland flow, should it occur it will be intercepted by the new gullies and the system will deal with overland flow like it would deal with rainfall runoff from the site itself. The system can clearly not distinguish between flow caused by direct rainfall or overland flow from off-site: if there is capacity in the new system it will manage the flow.

The joint probability of having the peak 1 in 100 year (plus climate change) event occurring at the same time as overland flow is very low. With this in mind the proposed development layout, floor levels and drainage system affords a practicable and reasonable level of protection for the new residents and off-site third parties."

5.8 A letter and dossier of details of flooding in the town has been prepared by Easingwold Town Council. The information has been supplied to the County and District Council and Yorkshire Water. Concern has been raised that construction works on the application site may change the way that the land drains and that this could cause flooding problems. This has also been addressed by the applicant's agents (Alan Wood and Partners) who respond:

"A drainage system on the site is an improvement compared to the existing situation: now the rainfall can run off the site at an uncontrolled rate and manner. If the site is waterlogged (and therefore rainfall is not absorbed by the land on the site and can runoff, like it does from a roof or road), there is nothing to protect the residents in the surrounding area from water coming towards them. With a developed site that has surface water drainage collecting and effectively impounding the water and having a controlled rate of discharge, the surrounding area is afforded a level of protection that does not currently exist."

- 5.9 Yorkshire Water accepts that the attenuated surface water flows can be directed to its surface water sewer. It is considered that subject to planning conditions to require approval of the details of the drainage systems and to require the completion of those systems, and maintenance arrangements thereafter, the scheme is acceptable in this respect.
- 5.10 The Council's Corporate Facilities Manager has also given consideration to the range of information available on the drainage arrangements and the history of flooding including that prepared by Easingwold Town Council. He writes:-

"I have considered the issues surrounding the determination of planning application 14/02285/FUL, this includes an assessment of information provided by the applicant and associated with application, and I have spoken to Yorkshire Water Services representatives in their technical and planning teams. The present non-determination of the application is related to our understanding of how acceptable sewerage services can be provided to the proposed development.

My summary response is that the Council should grant the application.

The basis of my advice is that there is insufficient evidence available to Yorkshire Water Services, (our consultee in relation to sewerage services), of problems in the catchment of sufficient magnitude that would enable them to reasonably sustain an objection to the application. I would not advise the Council raising an objection on sewerage grounds"

5.11 Following the submission of further details of the drainage problems in Easingwold to Yorkshire Water on 30th October 2015, Yorkshire Water have again stated:

"We maintain our view that the planning application can be considered at the Planning Committee subject to the recommended conditions."

It is concluded that there is no reason to withhold a grant of planning permission on the basis of the drainage issues.

Density

5.12 The density of the scheme is lower than 40 dwellings per hectare, the approximate density identified in policies EH2 and EH3 and results in a lower number of dwellings than identified in those policies. The reduced density allows for an increased amount of open space around the site and is considered to add to the quality of the environment that can be achieved and is therefore an appropriate response to the policy requirements.

Housing need, affordable housing and housing type

5.13 The scheme proposes that 50% of the dwellings are provided as affordable housing units in accordance with Policy CP9. The proposal also accords with the allocation policy requirements to achieve 50%. The previously approved development off Oxenby Place also achieved 50% affordable housing (11 units) the allocations at EH2 and EH3 has achieved the requirement and together with the 58 units to be provided within this proposed will provide a total of 69 affordable homes in Easingwold. A planning obligation is under preparation to secure the occupation of the dwellings for people who have a local need for affordable housing.

- 5.14 The mix of sizes and types of dwellings has been influenced by advice from the Council's Housing and Planning Policy Manager. The mix achieves mainly 2 and 3 bedroom units in two storey dwellings and includes 5% bungalows. The applicant considers that this meets the requirements of the draft Size, Type and tenure of New Homes Supplementary Planning Document because they believe its 10% requirement only applies to market housing (i.e. 5% overall of a scheme with 50% affordable housing). That is not the intention of the draft SPD and no evidence has been provided to suggest that the accommodation needs and preferences of older people with the means to buy their own property differ significantly from the needs and preferences of those without. The details below show the need for housing suitable for older people.
- 5.15 All Hambleton's service centres have a particularly high percentage of older people or are identified places where older people want to live. This is comparatively high in Easingwold:

Area	% Households aged over 65		
Hambleton	27%		
Easingwold	36%		
Rest of Sub Area	22%		

- 5.16 Hambleton's population is ageing and at an increasing rate, with a considerably higher percentage of older people than other areas of the country and region (23% compared with 17% for England) and this is increasing year on year.
- 5.17 The 2011 North Yorkshire Strategic Housing Market Assessment found that 49% of older people currently living in Hambleton intend to move to smaller properties and 46% wanted a bungalow. This is more likely to apply to people in affordable housing in Easingwold because such housing comprises a comparatively large part of the stock within the town:

Tenure of Household Reference persons aged 65+

Area	Owned	Shared Ownership	Social Rented	Private Rented	Rent Free
Hambleton	76%	0%	15%	6%	2%
Easingwold	71%	1%	21%	5%	3%
Easingwold Sub Area	79%	0%	12%	6%	3%

There is a high level of older home owners who would have sufficient equity to buy a smaller property and would be unlikely to qualify for affordable housing because of the £60k ceiling (total income and /or assets) excludes many older people from registering on North Yorkshire's Homechoice.

5.18 The evidence supports the policy objective of providing more bungalows in all eligible housing schemes, regardless of tenure, and demonstrates that this is of particular concern in Easingwold. The "Size, Type and Tenure SPD" was adopted on 15 September 2015, although the application was made before the adoption of the SPD the policy requirements are set within the adopted LDF policy CP8 and DP13 and the NPPF. Whilst the need for affordable bungalows appears to be higher in Easingwold

than other parts of the district it is important that housing need is addressed in a consistent way. The offer of 10% bungalows is therefore very welcome and members are entitled to give this matter significant weight in their decision.

Access route and traffic assessment

- 5.19 The proposed vehicular access routes to the site accords with the Diagram EH2/EH3 in the Allocations DPD showing the main vehicular access from Crabmill Lane and Meadow Springs Way and emergency access from Oxenby Place.
- 5.20 Consideration has been given to the need to upgrade roads and junctions away from the site. The conclusion of NYCC Highways and the applicant's consultant is that there is no need for upgrades beyond the frontage of the site.
- 5.21 An emergency access route is shown linking the development to the north of Meadow Springs Way to Oxenby Place and subject to additional details relating to the crossing of land outside the site the provision of a link can be made a condition of approval. The Highway Authority have also expressed a view that the Crabmill Lane site should also have an emergency link road but acknowledge that if this cannot be achieved a widening of the carriageway may be an appropriate alternative. Amended drawings have been submitted on 14 July 2015 showing a widening of the spine road from Crabmill Lane. The final response from the Highway Authority is awaited.

Open space within the site and links to existing and proposed footpath routes

- 5.22 Policy DP37 requires new housing developments to contribute towards the achievement of the local standards set out in the policy (further explained in the Open space, sport and recreation Supplementary Planning Document of February 2011).
- 5.23 The policy identifies 5 different types of open space and requires assessment of the level of need by each of the 5 types before calculating how much should be provided on site and then establishing the level of facilities to be provided off site and the calculation of the level of contribution and arrangements for on-going maintenance.
- 5.24 The SPD guides that a development of 116 homes should provide on-site facilities for amenity green space, play areas for children, facilities for young people and teenagers and may require allotments. The open space audit for Easingwold shows that there is no shortfall of space for young people and teenagers facilities but improvements to quality may be necessary, there is not shortfall in allotment provision and so the priority to provide allotments is reduced. The SPD guides that a scheme of 116 dwellings is too small to make on-site provision of outdoor sports facilities appropriate.
- 5.25 The development can be anticipated to give rise to a population of about 275 people (116 dwellings x standard occupancy of 2.37 people per dwelling).
- 5.26 The proposal shows that a total of about 5,700 sq. m of open space within the 3.47 hectare site. The open space shown is suitable as amenity green space and could accommodate play areas for children. Some of the 5,700 sq. m is occupied by trees (some subject to Tree Preservation Orders) and hedgerows within the site and make both a contribution towards the maintenance of the landscape of the town and site as well as providing amenity space.
- 5.27 No details are currently shown of equipped children's play space, but this and the requirement for ongoing management can be sought be means of a planning condition.

5.28 The layout of the site shows that suitable links can be provided to the existing network of paths to enable walking as a sustainable means of travel within the town as well as an amenity to the area.

Contributions towards off-site infrastructure

- 5.29 The allocations document refers to contribution to a range of infrastructure and more recently the Council has identified and prioritised infrastructure projects through the adoption of the Community Infrastructure Levy (CIL). The site provides links to the existing footpath network. There are no dedicated cycle track links within the vicinity of the site and no proposals for creation of cycle tracks that require funding have been notified.
- 5.30 The scheme requires additional drainage infrastructure as noted above.
- 5.31 The provision of funding towards additional school places falls within CIL and is on the Regulation 123 List, no further funding can be sought via planning obligation as this would result in doubling the contribution made by the developer.

Trees, hedgerows and ecology

- 5.33 The tree survey submitted with the application identifies the condition of each of the trees within and around the site. A Tree Preservation Order has been made in respect of all of the trees that are identified to be the highest quality.
- 5.34 Additionally there are 17 hedgerows within the site that have been assessed for their importance in the context of the Hedgerow Regulations 1997 and for their value as a hedgerow network with regard to the mix of species present, as wildlife corridors and supporting faunal groups.
- 5.35 The survey work (29 April 2015) and report by Brooks Ecological found that one hedgerow was important under the terms of the Hedgerow Regulations and noted that this hedgerow is retained as part of the scheme. Hedgerow 3 lies on the eastern boundary of the site running for 62 metres northwards from Crabmill Lane, it is important (by virtue of containing at least 5 woody species and at least 3 associated features). 5 other hedgerows (hedges 4, 6, 6A, 12 and 13 (combined length 357 metres)) fall just short of definition as important under the Regulations.
- 5.36 The UK Biodiversity Action Plan identifies hedgerows as a Priority Habitat and one of the aims of the plan is for no net loss of hedgerow and Brooks Ecological advocate that where possible hedgerows should be retained and loss resulting from the development should be compensated through new planting and gapping up of retained hedgerows (particularly hedge 15). Brooks Ecological find that there is potential with suitable mitigation and long term management to create a greater length of high value hedgerow and to maintain wildlife corridor function of the network.
- 5.37 Brooks Ecological recommend that adequate protection should be installed during construction to ensure no detrimental effect on hedgerow 3.
- 5.38 A protected species survey has been undertaken. An inspection of the trees that are proposed to be felled has been undertaken and concludes that none of the trees are used as bat roosts.
- 5.39 There is scope to safeguard the bio-diversity of the site by protecting areas during construction phases and by careful design of a soft and hard landscaping. Both these aspects can be secured by planning condition.

Design

- 5.40 The site has a series of different characters, from the higher density of the housing on Kellbalk Lane set off from the site by the tree lined Kellbalk Lane, the lower density housing areas on Crabmill Lane, Highland Court and Orchard Close and finally the countryside edge to the east of the site.
- 5.41 The layout of the housing on the site has varying levels of density with the use of detached dwellings on the south edge facing Crabmill Lane and facing the central open space with a mix of detached, semi-detached and terraces of three units across the site.
- 5.42 Fifteen house types are proposed, including some bungalows and some in two-storey form with a third floor in the roof space. Most dwellings are 2 storey (78 of the 116 proposed).
- 5.43 The layout of the dwellings provides an appropriate legible street running north-south with secondary cul-de-sac arrangements and developments facing on to the open space areas. The scheme has evolved following input from consultees and advice from officers of the Council to a point where it is considered that the scheme achieves an acceptable quality of design.
- 5.44 The applicants undertook a pre-application community consultation event. Following that event a report was prepared that identified 10 issues to be addressed, one of these related to design and challenged the developer that the scheme proposed did not reflect the historic character of Easingwold. The resulting scheme is one of that uses design details doors and fenestration that do not reflect the history of Easingwold. The elevations of the dwellings are however reasonably varied and would provide an appropriate street scene that is not out of keeping with the varied character of the surrounding that include the modern terraces of Kellbalk Lane and varied character of the homes on Crabmill Lane. Overall the scheme is considered to meet the requirements of DP32.
- 5.45 Concern has been raised that the development may cause a loss of light to neighbouring dwellings on Kellbalk Lane. The distance between the proposed and existing dwellings is a minimum of 20.5m and the nearest window to window distance is 22m. The orientation of the dwellings is not 'square-on' but set at angles thereby reducing the potential impact. The separation distance between dwellings on Kellbalk Lane and the proposed dwellings is commonly in excess of 30 metres. The separation distances to all other boundaries achieve the 21m separation, a minimum standard that this and many Council has used for many years.

Landscape impact

5.46 The allocation sites are in a position that is not widely visible from public roads. The development proposals are made to enable the retention of the trees and hedgerows on the boundaries of the site. This would substantially reduce the visual impact of the development from views outside of the town and would enable the scheme without undue harm to the landscape.

Conclusion

5.47 The site is allocated for residential development and the proposed scheme has shown to achieve the requirements of the allocation document. Whilst there are concerns relating to the matters of drainage these can be addressed by condition and the issues relating to the capacity of healthcare and other social infrastructure in the

town do not justify withholding planning permission for an otherwise acceptable development. The scheme will provide both 10% bungalows and 50% affordable housing. Taking in to account all the matters raised by consultees and those making representations the planning balance falls in favour of approval.

6.0 RECOMMENDATION

- 6.1 That subject to any outstanding consultations and the completion of a Planning Obligation to secure an appropriate affordable housing content, the application is **GRANTED** subject to the following conditions:
- 1. The development hereby permitted shall be begun within three years of the date of this permission.
- 2. No above ground construction work shall be undertaken until details and samples of the materials to be used in the construction of the external surfaces of the development have been made available on the application site for inspection (and the Local Planning Authority have been advised that the materials are on site) and the materials have been approved in writing by the Local Planning Authority. The development shall be constructed of the approved materials in accordance with the approved method.
- 3. No development shall commence on site other than investigative works or the depositing of material on the site until a detailed scheme(s) of highway construction and related works have been submitted to and approved in writing by the Local Planning Authority. Thereafter the development shall be undertaken and maintained in full compliance with the approved drawings and details, including the programme of work, unless otherwise approved in writing by the Local Planning Authority. (The minimum requirements of the scheme(s) likely to be necessary to meet this condition are set out in Informative 1 to this decision.)
- 4. No dwelling to which this planning permission relates shall be occupied until the carriageway and any footway/footpath from which it gains access is constructed to base course macadam level and/or block paved and kerbed and connected to the existing highway network with street lighting installed and in operation. The completion of all road works, including any phasing, shall be in accordance with a programme approved in writing with the Local Planning Authority before the first dwelling of the development is occupied.
- 5. Notwithstanding the provisions of the Town and Country Planning General Permitted Development Order 1995 or any subsequent Order, the garages on plots 1, 17, 43, 44, 45, 52, 53, 80, 88, 94 and 95 shall not be converted into habitable accommodation and shall be kept available for the parking of vehicles.
- 6. All doors and windows on elevations of the building(s) adjacent to the existing and/or proposed highway shall be constructed and installed such that from the level of the adjacent highway for a height of 2.4 metres they do not open over the public highway and above 2.4 metres no part of an open door or window shall come within 0.5 metres of the carriageway. Any future replacement doors and windows shall also comply with this requirement.
- 7. The development shall not be commenced until a plan has been submitted to and approved by the Local Planning Authority to show all existing trees and hedges which are to be felled or retained together with the positions, height and design of protective fences, the areas for the storage of materials and the stationing of machines and huts, and the direction and width of temporary site roads and accesses. The development shall not be commenced until tree and hedge guards, at least 1.5

metres high, have been erected on the perimeter of the branch spread of trees and hedgerows (or, in the case of a fastigiated tree such as a Lombardy Poplar, have been erected to enclose an area with a radius of 6 metres from the trunk) of all the trees shown as being retained. The guards shall be maintained in position and in good order during the whole period of works on site. Works, including the removal or deposit of earth or other materials shall not be carried out within the tree guards without the prior consent of the Local Planning Authority.

- 8. No dwelling shall be occupied until a detailed landscaping scheme indicating the type, height, species and location of all new trees and shrubs, has been submitted to and approved by the Local Planning Authority. No dwelling shall be occupied after the end of the first planting and seeding seasons following the approval of the landscaping scheme, unless those elements of the approved scheme situate within the curtilage of that dwelling or between that dwelling and the boundary of the site or areas of open space have been implemented. Any trees or plants which within a period of 5 years of planting die, are removed, or become seriously damaged or diseased, shall be replaced with others of similar size and species.
- 9. No dwelling shall be occupied until the boundary walls, fences and other means of enclosure to that dwelling have been constructed in accordance with the details approved in accordance with drawing number 686/001 Rev D, or any amended version or additional detail that has been approved by the Local Planning Authority. All boundary walls, fences and other means of enclosure shall be retained and no part thereof shall be removed without the prior consent of the Local Planning Authority.
- 10. No development above ground shall commence until details that show how 'Secured by Design' principles have been incorporated into the scheme have been submitted for the written approval of the Local Planning Authority and once approved the development shall be implemented in accordance with the approved 'Secured by Design' details prior to occupation or use of any part of the development hereby approved.
- 11. Unless otherwise approved in writing by the Local Planning Authority, there shall be no piped discharge of surface water from the development prior to the completion of the approved surface water drainage works. Surface water discharge to public sewer shall not exceed 5 litres a second and the submitted details shall include the means by which this attenuation rate will be achieved. Furthermore, no buildings shall be occupied or brought into use prior to completion of the approved surface drainage works.
- 12. The development shall not begin until arrangements (including a timetable for implementation and management plan) for the provision of on-site amenity space, children's play and young people's facilities have been submitted to and approved in writing by the Local Planning Authority. The on-site amenity space, children's play and young people's facilities shall be provided in accordance with the approved arrangements and the future management transferred to a management company to be managed in perpetuity.

The reasons for the above conditions are:

1. To ensure compliance with Sections 91 and 92 of the Town and Country Planning Act 1990 and where appropriate as amended by Section 51 of the Planning and Compulsory Purchase Act 2004.

- 2. To ensure that the external appearance of the development is compatible with the immediate surroundings of the site and the area as a whole in accordance with Hambleton Local Development Framework Policy CP17.
- 3. In accordance with Policy CP1, CP2, DP3 and DP4 and to secure an appropriate highway constructed to an adoptable standard in the interests of highway safety and the amenity and convenience of highway users.
- 4. In accordance with Policy CP1, CP2, DP3 and DP4 and to ensure safe and appropriate access and egress to the dwellings, in the interests of highway safety and the convenience of prospective residents.
- 5. In accordance with Policy CP1 and DP1 and to ensure the retention of adequate and satisfactory provision of off-street accommodation for vehicles generated by occupiers of the dwelling and visitors to it, in the interest of safety and the general amenity the development.
- 6. In accordance with Policy CP1 and to protect pedestrians and other highway users.
- 7. To ensure that existing trees within the site, which are of amenity value, are adequately protected during the period of construction in accordance with Local Development Framework Policies CP16, DP31 and DP32.
- 8. To protect the amenity of the neighbouring residents and to ensure that the development is appropriate to the character and appearance of its surroundings.
- 9. To protect the amenity of the residents and their neighbouring residents and to ensure that the development is appropriate to the character and appearance of its surroundings and to provide security to the new dwellings.
- 10. Prior to the development commencing details that show how 'Secured by Design' principles have been incorporated into the scheme shall be submitted for the written approval of the Local Planning Authority and once approved the development shall be implemented in accordance with the approved 'Secured by Design' details prior to occupation or use of any part of the development hereby approved.
- 11. To ensure that no foul or surface water discharges take place until proper provision has been made for their disposal in accordance with the LDF policies CP1, CP21 and DP6.
- 12. To ensure that the on-site amenity space is provided and maintained in accordance with the provisions of the LDF Policies CP19 and DP37.

Attention is drawn to the following Informatives:

- 1. The minimum works required to be submitted in pursuit of condition 3 are as follows:
 - a. Detailed engineering drawings to a scale of not less than 1:500 and based upon an accurate survey showing:
 - the proposed highway layout including the highway boundary;
 - dimensions of any carriageway, cycleway, footway and verges;
 - visibility splays including measures to give clear visibility of 45m measured along both channel lines of the major road Crabmill Lane from a point measured 2.4m down the centre line of the access road, and pedestrian visibility splays giving visibility of 2 metres x 2 metres measured down each side of the access andthe back edge of the footway of the major road have been provided. (In both cases the eye height will be 1.05 and the object

- height shall be 0.5m. Once created, these visibility areas shall be maintained clear of any obstruction and retained for their intended purpose at all times.);
- the proposed buildings and site layout, including levels; accesses and driveways including initial site access; drainage and sewerage system including measures to prevent surface water discharging on to the highway; lining and signing; traffic calming measures; and all types of surfacing (including tactiles), kerbing and edging.
- b. Longitudinal sections to a scale of not less than 1:500 horizontal and not less than 1:50 vertical along the centre line of each proposed road showing the existing ground level; the proposed road channel and centre line levels; and full details of surface water drainage proposals.
- c. Full highway construction details including: typical highway cross-sections to scale of not less than 1:50 showing a specification for all the types of construction proposed for carriageways, cycle ways and footways/footpaths; when requested cross sections at regular intervals along the proposed roads showing the existing and proposed ground levels; kerb and edging construction details; and typical drainage construction details.
- d. Details to prevent the deposit of mud and other debris on the public highway during construction and details of the timing and routing of construction traffic that exceeds 7.5 tonnes in weight and site compound, staff and visitor parking layout and all other matters relating to construction site management.
- e. Details of all proposed street lighting.
- f. Drawings for the proposed new roads and footways/footpaths giving all relevant dimensions for their setting out including reference dimensions to existing features.
- g. Full working drawings for any structures which affect or form part of the highway network.
- h. A programme for completing the works including any repairs to the highways over which access is taken.



Parish: East HarlseyCommittee Date :12 November 2015Ward: Osmotherley & SwainbyOfficer dealing :Mrs B Robinson

Target Date: 8 October 2015

15/01838/OUT

Application for Outline Planning Permission for construction of a single detached dwelling with all matters reserved.

at Land Adjacent Cat And Bagpipes Inn East Harlsey Northallerton North Yorkshire for Mr S Barr

- 1.0 SITE DESCRIPTION AND PROPOSAL
- 1.1 The site is located opposite the junction of the north-south through-road within the village street. The site dimensions are $15m \times 36m$ (max dimensions). The site is at present rough ground with grass and weeds, and a rough hardstanding at the front. There is an existing access from the highway which serves the adjacent pub. The site includes a brick built single garage, with a steeply pitched roof. At the front of the site the land slopes down towards the road. Overlapping the front of the site there is a hedged enclosure around the village war memorial.
- 1.2 To the south of the site there is a two storey public house constructed of brick with a distinctive plain tiled, hipped roof with dormers. The pub is set back from the road with a car park to the front which is partially cobbled. To the rear facade of the pub is a first floor conservatory sited on top of a single storey flat roof addition. On the north side of the pub there is an attached double garage with flat roof, and a pitched roof single garage, similar in style to the pub. The single garage is within the application site.
- 1.3 To the north of the site, Baildon is a large detached house constructed of brown brick. The house is off-set from the boundary by approximately 13m, and has windows facing the site.
- 1.4 A public right of way enters the site from the north east corner, and then runs along the outside of the north boundary, between the site and Baildon, where it is partially enclosed by fencing.
- 1.5 Opposite the site the village road is lined houses of a variety of styles.
- 1.6 The application seeks planning permission in outline for the construction of a single dwelling, with all matters reserved.
- 2.0 RELEVANT PLANNING HISTORY
- 2.1 2/91/043/0073 Outline application for the construction of a dwellinghouse. Granted
- 2.2 2/96/043/0073B Renewal of consent for a dwelling. Granted
- 3.0 NATIONAL AND LOCAL POLICY:
- 3.1 The relevant policy of the Development Plan and any supplementary planning policy advice are as follows;

Core Strategy Policy CP1 - Sustainable development
Development Policies DP1 - Protecting amenity
Core Strategy Policy CP4 - Settlement hierarchy
Interim Guidance Note - adopted by Council on 7th April 2015
Core Strategy Policy CP17 - Promoting high quality design
Development Policies DP32 - General design
National Planning Policy Framework

4.0 CONSULTATIONS

4.1 Parish Council - Refuse.

Plot full of rubbish - difficult to develop safely.

Too small to build a detached dwelling

Insufficient information to decide whether suitable for development

Footpath and an electricity supply pass through and over the plot - need to consider how these would be affected.

4.2 Neighbours and site notice - observations received are summarised below.

i. object

No need for dwelling. Village lacks facilities including school shop or post office or any public transport. Site has been used as a dumping ground - needs environmental assessment (asbestos widely used during the period of the railway wagon formerly on the site). Previous approval was in association with the occupation of the pub and should not be a precedent. Highway safety concerns at junction. Loss of privacy (to neighbouring house).

Footpath crosses site - needs to be maintained.

Existing site provides level path to the war memorial.

Danger from electricity supply crossing site.

Comment on positon of Site Notice.

ii. Object

Previous approval only given due to connection of owner with the pub.

Conflict of interest with nature of pub use - potential for noise.

Land used as over flow car park. Access to war memorial and footpath.

Overcrowded appearance. Effect on the stand-alone pub

- 4.3 NYCC Highways note that visibility is restricted but note existing access and comment that a refusal would be difficult to sustain. Conditions requested.
- 4.4 Yorkshire Water -
- 4.5 EHO No objection
- 4.6 Ramblers No objection. Query regarding the line of the adjacent public footpath as it leaves the road and significance of a manhole cover in the car park will be followed up by the Parish Council.

5.0 OBSERVATIONS

5.1 East Harlsey is a village without status within the hierarchy set out in CP4 as adopted 2008. In 2015 the Council has adopted an Interim Guidance Note which updates the hierarchy and includes East Harlsey as a secondary village. The Guidance provides for a more flexible consideration of new development at the edge of settlements. Accordingly the main consideration is whether the development fits the criteria set out in the Interim Guidance, and thereafter whether the proposed development is in accordance with other relevant policies of the local plan including the amenity of nearby occupiers (CP1 and DP1, design (CP17 and DP32) and any highway safety issues.

5.2 Interim Guidance Criteria:

Small scale housing development will be supported in villages where it contributes towards achieving sustainable development by maintaining or enhancing the vitality of the local community AND where it meets ALL of the following criteria:

- 1. Development should be located where it will support local services including services in a village nearby.
- 2. Development must be small in scale, reflecting the existing built form and character of the village.
- 3. Development must not have a detrimental impact on the natural, built and historic

environment.

- 4. Development should have no detrimental impact on the open character and appearance of the surrounding countryside or lead to the coalescence of settlements.
- 5. Development must be capable of being accommodated within the capacity of existing or planned infrastructure.
- 6. Development must conform with all other relevant LDF policies.
- 5.3 The inclusion of East Harlsey in the updated settlement hierarchy as a secondary village indicates it is considered to be a reasonably 'sustainable' location where an additional dwelling can be considered to support local services, which in this case include a pub, and a village hall.
- 5.4 As a single dwelling it is small in scale. The site is between existing developments and the development reflects the existing built form of the village.
- 5.5 The site is a largely unused space and other than hedges, which could be retained as required, would not be detrimental to the natural environment. The surroundings do not include Listed Buildings and overall the development of the site would not be harmful to the historic or built environment, subject to considerations of design, which if the application is approved can be controlled with the consideration of a reserved matters application.
- 5.6 The site is within the built up part of the village and will not impact on the rural surroundings.
- 5.7 There is no evidence that a single extra dwelling cannot be accommodated within the capacity of the existing infrastructure.
- 5.8 Overall therefore, the proposal is acceptable development in the terms of the Interim Policy Guidance.
- 5.9 With regard to the amenity of neighbours, Baildon is offset, and benefits from a established hedge. It is considered that through a suitably designed and laid out property, a house can be developed on this site without detriment to the amenity of neighbouring occupiers.
- 5.10 Occupiers of the proposed house will be in close proximity to the pub and potentially vulnerable to noise and activity during opening hours. Many or most village pubs have a similar relationship with neighbouring properties. The Environmental Health Officer has raised no objections to the proposed development and as such the proposed development is not considered to be prejudicial to the existing pub use nor is the existence of the pub use considered to be prejudicial to the amenity of the proposed dwelling.
- 5.11 Siting and design are reserved to be considered later. However, it can be seen that the size and form of the proposed plot is capable of accommodating a dwelling. The site is relatively small in relation to the pub, and the neighbouring house, and the issue to consider is whether a dwelling could be reasonably accommodated on the site without appearing to be squeezed in, in comparison with neighbouring properties.
- 5.12 There is relatively generous separation between the site and Baildon, and a new house would not therefore appear crowded in a northerly direction. A house on the plot would be relatively close to the pub, but will benefit from the 'air space' over the pub garage and it is considered that there is scope to consider that an acceptable design could be arrived at which is sensitive to the design of the pub building and which would sit comfortably alongside it, in the street scene.
- 5.13 With regard to highway safety, there is an existing access from the road. Attention has been drawn by the highway authority to insufficient visibility. The existing access is not

distinct or separate from that of the pub and by the level of wear may have been used as part of the pub access. Taking into account that the access is not objected to by the highway authority, the use of the access in connection with one dwelling is considered on balance to be acceptable, particularly as there is scope to ensure that the details approved in connection with the reserved matters will retain maximum visibility for the access and with car users on the adjacent pub site. On this basis the proposal is capable of being considered acceptable in highway safety terms, subject to appropriate conditions.

- 5.13 With regard to the concerns of the Parish Council and neighbours in terms of the sustainability of the site, it is noted that the village has been allocated as a 'secondary village' within the revised settlement hierarchy. It will be the responsibility of the developer to dispose of any waste on the ground appropriately, and also take any necessary measures to redirect the electricity supply with the relevant authorities. Subject to details to be contained within a reserved matters application the proposed development is not considered to have any significant adverse impact on highway safety. Again subject to a suitable design the development of this site is not considered to have any significant impact on neighbouring amenity.
- 5.14 Access to the public footpath must be retained by law and due to the position of the footpath, this is capable of being achieved through a suitable layout. It should be noted that the previous consent was not tied to the public house by condition as suggested in representations.
- 5.14 The development is CIL liable.

SUMMARY

- 5.15 Due to its size and location the site is capable of accommodating a dwelling without unacceptable harm to the surroundings or the amenities of neighbours or road safety and is able to comply with the above policies.
- 5.16 The Local Planning Authority has acted positively and proactively in determining this application by assessing the proposal against all material considerations, including planning policies and any representations that may have been received and subsequently determining to grant planning permission in accordance with the presumption in favour of sustainable development, as set out within the National Planning Policy Framework.

6.0 **RECOMMENDATION**:

- **6.1** That subject to any outstanding consultations the application be **GRANTED** subject to the following condition(s)
 - 1. Application for the approval of all of the reserved matters shall be made to the Local Planning Authority not later than three years from the date of this decision and the development hereby approved shall be begun on or before whichever is the later of the following dates: i) Five years from the date of this permission ii) The expiration of two years from the final approval of the reserved matters or in the case of approval on different dates, the final approval of the last such matter to be approved.
 - 2. The development shall not be commenced until details of the following reserved matters have been submitted to and approved by the Local Planning Authority: (a) the siting, design and external appearance of each building, including a schedule of external materials to be used; (b) the means of access to the site; (c) the landscaping of the site.
 - 3. Prior to development commencing, details and samples of the materials to be used in the construction of the external surfaces of the development shall be made available on the application site for inspection and the Local

Planning Authority shall be advised that the materials are on site and the materials shall be approved in writing by the Local Planning Authority. The development shall be constructed of the approved materials in accordance with the approved method.

- 4. The development hereby approved shall not be commenced until details of the foul sewage and surface water disposal facilities have been submitted and approved in writing by the Local Planning Authority.
- 5. The use of the development hereby approved shall not be commenced until the foul sewage and surface water disposal facilities have been constructed and brought into use in accordance with the details approved under condition above.
- 6. The development shall not be commenced until details relating to boundary walls, fences and other means of enclosure for all parts of the development have been submitted to and approved in writing by the Local Planning Authority
- 7. No dwelling shall be occupied until the boundary walls, fences and other means of enclosure have been constructed in accordance with the details approved in accordance with condition above. All boundary walls, fences and other means of enclosure shall be retained and no part thereof shall be removed without the prior consent of the Local Planning Authority.
- 8. There shall be no access or egress by any vehicles between the highway and the application site until full details of any measures required to prevent surface water from non-highway areas discharging on to the existing or proposed highway together with a programme for their implementation have been submitted to and approved in writing by the Local Planning Authority. The works shall be implemented in accordance with the approved details and programme.
- 9. Unless otherwise approved in writing by the Local Planning Authority, there shall be no excavation or other groundworks, except for investigative works, or the depositing of material on the site until the access(es) to the site have been set out and constructed in accordance with the published Specification of the Highway Authority and the following requirements:
- (i) The details of the access shall have been approved in writing by the Local Planning Authority. (ii) The final surfacing of any private access shall not contain any loose material that is capable of being drawn on to the existing public highway.
- 10. Unless otherwise approved in writing by the Local Planning Authority, there shall be no excavation or other groundworks, except for investigative works, or the depositing of material on the site in connection with the construction of the access road or building(s) or other works hereby permitted until full details of the following have been submitted to and approved in writing by the Local Planning Authority:

 (i) vehicular turning and parking arrangements
- 11. No part of the development shall be brought into use until the approved vehicle access, parking, manoeuvring and turning areas approved under condition 10 are available for use unless otherwise approved in writing by the Local Planning Authority. Once created these areas shall be maintained clear of any obstruction and retained for their intended purpose at all times.

Community Infrastructure Levy

Please note that the proposed development is liable under the Community Infrastructure Levy Charging Schedule, adopted by Hambleton District Council on the 07 April 2015. Details of the charging schedule are available on the Council website. www.hambleton.gov.uk

The reasons for the above conditions are:-

- 1. To ensure compliance with Section 92 of the Town and Country Planning Act, 1990
- 2. To enable the Local Planning Authority to properly assess these aspects of the proposal, which are considered to be of particular importance, before the development is commenced.
- 3. To ensure that the external appearance of the development is compatible with the immediate surroundings of the site and the area as a whole in accordance with Hambleton Local Development Framework Policy CP17.
- 4. To ensure that the development is appropriate in terms of amenity in accordance with Local Development Framework Policies CP1 and DP1.
- 5. In order to avoid the pollution of watercourses and land in accordance with Local Development Framework CP21 and DP43
- 6. To protect the amenity of the neighbouring residents and to ensure that the development is appropriate to the character and appearance of its surroundings.
- 7. To protect the amenity of the neighbouring residents and to ensure that the development is appropriate to the character and appearance of its surroundings.
- 8. In accordance with CP2 and DP4 and in the interests of highway safety
- 9. In accordance with CP2 and DP4 and to ensure a satisfactory means of access to the site from the public highway in the interests of vehicle and pedestrian safety and convenience.
- 10. In accordance with CP2 and DP4 and to ensure appropriate on-site facilities in the interests of highway safety and the general amenity of the development.
- 11. In accordance with CP2 and DP4 and to ensure appropriate on-site facilities in the interests of highway safety and the general amenity of the development.

Parish: Newby Wiske Ward: Morton on Swale

5

Committee Date: 12 November 2015
Officer dealing: Mrs H M Laws
Target Date: 16 November 2015

15/01542/OUT

Outline planning application for construction of one, two storey dwellinghouse as amended by email received by Hambleton District Council on 23 September 2015 and plan received on 12 October 2015.

at Land To The South Of Willow Garth Newby Wiske North Yorkshire for Mr D Marwood.

- 1.0 SITE DESCRIPTION AND PROPOSAL
- 1.1 The site lies at the northern end of the village on the western side of the road and is within the Newby Wiske Conservation Area. The area covers approximately 0.1 hectares and has a frontage of approximately 36m along the village street bounded by a mature, well established hedgerow.
- 1.2 The site lies immediately adjacent to a dwelling known as Merryn. Access into the application site is a secondary access for Merryn and also provides a farm access to fields to the west. A small group of existing buildings lies on the site, which is used as a joiner's store. These buildings would be removed as a result of the proposed development.
- 1.3 The application has been amended since first submitted, to reduce the size of the site and amend the number of dwellings proposed from two to one. The ground level of the application site is at a similar level to the ground level of Merryn; the land to the north, which has been removed from the application site boundary, rises significantly. The application site as amended forms part of the same paddock as the higher land and is not divided by any boundary. Mature hedgerows bound the paddock to all sides.
- 1.4 It is proposed to construct a detached two storey dwelling. All matters are reserved but illustrative details of a dwelling have been received. The scheme shows a cottage style double fronted property finished in brickwork and clay pantiles with timber framed horizontal sliding sash windows.
- 2.0 PLANNING & ENFORCEMENT HISTORY
- 2.1 87/0665/OUT Outline application for the construction of a bungalow. Permission refused 29/9/1987 for the following reasons:
- 1. The proposed development would be contrary to Policy H5 of the Vale of Mowbray Local Plan. That Policy defines the village limits outside which development should be resisted in the interests of safeguarding the amenities of the area, and the proposed site lies outside those limits.
- 2. The proposal cannot be considered to be infilling and would constitute an undesirable extension of the village objectionable on amenity grounds. (Infill development is defined as the development of a minor gap in the otherwise built up frontage.)
- 3.0 RELEVANT PLANNING POLICIES:
- 3.1 The relevant policy of the Development Plan and any supplementary planning policy advice are as follows:

Core Strategy Policy CP1 - Sustainable development

Core Strategy Policy CP2 - Access

Core Strategy Policy CP4 - Settlement hierarchy

Core Strategy Policy CP16 - Protecting and enhancing natural and man-made assets

Core Strategy Policy CP17 - Promoting high quality design

Core Strategy Policy CP21 - Safe response to natural and other forces

Development Policies DP1 - Protecting amenity

Development Policies DP4 - Access for all

Development Policies DP28 - Conservation

Development Policies DP30 - Protecting the character and appearance of the countryside

Development Policies DP32 - General design

Interim Guidance Note - adopted by Council on 7th April 2015

National Planning Policy Framework - published 27 March 2012

4.0 CONSULTATIONS

- 4.1 Parish Council no comments received (expiry date for representations 24/10/2015)
- 4.2 NYCC Highways conditions are recommended including one requiring the provision of a visibility splay. In order for this to be achieved the length of the hedge along the splay will need to be removed. Any proposed fence line or hedge line will need to be set back to keep the visibility splay.
- 4.3 Yorkshire Water conditions recommended requiring the provision of a separate system for surface water discharge.
- 4.4 HDC Environmental Health no objections
- 4.5 Site notice/local residents an objection has been received from a local resident prior to the submission of amendments, which is summarised as follows:
- 1. My primary concern is the access to the site. The site includes a right of way, this right of way has existed for a very long time and is used as a primary access to the field situated to the west of the site and as a secondary access to Merryn. The access lane shared by the applicant, Mr Richardson who owns the field to the west of the site and the occupants of Merryn, is already dangerous and has to be used with caution. The increased usage of the access proposed by constructing one or two dwellings would only intensify this danger.
- 2. The Local Highways Authority states that the visibility splays must give visibility for 43 metres each way from a point two metres down the centre line of the access road. The current access point will not allow for this even with the applicants suggested new access radius on the Indicative layout plan. Moving the access north to increase visibility to the south would not only hinder access for Mr Richardson's farm vehicles but would be moving the access closer to the blind brow of the hill, reducing visibility for those using the access and oncoming traffic. At peak times there is a large flow of traffic from both directions though the village to numerous businesses and the local school.
- 3. Another concern is that the proposed new dwellings would be extending the village, the houses would be the last in the village when heading north. There is no question that this will dramatically change the look of the village and its heritage. If, as proposed the dwellings were to emulate the look of the bungalows further south in the village, this more modern look would detract from the traditional style of the listed cottages that are currently the first thing you see on entrance to the village from the north.

Following the receipt of amended plans the following comments have been submitted:

Whilst we appreciate the number of dwellings has been reduced and the design of the cottage is in keeping with the traditional dwellings around it, our primary concern still remains the safety of the access.

Currently the access is only used on occasion, there are not vehicles going in and out every day. The increased usage of the access by occupants in the proposed dwelling will add to an already dangerous area of road. At peak times there is a large flow of traffic that rarely travels at the speed limit of 30mph.

Due to the above comments and the access issues raised in our previous letter, our objection still stands.

5.0 OBSERVATIONS

- 5.1 The main issues for consideration in this case relate to the principle of a new dwelling in this location outside Development Limits, an assessment of the likely impact of the proposed dwelling on the character and appearance of the Conservation Area, the rural landscape, neighbour amenity, highway safety and developer contributions.
- 5.2 The site falls outside of Development Limits as Newby Wiske does not feature within the settlement hierarchy defined within Policy CP4 of the Core Strategy. Policy DP9 states that development will only be granted for development "in exceptional circumstances". The applicant does not claim any of the exceptional circumstances identified in Policy CP4 and, as such, the proposal would be a departure from the development plan. However, it is also necessary to consider more recent national policy in the form of the NPPF. Paragraph 55 of the NPPF states:

"To promote sustainable development in rural areas, housing should be located where it will enhance or maintain the vitality of rural communities. For example, where there are groups of smaller settlements, development in one village may support services in a village nearby. Local planning authorities should avoid new isolated homes in the countryside unless there are special circumstances".

- 5.3 To ensure appropriate consistent interpretation of the NPPF alongside Policies CP4 and DP9, the Council has adopted Interim Policy Guidance (IPG) relating to Settlement Hierarchy and Housing Development in the Rural Areas. This guidance is intended to bridge the gap between CP4/DP9 and the NPPF and relates to residential development within villages. The IPG has brought in some changes and details how Hambleton District Council will now consider development in and around smaller settlements and has included an updated Settlement Hierarchy.
- 5.4 In the settlement hierarchy contained within the IPG, Newby Wiske is defined as an "other settlement"; within the IPG small scale development adjacent to the main built form of the settlement "will be supported where it results in incremental and organic growth". To satisfy criterion 1 of the IPG the proposed development must provide support to local services including services in a village nearby. The site lies on the edge of the village of Newby Wiske which is identified in the IPG as an example of a cluster village together with South Otterington. The two villages have long been linked economically and socially which continues to the present day and collectively have a church, a primary school and a pub. Each village is readily accessible to each other on foot or bicycle as well as by car on the local road network. Newby Wiske is less than 0.5km distance from South Otterington and the application site is a further 0.5km through the village. Criterion 1 would be satisfied.
- 5.5 It is important to consider the likely impact of the proposed development with particular regard to criteria 2, 3 and 4 of the IPG. The proposed dwellings would be on undeveloped land that lies at the end of the row of residential properties on the western side of the main village street. There is a house and farm beyond the edge of the village on the opposite side of the street. The following detailed advice within the IPG is considered to be relevant:
- "Proposals will be assessed for their impact on the form and character of a settlement. Consideration should be given to the built form of a settlement, its historical evolution and its logical future growth and how the proposal relates to this."
- "Any detrimental impact on the character, appearance and environmental quality of the surrounding area should be avoided and development should not compromise the open and rural character of the countryside."
- 5.6 The proposed development would extend the built part of Newby Wiske further along the western side of the village street and is therefore a form of ribbon development. The application site, although undeveloped, does not form part of the countryside but has more in common with the village. The boundary between the application site and the rural landscape to the west is established by a timber fence and hedgerow and the land is not in agricultural use.

- 5.7 The existing buildings are in poor condition and do not contribute positively to the character and appearance of the Conservation Area. As such it is considered that their removal and replacement with a dwelling of a high standard of design, would improve the appearance of the site. The removal of the buildings must be balanced against the loss of a length of the existing hedgerow of approximately 23m that would make the site more visible. The proposed development would significantly alter the appearance of the site but it must be considered whether this would detract from the character and appearance of the Conservation Area. A condition can be imposed requiring the replacement of a hedgerow behind the visibility splay, which once established would have the same effect as the existing hedgerow.
- 5.8 The submitted details, although illustrative only, demonstrates that a traditional form of dwelling could be provided on the site. It is considered that the development proposed, without the loss of rural landscape, would appropriately respect the general built form of the village. There would be no harmful impact to the natural, built and historic environment.
- 5.9 The closest neighbour is Merryn immediately to the south. A new dwelling, either single or two storey, could be designed to prevent overshadowing or overlooking.
- 5.10 The Highway Authority has no objections regarding the proposed development. It is not considered that the proposed development would adversely impact highway safety and conditions are recommended including the requirement to provide a visibility splay. The existing access serves the joiner's workshop, the field to the rear and Merryn and the visibility is substandard. The proposed development provides an opportunity to improve the visibility for all users.
- 5.11 Under the Community Infrastructure Levy (CIL) regulations the proposed dwellings are liable for payment of CIL at a rate of £55 per sq. m, the rate adopted by the Council on 7 April 2015.
- 5.12 It is considered that the proposal is in accordance with the Interim Policy Guidance document in that the housing development, subject to the subsequent approval of detailed plans in respect of scale, design and materials and would have no adverse impact on landscape character, residential amenity and highway safety.
- 5.13 The proposed development is acceptable and approval of the application is recommended.

6.0 RECOMMENDATION: that subject to any outstanding consultations the application be GRANTED subject to the following conditions:

- 1. Application for the approval of all of the reserved matters shall be made to the Local Planning Authority not later than three years from the date of this decision and the development hereby approved shall be begun on or before whichever is the later of the following dates: i) Five years from the date of this permission ii) The expiration of two years from the final approval of the reserved matters or in the case of approval on different dates, the final approval of the last such matter to be approved.
- 2. The development shall not be commenced until details of the following reserved matters have been submitted to and approved by the Local Planning Authority: (a) the layout, scale and appearance of each building, including a schedule of external materials to be used; (b) the means of access to the site; (c) the landscaping of the site.

- 3. Prior to development commencing, details and samples of the materials to be used in the construction of the external surfaces of the development shall be made available on the application site for inspection and the Local Planning Authority shall be advised that the materials are on site and the materials shall be approved in writing by the Local Planning Authority. The development shall be constructed of the approved materials in accordance with the approved method.
- 4. All new, repaired or replaced areas of hard surfacing shall be formed using porous materials or provision shall be made to direct run-off water from the hard surface to an area that allows the water to drain away naturally within the curtilage of the property.
- 5. The development shall not be commenced until a detailed landscaping scheme indicating the type, height, species and location of all new trees and shrubs, has been submitted to and approved by the Local Planning Authority. No dwelling shall be occupied after the end of the first planting and seeding seasons following the approval of the landscaping scheme, unless those elements of the approved scheme situate within the curtilage of that dwelling have been implemented. Any trees or plants which within a period of 5 years of planting die, are removed, or become seriously damaged or diseased, shall be replaced with others of similar size and species.
- 6. The dwelling shall not be occupied after the end of the first planting season following the commencement of the development unless a hedge has been planted along the highway frontage immediately behind the visibility splays required by another condition of this permission. This shall include the re-alignment of the hedge over the field adjoining the application site, behind the visibility splay. The hedge shall be thorn or beech, or such other species as may be agreed in writing by the Local Planning Authority before planting. Any hedging removed, dying, becoming seriously damaged or becoming seriously diseased within 5 years of planting shall be replaced by hedging of similar size and species to that originally planted.
- 7. Prior to development commencing detailed cross sections shall be submitted to and approved in writing by the Local Planning Authority, showing the existing ground levels in relation to the proposed ground and finished floor levels for the development and the relationship of the proposed development to the existing dwelling to the south. The levels shall relate to a fixed Ordnance Datum. The development shall be constructed in accordance with the approved details and thereafter be retained in the approved form.
- 8. No piped discharge of surface water from the application site shall take place until works to provide a satisfactory outfall for surface water other than the public sewer have been completed in accordance with details to be submitted to and approved by the Local Planning Authority before development commences.
- 9. There shall be no access or egress by any vehicles between the highway and the application site until full details of any measures required to prevent surface water from non-highway areas discharging on to the existing or proposed highway together with a programme for their implementation have been submitted to and approved in writing by the Local Planning Authority. The works shall be implemented in accordance with the approved details and programme.

- Unless otherwise approved in writing by the Local Planning Authority, there shall be no excavation or other groundworks, except for investigative works, or the depositing of material on the site until the access to the site have been set out and constructed in accordance with the published Specification of the Highway Authority and the following requirements; The details of the access shall have been approved in writing by the Local Planning Authority in consultation with the Highway Authority: d. crossing of the highway verge shall be constructed in accordance with the Standard Detail number E6; e. Any gates or barriers shall be erected a minimum distance of 6 metres back from the carriageway of the existing highway and shall not be able to swing over the existing or proposed f. That part of the access extending 6 metres into the site from the carriageway of the existing highway shall be at a gradient not exceeding 1:15; h. The final surfacing of any private access within 6 metres of the public highway shall not contain any loose material that is capable of being drawn on to the existing or proposed public highway; ΑII works accord with the approved details unless otherwise agreed in writing by the Local Planning Authority.
- 11. There shall be no access or egress by any vehicles between the highway and the application site (except for the purposes of constructing the initial site access) until splays are provided giving clear visibility of 43 metres measured along both channel lines of the major road C10 from a point measured 2 metres down the centre line of the access road. The eye height will be 1.05 metres and the object height shall be 0.6 metres. Once created, these visibility areas shall be maintained clear of any obstruction and retained for their intended purpose at all times.
- 12. Unless otherwise approved in writing by the Local Planning Authority, there shall be no excavation or other groundworks, except for investigative works, or the depositing of material on the site in connection with the construction of the access road or building(s) or other works hereby permitted until full details of the following have been submitted to and approved in writing by the Local Planning:b. vehicular and pedestrian accesses c. vehicular parking d. vehicular turning arrangements e. manoeuvring arrangements
- 13. There shall be no access or egress by any vehicles between the highway and the application site until details of the precautions to be taken to prevent the deposit of mud, grit and dirt on public highways by vehicles travelling to and from the site have been submitted to and approved in writing by the Local Planning Authority. These facilities shall include the provision of wheel washing facilities where considered necessary by the Local Planning Authority. These precautions shall be made available before any excavation or depositing of material in connection with the construction commences on the site and be kept available and in full working order and used until such time as the Local Planning Authority agrees in writing to their withdrawal.
- 14. Unless approved otherwise in writing by the Local Planning Authority there shall be no establishment of a site compound, site clearance, demolition, excavation or depositing of material in connection with the construction on the site until proposals have been submitted to and approved in writing by the Local Planning Authority for the provision of:

 a. on-site parking capable of accommodating all staff and sub-contractors vehicles clear of the public highway b. on-site materials storage area capable of accommodating all materials required for the operation of the site. c. The

approved areas shall be kept available for their intended use at all times that construction works are in operation.

15. The permission hereby granted shall not be undertaken other than in complete accordance with the location plan and email received by Hambleton District Council on 23 September and 12 October 2015 unless otherwise approved in writing by the Local Planning Authority.

The reasons for the above conditions are:-

- 1. To ensure compliance with Section 92 of the Town and Country Planning Act, 1990
- 2. To enable the Local Planning Authority to properly assess these aspects of the proposal, which are considered to be of particular importance, before the development is commenced.
- 3. To ensure that the external appearance of the development is compatible with the immediate surroundings of the site and the area as a whole in accordance with Hambleton Local Development Framework Policy CP17.
- 4. To reduce the volume and rate of surface water that drains to sewers and watercourses and thereby not worsen the potential for flooding in accordance with Hambleton LDF Policies CP21 and DP43.
- 5. In order to soften the visual appearance of the development and provide any appropriate screening to adjoining properties in accordance with LDF Policies CP16 and DP30.
- 6. In order to soften the visual appearance of the development in accordance with LDF Policies CP16 and DP28.
- 7. To protect the amenity of the neighbouring residents and to ensure that the development is appropriate to the character and appearance of its surroundings in accordance with Policies CP1, DP1, CP17 and DP32 of the Hambleton Local Development Framework.
- 8. To ensure that the site is properly drained and surface water is not discharged to the foul sewerage system which will prevent overloading in accordance with LDF Policies CP21 and DP43.
- 9. In accordance with LDF Policies CP2 and DP4 and in the interests of highway safety.
- 10. In accordance with LDF Policies CP2 and DP4 and to ensure a satisfactory means of access to the site from the public highway in the interests of vehicle and pedestrian safety and convenience.
- 11. In accordance with LDF Policies CP2 and DP4 and in the interests of road safety.
- 12. In accordance with LDF Policies CP2 and DP4 and to ensure appropriate on-site facilities in the interests of highway safety and the general amenity of the development.

- 13. In accordance with LDF Policies CP2 and DP4 and to ensure that no mud or other debris is deposited on the carriageway in the interests of highway safety.
- 14. In accordance with LDF Policies CP2 and DP4 and to provide for appropriate on-site vehicle parking and storage facilities, in the interests of highway safety and the general amenity of the area.
- 15. In order that the development is undertaken in a form that is appropriate to the character and appearance of its surroundings and in accordance with the Development Plan Policies.

Parish: Potto

Ward: Osmotherley & Swainby

6

Committee Date: 12 November 2015 Officer dealing: Mrs B Robinson Target Date: 12 November 2015

15/01823/FUL

Construction of detached dwellinghouse and double garage. at Land At The Old Orchard Cooper Lane Potto North Yorkshire for Mr J Graham.

1.0 SITE DESCRIPTION AND PROPOSAL

- 1.1 The site forms part of a grassed paddock at the rear of a single storey house The Old Orchard. It lies to the west of the car park serving the Dog and Gun public house. The car park pub and Old Orchard are sited on land elevated from the road level and accessed up a short slope. The Old Orchard lies at the north end of Potto village. The plot is enclosed on the south and east side by a close boarded timber fence approximately 2 metres high and is enclosed on the west side by a hedge. To the west is an agricultural style building, with consent for use as storage.
- 1.2 The site is roughly rectangular, dimensions 30m x 16m. The site is accessed across the south side of the car park to the Dog and Gun, which also serves the rear of dwellings located on the western 'limb' of Cooper Lane, to the south.
- 1.3 The proposal is a two storey dwelling, facing east, with attached double garage to the front, forming an L shape. The house and garage are intended to be faced in coursed stone work on their front façade. The building is otherwise rendered. The roof is to be red clay pantiles. The building is broadly traditional in style with water tabling to the roof edges of house and garage and a chimney on the south gable end.

2.0 RELEVANT PLANNING HISTORY

2.1 The Old Orchard has a planning history of conversion and extension, from its origins as a garage building.

3.0 NATIONAL AND LOCAL POLICY:

3.1 The relevant policy of the Development Plan and any supplementary planning policy advice are as follows;

Core Strategy Policy CP1 - Sustainable development

Development Policies DP1 - Protecting amenity

Core Strategy Policy CP2 - Access

Core Strategy Policy CP4 - Settlement hierarchy

Interim Guidance Note - adopted by Council on 7th April 2015

Core Strategy Policy CP17 - Promoting high quality design

Development Policies DP32 - General design

National Planning Policy Framework

4.0 CONSULTATIONS

4.1 Parish Council - Approve. Additional comments by the PC on;

i.Planning statement

Interim Policy Guidance Note and precedent for future developments; house not 'affordable'; development this side of Cooper Lane not part of traditional village; gable end will have impact on outlook from nos 6 and 8 (Cooper Lane); impact of pub and car park activities on occupiers, and concerns about safety (construction vehicles across car park); existing access serves 4 properties - 4-8 Cooper Lane, not 2, as stated (in planning statement).

ii. Proposed design

No objection to design; clarification needed whether septic tank or mains drainage.

- 4.2 NYCC Highways conditions requested.
- 4.3 Northumbrian Water no comments to make.
- 4.4 Site notice and neighbours Objections (summarised)
- i. Access is across neighbours property and the right of access is disputed.
- ii. Outside development limits and since 2010 no further development in Potto. Lack of infrastructure in village
- iii.House is not in-keeping with area, this type of house available relatively close by eg at Ingleby Barwick. No local demand for this type of 'executive' home
- iv. Precedent
- v. History of creeping development at The Old Orchard, from origins as a detached garage, originally and successive development to form dwelling with its own detached garage.
- vi. Noise nuisance from pub, including smoking facilities.
- vii. Not in accordance with Potto Parish development plan
- viii access from the proposed gates into car park would be awkward and hazardous to others using the lane.
- viii. Reduced area of (original) pub car park, which is now insufficient.

5.0 OBSERVATIONS

5.1 Potto is a village without status within the hierarchy set out in CP4 as adopted 2008. In 2015 the Council has adopted an Interim Guidance Note which updates the hierarchy and includes Potto as an 'other' settlement. The Guidance Note provides for a more flexible consideration of new development at the edge of settlements. Accordingly, the main consideration is whether the development is in accordance with the criteria of the Interim Guidance, and thereafter whether it is in accordance with any other relevant policies of the local plan including the amenity of nearby occupiers (CP1 and DP1, design (CP17 and DP32) and any highway safety issues.

5.2 Interim Guidance Criteria:

Small scale housing development will be supported in villages where it contributes towards achieving sustainable development by maintaining or enhancing the vitality of the local community AND where it meets ALL of the following criteria:

- 1. Development should be located where it will support local services including services in a village nearby.
- 2. Development must be small in scale, reflecting the existing built form and character of the village.
- 3. Development must not have a detrimental impact on the natural, built and historic environment.
- 4. Development should have no detrimental impact on the open character and appearance of the surrounding countryside or lead to the coalescence of settlements.
- 5. Development must be capable of being accommodated within the capacity of existing or planned infrastructure.
- 6. Development must conform with all other relevant LDF policies.
- 5.3 Potto includes a pub and a modern village hall offering a range activities and is relatively close to the wider range of facilities at Hutton Rudby. Overall it is a reasonably sustainable location where an additional dwelling can be considered to support local services.
- 5.4 As a single dwelling it is small in scale. The proposal is back-land development in relation to the road however it is contained within the existing built pattern of the village and is an acceptable fit with the existing built form of the village.
- 5.5 The site is a grassed area associated with an existing house and development of the nature proposed would not be detrimental to the natural environment. The surroundings do not include Listed Buildings and the development would not be harmful to the historic or built

environment, subject to considerations of design.

- 5.6 The site is within the built up part of the village and will not impact on the rural surroundings.
- 5.7 There is no evidence that a single extra dwelling cannot be accommodated within the capacity of the existing infrastructure.
- 5.8 Overall the proposed development is considered to be acceptable in the terms of the Interim Policy Guidance.

Design

5.9 The proposed house uses traditional design detailing, including coursed stone on the front, and water tabling features. The upper windows are 'tucked' in close to the eaves and the overall effect is traditional and appropriate in the village setting. Other properties nearby are mainly c20th and the proposed house will not affect the setting of any Listed Building. The proposal provides for private amenity space on the south west side. Because of the set back from the public road the proposed house will not have a significant impact on the nearby street-scene. From a greater distance, the upper parts will be viewed as part of the general 'huddle' of houses on the western limb of Cooper Lane.

Amenity of neighbours and residents of the proposed house

- 5.10 With regard to the amenity of neighbours, the rear of nos 6 and 8 Cooper Lane will face the proposed dwelling, but there is sufficient distance (approximately 26 metres) to ensure that amenities are not unacceptably affected. The proposed house is offset from the main outlook of the Old Orchard and over 30m away and will not be harmful to the amenities of occupiers there.
- 5.11 Occupiers of the proposed house will be in close proximity to the pub and potentially vulnerable to noise and activity during opening hours. However, this is not n unusual relationship locally and the existing pub will have a similar impact on other existing properties. This has not been a significant issue in the past and there is no reason to believe that this would be a problem in the future. The proposed dwelling will benefit from a satisfactory level of amenity.

Access and highway safety

5.12 The site is accessed across the pub car park. This is a feature of other nearby properties, and would not in principle preclude approval. The right to use the westerly part of the access route is disputed by neighbouring owners and this is a matter which any developer would have to resolve and it will be appropriate to draw attention to the point in an informative, this is otherwise not a matter for the consideration of this application.

Parish Council and neighbour observations

- 5.13 Comments draw attention to the restrictions to development in Potto that have been in place in recent years. However, the more recent Interim Policy Guidance gives scope to consider small developments, subject to criteria, and refusal for this reason would not be justified.
- 5.14 The planning history of the Old Orchard is not material to the consideration of this proposal.
- 5.15 The site is not part of the pub car park and the proposal has a neutral effect on parking provision for the public house.
- 5.16 Potto Village Plan nearing the end of its life (2008 -2017) and policies of restraint are to be considered in the light of more recent government priorities and the Councils responses, including the Interim Guidance.

- 5.17 As noted above the design of the proposed house is considered to be in-keeping with the traditions of the area. The proposed dwelling will not be an affordable dwelling, however this would not be required for this development. In scale it is in keeping with neighbouring houses on Cooper Lane. The proposal will not have an unacceptable harmful effect on the amenities of neighbours.
- 5.18 Safety during the construction period will be a matter for the developer of the site and will be dependant on the good sense of drivers and others using the car park but particularly taking into account the likely lesser use of the pub car park during day time hours when works will be undertaken. This issue is not considered to preclude the approval of this scheme.
- 5.19 The access is at the corner of the site where there is scope for a reasonably convenient turn onto the access track, and taking into account the low level of use of the further part of the drive to no 8, normal attentive driving should ensure that safety is not compromised.
- 5.20 The development is CIL liable.

SUMMARY

- 5.21 Due to its design and materials and location the proposal is appropriate to the local setting and will not have a harmful effect on the amenities of neighbouring occupiers or the wider surroundings and is able to comply with the above policies.
- 5.22 The Local Planning Authority has acted positively and proactively in determining this application by assessing the proposal against all material considerations, including planning policies and any representations that may have been received and subsequently determining to grant planning permission in accordance with the presumption in favour of sustainable development, as set out within the National Planning Policy Framework.

6.0 RECOMMENDATION:

- 6.1 That subject to any outstanding consultations the application be **GRANTED** subject to the following condition(s)
 - 1. The development hereby permitted shall be begun within three years of the date of this permission.
 - 2. The permission hereby granted shall not be undertaken other than in complete accordance with the drawing(s) numbered 01 A; 02 B; received by Hambleton District Council on 10th of August and 2nd September 2015 unless otherwise approved in writing by the Local Planning Authority.
 - 3. Prior to development commencing, details and samples of the materials to be used in the construction of the external surfaces of the development shall be made available on the application site for inspection and the Local Planning Authority shall be advised that the materials are on site and the materials shall be approved in writing by the Local Planning Authority. The development shall be constructed of the approved materials in accordance with the approved method.
 - 4. The development hereby approved shall not be commenced until details of the foul sewage and surface water disposal facilities have been submitted and approved in writing by the Local Planning Authority.
 - 5. The use of the development hereby approved shall not be commenced

until the foul sewage and surface water disposal facilities have been constructed and brought into use in accordance with the details approved under condition 4 above.

- 6. The development shall not be commenced until details relating to boundary walls, fences and other means of enclosure for all parts of the development have been submitted to and approved in writing by the Local Planning Authority
- 7. No dwelling shall be occupied until the related parking facilities have been constructed in accordance with the approved drawing. Once created these parking areas shall be maintained clear of any obstruction and retained for their intended purpose at all times.

Community Infrastructure Levy

Please note that the proposed development is liable under the Community Infrastructure Levy Charging Schedule, adopted by Hambleton District Council on the 07 April 2015. Details of the charging schedule are available on the Council website. www.hambleton.gov.uk

The reasons for the above conditions are:-

- 1. To ensure compliance with Sections 91 and 92 of the Town and Country Planning Act 1990 and where appropriate as amended by Section 51 of the Planning and Compulsory Purchase Act 2004.
- 2. In order that the development is undertaken in a form that is appropriate to the character and appearance of its surroundings and in accordance with the Development Plan Policy(ies) .
- 3. To ensure that the external appearance of the development is compatible with the immediate surroundings of the site and the area as a whole in accordance with Hambleton Local Development Framework Policy CP17.
- 4. In order to avoid the pollution of watercourses and land in accordance with Local Development Framework CP21 and DP43
- 5. In order to avoid the pollution of watercourses and land in accordance with Local Development Framework CP21 and DP43
- 6. To protect the amenity of the neighbouring residents and to ensure that the development is appropriate to the character and appearance of its surroundings.
- 7. In accordance with policy CP2 and DP4 and to provide for adequate and satisfactory provision of off-street accommodation for vehicles in the interest of safety and the general amenity of the development.



Parish: Northallerton Committee Date : 12 November 2015

Ward: Northallerton South Officer dealing: Mrs T Price Target Date: 20 October 2015

15/01897/OUT

Outline application for the demolition of the band room and construction of 4 dwellings as amended by plans received by Hambleton District Council on 18th September 2015 and 19th October 201

1.0 SITE DESCRIPTION AND PROPOSAL

- 1.1 This application seeks outline planning permission for the demolition of the existing band room and construction of 3 flats and a dwellinghouse at The Band Room, Romanby Road in Northallerton.
- 1.2 The plot is rectangular in shape measuring approximately 15m wide and 24m deep. The north east part of the site is occupied by the existing two storey flat roofed band room and single storey pitched roofed wing and the south part by gravelled car parking. Northallerton Silver Band use the building up to four nights per week for rehearsal and meeting purposes. The unsurfaced car park adjacent to the building provides parking for 9-10 cars and other band members make use of on street parking as necessary. The site is generally flat and has no existing planting or landscape features. Road access is direct from Romanby Road though a narrow gap between low walls with shared use for residents' access to the rear of Bridge Terrace.
- 1.3 Outline planning permission is sought considering access only with layout, design and landscaping to be dealt with at reserved matters. Revised preliminary drawings submitted on the 19th October 2015 show that the new development would reinforce the linear nature of the existing street-scene between Bridge Terrace and 90-96 Romanby Road and a similar height to the adjoining properties on Romanby Road. The access is to remain off Romanby Road as existing.
- 1.4 The Silver Band is to locate to an alternative location in Northallerton.

2.0 RELELVANT HISTORY

- 2.1 15/01196/OUT Demolition of band room and construction of 3 flats and 1 dwellinghouse; withdrawn 26 August 2015.
- 2.2 07/02088/FUL Single storey extension to existing building; Granted 18 September 2007.
- 2.3 2/03/500/0077A Variation of condition 03 of Planning decision notice reference 2/95/500/0077 to allow the increase in use of the band practice room to four nights a week 6.00 pm to 9.30pm and Saturday afternoons 1.30pm to 4.30pm; Granted 2 March 2004.
- 2.4 2/95/500/0077, Alterations to existing ground floor office/storage accommodation for use in connection with existing first floor band room and demolition of existing adjoining disused dwelling, approved 13.10.1995
- 2.5 2/91/123/0006H Construction of 2 Terraced Dwellings; Granted 2 September 1991.

3.0 NATIONAL AND LOCAL POLICY

3.1 The relevant policies are:

Core Strategy Policy CP1 - Sustainable development

Core Strategy Policy CP2 - Access

Core Strategy Policy CP4 - Settlement hierarchy

Core Strategy Policy CP17 - Promoting high quality design

Development Policies DP1 - Protecting amenity

Development Policies DP4 - Access for all

Development Policies DP6 - Utilities and infrastructure

Development Policies DP32 - General design

National Planning Policy Framework

4.0 CONSULTATIONS

- 4.1 Town Council Refuse until the vehicle access and parking is solved. The size and scope of the development is not in keeping with the surrounding area and the application is not in the interest of the residents. Recommend deferring until the problems are resolved and for it to go to Committee.
- 4.2 NYCC Highways No objection to the scheme subject to condition
- 4.3 Neighbours/site notice expired 20th July 2015 Eleven representations have been received and are briefly summarised below:
 - Opposition to three storey development; it would dwarf the area
 - Where will the building materials go
 - Parking concerns
 - Drainage concerns, current problems in the area
 - No room for refuse collections
 - Greedy development
 - Narrow access
 - Any covenants on the land?
 - Site too small
 - Back Lane belongs to Bridge Terrace residents; no right of access
 - Overlook private areas
 - Not consulted wide enough within the area
 - Private access road, no permission
 - 4 parking spaces too few
 - The new application remains the same
 - Access along a narrow lane
 - Ownership/access issues
 - No engagement with local community
 - Highway safety; Romanby Road is busy
 - A single dwelling should be considered
 - Band room only has permission for pedestrian access
- 4.4 Environmental Health Officer No objection
- 4.5 Yorkshire Water No observations
- 4.5 Network Rail No observations
- 4.6 Allertonshire Civic Society Believe this to be a scheme which verges on the excellent in both its devising and planning. As clearly expressed in their Design and Access statement it effectively addresses the broad range of issues which currently exist at this location, for the neighbouring properties, road users and the members of the Silver Band. First, the proposal removes what has to be admitted has long been a

'sore thumb' amid the local street scene. It will eradicate the last visual remnant of the old industrial area of steel works, foundry, linoleum factory, joinery workshop and coal yards, which were the product of the previously valuable but uncontrolled and unfortunate consequence of 'mixed development'. The outcome concept is admirable and complements the scale and size of the now adjoining persimmon estate, without bearing too heavily on the older terrace to the other side. Whilst the eventual architectural details are to be determined, the draft elevations seem satisfactory as an overall basis. The general appearance accords with modern preferences for this district. The tall gable facing onto the road creates an interesting variation in what otherwise could become a rather monotonous procession of similar buildings, even when having different styles. The suggested interior and ground layouts make ingenious use of the space available and more than resolve the present problems with the limited shared vehicle access and parking. We heartily endorse this proposed development and offer our support for the outline planning application

5.0 OBSERVATIONS

- 5.1 The main issues for consideration in this case relate to the principle of the outline application in this location together with any impact it may have upon the character of the area, the amenities of the neighbours and Highway safety.
- 5.2 The site is previously developed land which is located within the main development limits of Northallerton. Under the requirements of Policy CP4 and the provisions of the NPPF the proposed residential development is in principle acceptable as the site is located within a sustainable location. It is considered that aesthetically the site would be much improved through redevelopment as in its current state is harmful to the character or the area.
- 5.3 As the application is specifically looking at the principle of the development including the access only, the layout, scale, design and landscaping will be dealt with in the reserved matters application.
- 5.4 The site is within a largely residential area which is made up of two and three storey terraces and flats. The proposal is to form 4 dwellings but as this is at outline stage no details of heights or building size have been formally submitted. Taking into account that the properties in the area range from 2-3 storeys in height it is considered that the suggested 2.5 storey development in principle would be appropriate.
- 5.5 The Highway Authority has not objected to the application but has recommended conditions relating to discharge of surface water, turning and parking areas, parking for dwellings, precautions to prevent mud on the highway, on-site parking, on-site storage and construction traffic during the development. It is therefore considered that highway safety will not be compromised by a housing scheme of this nature.
- 5.6 Eleven representations have been received with concern regarding building height, parking arrangements, privacy issues, bin storage, scale and construction; all these issues are to be dealt with at the reserved matters stage. The drainage concerns can be dealt with through a planning condition to ensure the existing network is not compromised. The issue of covenants on the land and the access arrangements and ownership have been raised. However, these are considered civil matters and cannot be taken into consideration as a planning issue. It is noted that the applicant has submitted information which claims the site has right of access by a covenant. It is also noted that the application has been advertised by site notice and the closest 18 properties have been notified by letter.
- 5.7 Under the Community Infrastructure Levy (CIL) regulations the proposed dwellings

- are liable for payment of CIL at a rate of £55 per sq. m, the rate adopted by the Council on 7 April 2015.
- 5.8 Based on the above, the principle of the development is considered to be acceptable. The proposal in its outline form accords with the relevant national and local policies.

6.0 RECOMMENDATION

- 6.1 That subject to any outstanding consultations the application is **GRANTED** subject to the following conditions:
- 1. Application for the approval of all of the reserved matters shall be made to the Local Planning Authority not later than three years from the date of this decision and the development hereby approved shall be begun on or before whichever is the later of the following dates: i) Five years from the date of this permission ii) The expiration of two years from the final approval of the reserved matters or in the case of approval on different dates, the final approval of the last such matter to be approved.
- 2. No development shall commence until details of all the reserved matters have been submitted to and approved by the Local Planning Authority: (b) the siting, design and external appearance of each building, including a schedule of external materials to be used; (c) the landscaping of the site; (d) the layout of the proposed building(s) and space(s) including parking and any external storage areas; and (e) the scale (including the number) of buildings overall.
- 3. There shall be no access or egress by any vehicles between the highway and the application site until full details of any measures required to prevent surface water from non-highway areas discharging on to the existing or proposed highway together with a programme for their implementation have been submitted to and approved in writing by the Local Planning Authority in consultation with the Highway Authority. The works shall be implemented in accordance with the approved details and programme.
- 4. No part of the development shall be brought into use until the approved vehicle access, parking, manoeuvring and turning areas have been approved in writing by the Local Planning Authority. Once created these areas shall be maintained clear of any obstruction and retained for their intended purpose at all times
- 5. No dwelling shall be occupied until the related parking facilities have been constructed in accordance with the approved drawing "Proposed Redevelopment Bandroom, Romanby Road Outline Application Plan". Once created these parking areas shall be maintained clear of any obstruction and retained for their intended purpose at all times.
- 6. There shall be no access or egress by any vehicles between the highway and the application site until details of the precautions to be taken to prevent the deposit of mud, grit and dirt on public highways by vehicles travelling to and from the site have been submitted to and approved in writing by the Local Planning Authority in consultation with the Highway Authority. These facilities shall include the provision of wheel washing facilities where considered necessary by the Local Planning Authority in consultation with the Highway Authority. These precautions shall be made available before any excavation or depositing of material in connection with the construction commences on the site and be kept available and in full working order and used until such time as the Local Planning Authority in consultation with the Highway Authority agrees in writing to their withdrawal.
- 7. Unless approved otherwise in writing by the Local Planning Authority there shall be

no establishment of a site compound, site clearance, demolition, excavation or depositing of material in connection with the construction on the site until proposals have been submitted to and approved in writing by the Local Planning Authority for the provision of: a. on-site parking capable of accommodating all staff and subcontractors vehicles clear of the public highway; b. on-site materials storage area capable of accommodating all materials required for the operation of the site. The approved areas shall be kept available for their intended use at all times that construction works are in operation.

- 8. The development hereby approved shall not be commenced until details of the foul sewage and surface water disposal facilities have been submitted and approved in writing by the Local Planning Authority.
- 9. The permission hereby granted shall not be undertaken other than in complete accordance with the drawing(s) numbered 15023D(00)01revF received by Hambleton District Council on 19th October 2015 unless otherwise approved in writing by the Local Planning Authority.

The reasons for the above conditions are:

- 1. To ensure compliance with Section 92 of the Town and Country Planning Act, 1990
- To enable the Local Planning Authority to properly assess these aspects of the proposal, which are considered to be of particular importance, before the development is commenced.
- 3. In accordance with the policy and in the interests of highway safety.
- 4. In accordance with the policy and to provide for appropriate on-site vehicle facilities in the interests of highway safety and the general amenity of the development.
- 5. In accordance with the policy and to provide for adequate and satisfactory provision of off-street accommodation for vehicles in the interest of safety and the general amenity of the development.
- 6. In accordance with the policy and to ensure that no mud or other debris is deposited on the carriageway in the interests of highway safety.
- 7. In accordance with the policy and to provide for appropriate on-site vehicle parking and storage facilities, in the interests of highway safety and the general amenity of the area.
- 8. In order to avoid the pollution of watercourses and land in accordance with Local Development Framework CP21 and DP43
- 9. In order that the development is undertaken in a form that is appropriate to the character and appearance of its surroundings and in accordance with the Development Plan Policy(ies) CP1, CP2, CP4, CP17, DP1, DP4, DP32.

Informative

Please note that the proposed development is liable under the Community Infrastructure Levy Charging Schedule, adopted by Hambleton District Council on the 07 April 2015. Details of the charging schedule are available on the Council website. www.hambleton.gov.uk



Parish: Sessay

Ward: Sowerby & Topcliffe Officer dealing: Mr A J Cunningham

Committee Date :

12 November 2015

Target Date: 25 August 2015

15/01446/FUL

Construction of detached three bedroom two storey dwelling with attached garage to include access and means of enclosure as amended by plans and details received by Hambleton District Council on 6 August 2015, 1 September 2015 and 21 October 2015. at Montrose Main Street Sessay North Yorkshire for Mr & Mrs Algie.

1.0 PROPOSAL AND SITE DESCRIPTION

- 1.1 This application seeks planning consent for the construction of a dwelling and garage at land within the garden of and north of the dwelling known as "Montrose" that is in turn at the northern end of the village of Sessay, on the west side of the village street.
- 1.2 The proposed dwelling was initially to be served by the existing access to the north of the site at Montrose. However the amended details (received 21 October 2015) show an access central to the site frontage to provide access to both Montrose and the new dwelling, in order to achieve the required visibility splays.
- 1.3 Amended plans have been received on 6 August 2015 and 1 September 2015 revising the design of the dwelling, clarifying on the access proposed and the concerns of the local highway authority, confirming surface water drainage, and works to trees on site.
- 1.4 The dwelling would provide for a hall, lounge, dining area, w/c, utility and kitchen area at ground floor level. At first floor level 2 bedrooms, a bathroom, study and an ensuite bedroom are proposed.
- 1.5 The dwelling would measure approximately 8.6m x 10.8m, with a total height of approximately 7.3m. An attached garage to the south would measure approximately 5.7m x 6.3m, with a total height of approximately 4.8m.
- 1.6 The dwelling would be formed of brickwork and clay pantiles with white UPVC double glazed windows and doors.

2.0 RELEVANT PLANNING AND ENFORCEMENT HISTORY

- 2.1 10/00773/OUT Outline application for the construction of a dwelling; Withdrawn 2010.
- 2.2 10/00774/OUT Outline application for the construction of a dwelling; Withdrawn 2010.
- 2.3 There is no planning enforcement history.
- 2.4 15/00553/OUT Development on the east side of the village street in the centre of the village for a new village hall and 5 dwellings Approved 4 June 2015
- 2.5 15/00408/OUT Development on the east side of the village street 150m from the current site, outline for 5 dwellings Approved 22 June 2015

3.0 RELEVANT PLANNING POLICIES:

3.1 The relevant policy of the Development Plan and any supplementary planning policy advice are as follows;

Core Strategy Policy CP1 - Sustainable development

Core Strategy Policy CP2 - Access

Core Strategy Policy CP4 - Settlement hierarchy

Core Strategy Policy CP16 - Protecting and enhancing natural and man-made assets

Core Strategy Policy CP17 - Promoting high quality design

Core Strategy Policy CP19 - Recreational facilities and amenity open space

Development Policies DP1 - Protecting amenity

Development Policies DP2 - Securing developer contributions

Development Policies DP3 - Site accessibility

Development Policies DP4 - Access for all

Development Policies DP9 - Development outside Development Limits

Development Policies DP10 - Form and character of settlements

Development Policies DP30 - Protecting the character and appearance of the countryside

Development Policies DP32 - General design

Development Policies DP37 - Open space, sport and recreation

Development Policies DP43 - Flooding and floodplains

Interim Guidance Note - adopted by Council on 7th April 2015

4.0 CONSULTATIONS

4.1 Parish Council - comments made on cumulative scale of development is Sessay, design of house must respect local vernacular, issues of surface water flooding and poor drainage, replacement landscaping, likelihood of protected species, importance of detailing of dwelling, support garage being set back, use of traditional fenestration material, chimney stack needs re-designing.

In summary: support the principle of a house of this size, height and form in this location subject to the issues set out above being addressed. Commenting on the amended details of 1st September 2015 the Parish Council acknowledges the changes that have been made and have no further comments.

4.2 NYCC Highways - The design standard for the site is Manual for Streets and the required visibility splay is 2 metres by 43 metres. The available visibility as initially shown from the northern access was 2 metres by 12 metres. Consequently, the Local Highway Authority recommended that permission be refused.

The amended scheme has been the subject of discussion between the agent and the highway authority and is expected to be found acceptable, the final response is awaited.

4.3 EHO - This service has considered the potential impact on amenity and likelihood of the development to cause a nuisance and consider that there will be no negative impact. Therefore the Environmental Health Service has no objections

4.4 Yorkshire Water

4.5 Publicity – neighbours

A comment has been received regarding a) the access and b) whether this is the start of a large number of properties ruining the open aspect of the area. There is concern about the road width and the safety of new properties noting the existing problem of vehicles speeding through the village.

5.0 OBSERVATIONS

5.1 The main issues to be considered in this case is are the highway safety of the site access, the landscape impacts of the proposal and the whether the scheme is in

- compliance with the Interim Planning Guidance and can therefore be recommended for approval.
- The safety of the access of the site to the village street has been raised with the agent and discussed directly between the agent and the highway authority engineer. Amended plans have been supplied that show an access in a position with increased visibility splays. It is anticipated that the revised details will be found to be acceptable to the highway authority and that a recommendation of conditions to be applied to any permission will be supplied before the date of the Committee meeting.
- 5.3 The site is currently a garden to Montrose and has the appearance of a garden with shrubs trees and garden structures. The site is opposite open agricultural land and to the north and west is open land. The construction of a new dwelling on the site will have a significant impact but this will be limited by the relationship to the existing dwelling Montrose (that is currently the first of a long (about 1.1km) unbroken row of dwellings on the west side of the village) and that the proposal would extend that row. The final structure on the west side of the village street will remain a small brick built sewage pump house. A section of mixed thorn hedge and a few tall Leylandii trees and other smaller shrubs would be required to be removed to make way for the access and dwelling. Orchard trees are shown to be retained to the north of the proposed dwelling. It is considered that taking in to account the relationship to the dwelling, the pump house, the retained hedgerow and orchard trees the proposal will not cause significant harm to the openness of the landscape of the District.
- 5.4 The Interim Policy Guidance (IPG) sets out small scale housing proposals can be supported. Firstly the proposal is to contribute towards achieving sustainable development by maintaining or enhancing the vitality of the local community and then secondly the proposal must meet all the 6 criteria. Sessay is a small village with a community hall (and approval for a proposed replacement hall), with village school and Parish Church at the south end of the village. There are small businesses operating around the village including the angling lakes and caravan site (with approval for extension) at the north end of the village. Schemes have already been approved for not just the new village hall but also housing on the site of the current village hall and approval for housing 150m from the proposal site, on the northern end of the village on the east side of the village street. The village can be described as having vitality and there is no reason to conclude that the proposal for an additional dwelling would not assist with furthering that vitality. It is considered that the first test of the IPG is met.
- 5.5 The second test set by the IPG is the 6 criteria. The site is about 750 from the village hall in the centre of the village. This is within walking distance and despite the absence of paths through much of the northern part of the village the village street is not so busy with traffic to make walking unattractive. Beyond the centre of the village there is a footway that leads to the village school. It is considered that the few local services in the village will be support by construction of new dwellings.
- 5.6 The development of a single dwelling is small scale and the layout reflects the form of the linear village. As found above building a single dwelling would not cause harm to the landscape. There is no evidence of any natural or historic interests on or adjoining the site. Even though the scheme would extend the linear form of the village it would not lead to the coalescence of settlements.
- 5.7 The scheme if approved would be the 11th new dwelling to be approved in Sessay since the adoption of the IPG. The scheme for a further dwelling would not give rise to a demonstrable increased pressure on any local infrastructure that would give rise to any demonstrable harm to the existing community. For these reasons it is considered that the scheme achieves compliance with 5 of the 6 criteria of the IPG.

The final criteria is that the proposal conforms with the other relevant LDF policies. Subject to the final comments of the highway authority it is considered that the scheme does conform with the LDF Policies.

- 5.8 The Parish Council and neighbour comments are considered to have addressed by the amendments made. The design of the building is considered to be appropriate to the village and whilst larger than the older cottage 'Montrose" it would not have an overbearing impact upon the neighbouring property or create a harmful impact upon the streetscene.
- 5.9 The issues of poor drainage and flooding are matters that are stated in the application forms to be addressed by disposal to the main sewer and existing drain to a water course, through the submission of details by a planning condition further details can be required to ensure that the development would provide suitable drainage for the site and not worsen any existing problems.

6.0 RECOMMENDATION:

- 6.1 That subject to any outstanding consultations the application be **GRANTED**
 - 1. The development hereby permitted shall be begun within three years of the date of this permission.
 - 2. Prior to development commencing, details and samples of the materials to be used in the construction of the external surfaces of the development shall be made available on the application site for inspection and the Local Planning Authority shall be advised that the materials are on site and the materials shall be approved in writing by the Local Planning Authority. The development shall be constructed of the approved materials in accordance with the approved method.
 - 3. The development hereby approved shall not be commenced until details of the foul sewage and surface water disposal facilities have been submitted and approved in writing by the Local Planning Authority.
 - 4. The use of the development hereby approved shall not be commenced until the foul sewage and surface water disposal facilities have been constructed and brought into use in accordance with the details approved under condition above.
 - 5. Prior to development commencing detailed cross sections shall be submitted to and approved in writing by the Local Planning Authority, showing the existing ground levels in relation to the proposed ground and finished floor levels for the development. The levels shall relate to a fixed Ordnance Datum. The development shall be constructed in accordance with the approved details and thereafter be retained in the approved form.
 - 6. No development shall commence until a scheme for highway works has been submitted to and approved in writing by the Local Planning Authority. The scheme of works shall include a programme for the implementation and completion of works. The development shall proceed in accordance with the approved scheme in accordance with the approved programme. The scheme once implemented shall be retained in accordance with the approved scheme.
 - 7. The development shall not be commenced until a detailed landscaping scheme indicating the type, height, species and location of all new trees and

shrubs, has been submitted to and approved by the Local Planning Authority. No part of the development shall be used after the end of the first planting and seeding seasons following the approval of the landscaping scheme, unless the approved scheme has been completed. Any trees or plants which within a period of 5 years of planting die, are removed, or become seriously damaged or diseased, shall be replaced with others of similar size and species.

8. The permission hereby granted shall not be undertaken other than in complete accordance with the drawing(s) numbered 368/PD/03A, 05C, 01E and 02E received by Hambleton District Council on 21 October 2015 unless otherwise approved in writing by the Local Planning Authority.

The reasons are:-

- 1. To ensure compliance with Sections 91 and 92 of the Town and Country Planning Act 1990 and where appropriate as amended by Section 51 of the Planning and Compulsory Purchase Act 2004.
- 2. To ensure that the external appearance of the development is compatible with the immediate surroundings of the site and the area as a whole in accordance with Hambleton Local Development Framework Policy CP17.
- 3. In order to avoid the pollution of watercourses and land in accordance with Local Development Framework CP21 and DP43
- 4. In order to avoid the pollution of watercourses and land in accordance with Local Development Framework CP21 and DP43
- 5. To ensure that the development is appropriate in terms of amenity in accordance with Local Development Framework Policies CP1 and DP1.
- 6. In order to ensure that the development does not give rise to a loss of highway safety in accordance with the LDF Policies CP1, DP3 and DP4.
- 7. In order to soften the visual appearance of the development and provide any appropriate screening to adjoining properties in accordance with Local Development Framework Policy .
- 8. In order that the development is undertaken in a form that is appropriate to the character and appearance of its surroundings and in accordance with the Development Plan Policy(ies) CP1, DP1, CP17 and DP32.



Parish: StokesleyCommittee Date :12 November 2015Ward: StokesleyOfficer dealing :Mr Peter Jones

9

Target Date: 27 November 2015

15/01943/REM

Application for Reserved Matters for the construction of 178 dwellings. at White House Farm Stokesley North Yorkshire TS9 5LE for Taylor Wimpey (North Yorkshire) Ltd.

- 1.0 SITE DESCRIPTION AND PROPOSAL
- 1.1 This application is to deal with the Reserved Matters for the construction of 179 dwellings at Whitehouse Farm, Stokesley. This application is submitted following the grant of outline planning permission for the construction of up to 183 dwellings with associated infrastructure. All matters were reserved for subsequent approval, with the exception of access. The application site is an allocated site in the LDF Allocations Document site SH1, White House Farm, Stokesley.
- 1.2 To the west of the application site lies the main access road to Crab Tree Farm which lies to the north west of the application site and which will be separated from the new development by existing fields. Immediately to the north lies the allocated open space area SC1, which is intended to accompany development of this site and site SH2 (no application received to date) and which also currently comprises open fields. To the south lies the B1635, (Westlands) from which the access will be taken and the buildings associated with White House Farm (planning permission granted in outline for 25 dwellings), and to the east is existing residential development served by Hebron Road.
- 1.3 The application site consists of arable agricultural fields with hedgerows containing a number of trees along the northern and eastern boundaries. The applicants propose to retain the existing boundary vegetation apart from where the new access will be created from the B1635. Levels across the site vary, though the land rises very gently northwards from the B1635 and dips along the north-west boundary. The site is generally open in character.
- 1.4 The current proposal shows a mix of detached, semi-detached, and terraced houses ranging from two, to two and a half storeys in height. The size of the proposed dwellings is between two and four bedrooms. The applicant also proposes that some bungalows would be provided as part of the scheme.
- 1.5 The applicant also proposes a smaller area of open space (0.965 hectares) within the application site, adjacent to the eastern boundary.
- 1.6 The application site comprises an area of 6.9 hectares and a total of 183 dwellings are proposed. The developable area is 5.3 hectares and this would result in an average net density of 34.3 dwellings per hectare. The applicant has committed to 32% affordable dwellings on site (59), of which they indicate that 29% would be 3 bedroomed, 53% would be 2 bedroomed, 7% would be 1 bedroomed apartments and 11% would be 2 bedroomed bungalows.
- 1.7 As a whole the development will provide 9 no. 2 bed bungalows, 4 no. 1 bed dwellings 34 no. 2 bed dwellings (including the bungalows) and 51 no. 3 bed dwellings. 73 no 4 bed dwellings and 16no 5 bed dwellings.
- 1.8 The main vehicular access was previously approved through the outline permission and gains access into the site off Westlands (B1635), via a new roundabout which is proposed to be formed at the eastern end of the southern boundary. An emergency vehicle

/pedestrian and cycle access is proposed mid-way along the eastern boundary, linking in with Hebron Road. A further pedestrian /cycle access is also proposed linking with Leven Road and crossing the proposed public open space area adjacent to the eastern boundary. Both of these proposed new links would provide easy access for residents of the Hebron Road area to access the new area of public open space to the north, without having to walk along the B1635.

1.8 The application is supported by documents including: a Planning Compliance Statement; Design & Access Statement; Energy Assessment, Flood Risk Assessment; Arboricultural Reports; Ecological Report; Statement of Community Involvement; Site Plan; Site Location Plan, Layout plan along with elevations and other details.

2.0 RELEVANT PLANNING & ENFORCEMENT HISTORY

- 2.1 14/00714/OUT- Outline application for housing development (Use Class C3) of up to 183 dwellings, landscaping, means of access and associated infrastructure works, all matters reserved except for access.
- 2.2 12/00035/OUT Application refused and Appeal Upheld for Outline application for the construction of up to 213 dwellings, employment use (class B1) up to 2,900 sqm including means of access.
- 2.3 11/01300/OUT Outline application approved for the construction of up to 213 dwellings, employment use (class B1) up to 2,900 sq. m including means of access. The housing part of the scheme was refused and dismissed on appeal, on the grounds of insufficient affordable housing and lack of provision of public open space, but the employment proposals were allowed in May 2013.
- 2.4 02/01524/OUT Application refused for outline application for the construction of a care home with day centre facilities and 36 apartments for the elderly.

3.0 NATIONAL AND LOCAL POLICY:

3.1 The relevant policy of the Development Plan and any supplementary planning policy advice are as follows;

Core Strategy Policy CP1 - Sustainable development

Core Strategy Policy CP2 - Access

Core Strategy Policy CP4 - Settlement hierarchy

Core Strategy Policy CP5 - The scale of new housing

Core Strategy Policy CP5A - The scale of new housing be sub-area

Core Strategy Policy CP6 - Distribution of housing

Core Strategy Policy CP8 - Type, size and tenure of housing

Core Strategy Policy CP9 - Affordable housing

Core Strategy Policy CP16 - Protecting and enhancing natural and man-made assets

Core Strategy Policy CP17 - Promoting high quality design

Core Strategy Policy CP18 - Prudent use of natural resources

Core Strategy Policy CP19 - Recreational facilities and amenity open space

Core Strategy Policy CP20 - Design and the reduction of crime

Core Strategy Policy CP21 - Safe response to natural and other forces

Development Policies DP1 - Protecting amenity

Development Policies DP2 - Securing developer contributions

Development Policies DP3 - Site accessibility

Development Policies DP4 - Access for all

Development Policies DP6 - Utilities and infrastructure

Development Policies DP8 - Development Limits

Development Policies DP10 - Form and character of settlements

Development Policies DP13 - Achieving and maintaining the right mix of housing

Development Policies DP15 - Promoting and maintaining affordable housing

Development Policies DP29 - Archaeology

Development Policies DP30 - Protecting the character and appearance of the countryside

Development Policies DP31 - Protecting natural resources: biodiversity/nature conservation

Development Policies DP32 - General design

Development Policies DP33 - Landscaping

Development Policies DP34 - Sustainable energy

Development Policies DP36 - Waste

Development Policies DP37 - Open space, sport and recreation

Development Policies DP39 - Recreational links

Development Policies DP43 - Flooding and floodplains

Affordable Housing - Supplementary Planning Guidance - June 2008

Supplementary Planning Document - Sustainable Development - Adopted 22 September 2009

Allocations Document Policy SH1 - White House Farm and Crab Tree Farm,

Stokesley - adopted 21 December 2010

National Planning Policy Framework

4.0 CONSULTATIONS AND REPRESENTATIONS

Consultation Responses

- 4.1 Environmental Health No objections.
- 4.2 Highway Authority Further information was requested from the developer. Following receipt of this information the Highway Authority has no objections to the scheme.
- 4.3 Natural England No observations.
- 4.4 Police No objections, subject to conditions re security during construction.
- 4.5 Northern Gas Networks No objections, but developer should note the location of Northern Gas Network's infrastructure during construction.
- 4.6 Environment Agency No objections, subject to comments and conditions on the outline permission.
- 4.7 Northumbrian Water We can inform you that a sewerage rising main crosses the site and may be affected by the proposed development. Northumbrian Water do not permit a building over or close to our apparatus and therefore we will be contacting the developer direct to establish the exact location of our assets and ensure any necessary diversion, relocation or protection measures required prior to the commencement of the development. We will be contacting the developer/agent directly in this matter.

Representations

- 4.8 Site Notice 07 September 2015 expires 28 September 2015.
- 4.9 Newspaper Advertisement 18 August 2015 expires 12 October 2015.
- 4.10 One objection received expressing concerns about the principle of development in terms of the impact on ecology.
- 5.0 OBSERVATIONS

- 5.1 The main issues to consider in the determination of this application are matters relating to:
- a) The principle of the development
- b) Housing mix, type & tenure
- c) Affordable housing
- d) Public open space
- e) Education
- f) Highways issues
- g) Drainage & flood risk
- h) Design & layout
- i) Sustainable construction
- j) Landscape & visual impact
- k) Trees & ecology
- I) Archaeology
- m) Ground conditions
- n) Impact on residential amenity
- o) Other developer contributions
- p) Community engagement

The Principle of the Development

- 5.2 The application site is allocated for development in the LDF Allocations Document, adopted December 2010. This site was chosen for development in preference to other potential sites because it -
 - Respected the existing settlement form of Stokesley
 - Provided the most sustainable development option on available land
 - is located near to existing public transport routes, close to local shops and services
 - Is accessible from existing transport routes and are within reasonable walking / cycling distance of Stokesley Town Centre
 - Has satisfactory access to the road network with opportunities for pedestrian links to the town centre; and
 - Is available in the short to medium term.
- 5.3 Planning permission has been granted in outline with all matters reserved except for access which was approved through the outline permission.
- 5.4 The principle of the development of this site is established by the allocation and through the granting of the outline permission.
- 5.5 For reasons relating to flood risk, the net area proposed for housing development on this site was reduced to 5.33 hectares. This means that the number of dwellings overall has had to be reduced from 213 to 179 dwellings. The density of development proposed is now approximately 34 dwellings per hectare, which is slightly below the 35 dwellings per hectare required by Policy SH1, but was considered close enough to be acceptable at the outline stage.
- 5.6 The outline approval provides vehicle access into the site from Westlands as required by Policy SH1.

Affordable Housing and Mix, Type & Tenure

5.7 Policy CP8 of the adopted Core Strategy states that "Proposals for housing must take appropriate account of local housing needs in terms of size, type and tenure of dwellings. These needs will include appropriate provision for all sectors of the community, for example including the needs of elderly people."

- 5.8 The applicant's Planning Statement indicates that a mix of 2 5 bed dwellings (market housing) would be provided. The applicant has included some elderly persons' accommodation (e.g. bungalows) as part of this scheme.
- 5.9 Within the outline permission the applicant offered to provide 32% affordable housing on site. Whilst this did not meet the Policy CP9 50% target, this was justified for a number of reasons which impacted on viability including a reduction in site area due to flood risks and carrying out mitigation measures including raising floor levels of the dwellings, which will add to the costs of the scheme. This has been taken into account in considering the viability of the scheme overall and the 32% affordable housing provision was considered reasonable in this case.
- 5.10 A Section 106 Agreement has been entered into which includes the provision of 32% affordable housing, distributed throughout the development, with tenure mix and types in accordance with the Housing Manager's requirements.
- 5.11 In terms of mix the applicant's planning statement sets out the following house types and sizes which includes 9 no. 2 bed bungalows (5%), 4 no. 1 bed dwellings, 31 no. 2 bed dwellings and 51 no. 3 bed dwellings. As such, 50% of the dwellings proposed as part of the development are smaller homes suitable for individuals, couples and smaller families. The remainder of the proposed homes will be 4 and 5 bed. The applicant considers that whilst the proposed mix of housing does not fully meet the recently adopted Size, Type and Tenure SPD the scheme is fully compliant with the aims of the NPPF to create a sustainable, inclusive and mixed community. The level of bungalow provision falls below the target set by the adopted housing type and tenure SPD (10%) but given the market need in this area along with the fact that the viability of this site was set at the outline stage when the SPD was not an adopted policy of the authority, the proposed mix of dwellings set out at paragraphs 1.6 and 1.7 are considered to be reasonably acceptable in this case.

Public Open Space

- 5.12 Policy DP37 of the adopted Development Policies DPD requires new housing developments to contribute towards the achievement of the local standards by reducing or preventing both quantitative and qualitative deficiencies in recreation provision related to the development.
- 5.13 Section 4 of the Open Space, Sport & Recreation Supplementary Planning Document (Open Space SPD) identifies that developments of between 80 and 300 dwellings should make provision for amenity green space, public parks, play areas and facilities for teenagers on site. In addition, off-site contributions will normally also be sought to ensure Policy DP37 Standards are met, because the Council's evidence base indicates significant shortfalls in the amount of amenity space in all sub areas (see paragraph 4.12 of the Open Space SPD).
- 5.14 The applicant has confirmed that they would provide and equip site SC1, to the north of their development in association with the developer of the adjacent SH2 site, as required by the Allocations DPD. A smaller area of Public Open Space is also to be provided adjacent to the eastern boundary within part of the area which is now within Flood Zone 2. A commuted sum for off-site outdoor sports facilities was included within the S 106 agreement, together with a sum for maintenance costs.

Education

5.16 Policy DP2 of the adopted Development Policies DPD stipulates that contributions will be sought where necessary to ensure the achievement of sustainable development, including the provision of additional infrastructure whenever there is a need generated by the

new development. This includes, amongst other things, provision of additional children's services/facilities where existing services in the area have insufficient capacity to cater for the potential increase in the number of children, or are appropriately placed to serve a development, having regard to the need to minimise travel, consistent with Core Policy CP2.

- 5.17 In addition, Policy DP5 of the Development Policies DPD advises that support will be given to the provision and enhancement of community facilities with a view to maintaining sustainable communities. Policy DP6 on utilities and infrastructure seeks to ensure new development is capable of being accommodated by existing or planned services
- 5.18 NYCC Children and Young Peoples Service confirmed at outline stage that in this instance a contribution towards schools in Stokesley would not be required.

Highways Issues

- 5.19 At outline stage a Transport Assessment (TA) and a Travel Plan was submitted with the application. The TA examined the transport related impacts of the proposed development. Access to the site by all modes of transport has been fully considered with both positive and negative impacts identified. The TA concludes that the additional generated traffic can be readily accommodated on the local road network such that no mitigation measures for capacity or safety reasons are required.
- 5.20 A new roundabout was agreed under the outline permission. The current layout of the site has been agreed with the Highway Authority and as such the scheme is considered to be acceptable in these terms.
- 5.21 Cycle routes were proposed through to the adjacent housing via Hebron Road. However, following consultation with the local community and following a detailed assessment of this route the connection has been omitted. Other through routes are available and the scheme is considered to be suitably and sustainably connected in these terms.

Flood Risk & Drainage

- 5.22 Policy DP43 of adopted the Development Policies DPD outlines the Council's approach to development and flooding and states that development will only be permitted if it has an acceptably low risk of being affected by flooding assessed against the Environment Agency's flood zone maps, other local information and where all necessary mitigation measures on or off site are provided.
- 5.23 Policy DP6 of the adopted Development Policies DPD stipulates that new developments must be capable of being accommodated by existing or planned services, and must not have a seriously harmful impact on existing systems, worsening the services enjoyed by the community. These systems include surface water drainage and sewage disposal.
- 5.24 The layout of the scheme as submitted takes into account detailed discussions with the Environment Agency with regard to water flows across the site. The Environment Agency has raised no objections to the scheme as now submitted.
- 5.25 An attenuation tank is included as part of the wider development of the site. This is located outside the original red line boundary and has now been approved through submission of a separate application for planning permission. This separate application also included a foul water pumping station to facilitate sewerage for the site.
- 5.26 Whilst the site may be at some risk of flooding, the risk is relatively small as the scenario that is being planned for is a 1 in 1000 year event as opposed to a 1 in 100 year

event which is more commonly planned for by Flood Risk Assessments. Both the Environment Agency and the Council's Senior Drainage Engineer are unaware of any previous flooding at the application site and they are satisfied that the measures proposed in the scheme would deal adequately with any future flooding issues.

Design & Layout

- 5.27 Policy DP32 states that the design of all developments must be of the highest quality. Attention to the design quality of all development is essential. Development proposals must seek to achieve creative, innovative and sustainable designs that take into account local character and settings, and promote local identity and distinctiveness.
- 5.28 This approach has been strengthened by paragraph 56 of the National Planning Policy Framework (NPPF) which states that "The Government attaches great importance to the design of the built environment. Good design is a key aspect of sustainable development, is indivisible from good planning, and should contribute positively to making places better for people." The NPPF also emphasises, in paragraph 66, the importance of public engagement in evolving good design.
- 5.29 The proposed scheme is considered to respond well to the site constraints and the design and form of this part of Stokesley. The layout creates a pleasing and relatively open development form. On submission of the scheme the area of housing located in the north west quadrant had little open space and a generally constrained development form. This has been relaxed by extending the development form to create a more permeable western boundary to the site. As well as improving the private and public realm within the site this contributes to a more permeable development form with views into and out of the site, whilst creating a strong landscape buffer utilising hedgerows and mixed standard trees along the western boundary.

Sustainable Construction

- 5.30 Policy DP34 of the LDF requires all developments of 10 or more residential units to address sustainable energy issues, by reference to accredited assessment schemes and incorporate energy efficient measures which will provide at least 10% of their on-site renewable energy generation, or otherwise demonstrate similar energy savings through design measures.
- 5.31 The applicant has submitted a detailed Energy Assessment which sets out the energy performance of the development and the measures proposed to improve the performance of the scheme in order to meet the requirements of the policy.
- 5.32 The Energy Assessment sets out the baseline energy emissions in accordance with the Standard Assessment Procedure, for a development which would be compliant under Part L of the Building Regulations. The Assessment then proposes increased specifications in a number of areas of the development including wall insulation, roof insulation, glazing and doors. These energy efficiency measures are then complimented with the installation of 192 solar photovoltaic panels across the site in order to meet the 10% energy emissions requirement.

Landscape & Visual Impact

5.33 Policy DP30 of the adopted Development Policies DPD seeks to protect the character and appearance of the countryside. The design and location of new development should take account of landscape character and its surroundings, and not have a detrimental effect on the immediate environment and on any important long distance views. The landscape and visual impact of the development in the area was considered at the time that the site was allocated for residential development and through the determination of the

outline permission. It is considered that the site would not have a significant impact upon any important landscape features or character areas.

- The application notes that whilst there would be a loss of agricultural land as a result of the development the applicants intend to keep most of the existing landscape features such as the hedgerows and trees along the boundaries. The LDF Allocations Document sought a new woodland buffer to be created along the western boundary which would be designed to soften the appearance of the site when approaching along the B1365 from the west. The applicant has taken this on board but it was considered that a straight line of trees as suggested by the allocation would be an alien feature in this landscape on the edge of Stokesley. As such a more permeable approach has been taken in order to meet the requirements of the allocation whilst creating a suitable transition between the urban fringe and the open countryside beyond. The density of development along the western boundary of the site is low, allowing for significant levels of boundary planting in this area. Due to the significant reduction in the developable area of the site due to flood risk and the previously agreed levels of affordable housing the extension of the built area of the site into what would otherwise be landscape buffer was considered to be necessary to allow the scheme to progress. The layout and planting as now proposed will achieve the aims of the allocation, whilst facilitating a design and form of development that will meet the requirements of adopted policy.
- 5.35 The layout of the site, relationship of houses to one another and the size and form of private amenity space created by the scheme are considered to be appropriate.

Trees & Ecology

- 5.36 Policy DP31 of the adopted Development Policies DPD states that 'Permission will not be granted for development which would cause significant harm to sites and habitats of nature conservation...Support will be given...to the enhancement and increase in number of sites and habitats of nature conservation value'. In response to the requirements of this policy, the applicants in submitting the outline application submitted an Arboricultural Assessment, and an Ecological Appraisal. A survey and assessment of existing trees was carried out in accordance with British Standard 5837:2012 'Trees in Relation to Design, Demolition and Construction Recommendations' (BS5837) and a tree protection scheme has been submitted with this application.
- 5.37 The applicant proposes the retention of the majority of boundary trees, integrating them into the landscape buffer margins at the edges of the site. Overall, the Arboricultural report concludes that the proposed new development would retain the majority of trees on the site and that these should be adequately protected during construction works.
- 5.38 The Ecological Appraisal submitted at outline stage concludes that no habitats listed within the UK Biodiversity Action Plans are represented on site. No evidence was found of badgers, otters water voles or protected bird species.

Archaeology

- 5.39 Paragraph 128 of the NPPF states that "Where a site on which development is proposed includes or has the potential to include heritage assets with archaeological interest, local planning authorities should require developers to submit an appropriate Desk-Based Assessment and, where necessary, a field evaluation."
- 5.40 A written scheme of investigation required by conditions attached to the outline permission has been submitted and approved.

Ground Conditions

5.41 The applicant submitted a desk study Ground Conditions Report which concluded that significant contamination is unlikely to be present and that any made ground encountered will be isolated and relatively shallow.

Impact on Residential Amenity

5.42 There are a number of residential properties which back onto the application site. The proposed development has been laid out to minimise the impact on these properties. The orientation and separation distances of the proposed dwellings, is considered to satisfactorily protect the residential amenity of the existing and proposed properties.

Other Developer Contribution Requirements

- 5.43 Through the granting of the outline consent agreement was established for footpath and cycle-way contributions for links between Stokesley and Great Ayton. The applicant agreed to pay £54,964, towards the cost of the first phase of the proposed cycle route and this is covered by the section 106 Agreement.
- 5.44 Policy SH1 requires a contribution towards the upgrading of the potable water network if necessary to enable a suitable supply to be made available to the new development. The applicants have allowed for this in their viability assessment. However, Northumbrian Water has not confirmed whether or not this will be necessary. In the event that it is not required, the funding for this should be put towards additional on-site affordable housing provision. These matters are covered by the Section 106 Agreement.
- 5.45 It should be noted that this development is not liable under the Community Infrastructure Levy as the scheme was approved in outline prior to the adoption of CIL.

Community Engagement

- 5.46 Public consultation by developers of large sites at the pre-application stage should be a genuinely meaningful exercise and must be guided by the Council's Statement of Community Involvement and paragraph 66 of the NPPF.
- 5.47 Paragraph 66 of the NPPF sets an expectation that developers should work closely with those affected by their proposals to evolve designs that take account of the views of the community. This is reflected in the Council's Statement of Community Involvement (SCI), which requires that communities are offered genuine choice and a real opportunity to influence proposals in consultation exercises. The NPPF states that proposals that can demonstrate this in developing the design of the new development should be looked on more favourably.
- 5.48 The Council's SCI makes clear that developers should discuss and agree the exact nature of consultation in advance. In this case the applicant's agent did consult with the Council on the content and nature of the pre-application consultation. The applicant has undertaken community consultation by way of a leaflet shot to the local community (approx 1000 homes), an open day consultation event and through the Taylor Wimpey website. The public consultation carried out is considered to be commensurate with the scale of the proposed development.

CONCLUSION

5.49 The number of dwellings now proposed is lower than anticipated in both the allocation and later outline approval, primarily due to a portion of the site being within Flood Zone 2 and therefore incapable of accommodating housing. The affordable housing proposed (and agreed in the outline permission) is lower than 50% but given the viability assessment at outline stage the provision of 32% affordable housing of the tenure and mix

as detailed is considered to be acceptable in this case.

- 5.50 The design and layout of the scheme is considered to respond well to the site constraints and the character and form of this part of Stokesley.
- 5.51 The Local Planning Authority has acted positively and proactively in determining this application by assessing the proposal against all material considerations, including planning policies and any representations that may have been received and subsequently determining to grant planning permission in accordance with the presumption in favour of sustainable development, as set out within the National Planning Policy Framework.
- 5.52 Approval is recommended.

6.0 RECOMMENDATION:

- 6.1 That subject to any outstanding consultations and the requirements of the signed S106 agreement the application be **Granted.**
 - 1. The development hereby permitted shall be begun within two years of the date of this permission.
 - The permission hereby granted shall not be undertaken other than in complete accordance with the following drawings received by Hambleton District Council unless otherwise approved in writing by the Local Planning Location Plan 20320:LP received 25 August 2015, Authority. Planning Layout 20320:00 received 27 October 2015, Boundary Treatment Layout 20320:02 received 27 October 2015, QUAD/5/PL1 received 25 August 2015, Detailed Landscape Proposals. 2578/1 + 2578/2 + 2578/3 + 2578/4 Received 25 Augsut 2015, Boundary Treatment details. 20320:03 received 25 August 2015, Single Garage. 20320:SG received 25 Augsut 2015, Triple Garage 20320:TG received 25 August 2015, Double Garage. 20320:DG received 25 August 2015, Substation. GTC-E-SS-0012_R1-7 received 25 August 2015 House Plans: Gosford. PA34/5/PL1 received 25 August 2015, Midford. PA44/5/PL1 received 25 August 2015, Shelford. PA48/5/PL1 received 25 August 2015, Thornford. PA49/5/PL1 received 25 August 2015, Cotterdale. ZA25/5/PL1 received 25 August 2015, Bainbridge. Wilton. PB52/5/PL1. received 25 August 2015, Bradenham. PD48/5/PL1 received 25 August 2015, Downham. PD49/5/PL1 received 25 August 2015, Lavenham. PD51/5/PL1 received 25 August 2015, Eynsham. PD410/4/PL1 received 25 August 2015, Haddenham. PD411/5/PL1 received 25 August 2015, Kirby. PrB/5/PL1 received 25 August 2015, Langdale. PT43/5/PL1 received 25 August 2015, Stokesley 762. 762/5/PL1 received 25 August 2015, Bellerby. ZA34/5/PL1 received 25 August 2015, Milton ZT310/5/PL1 received 25 August 2015. Elevations: Midford PA44/5/PL2 received 25 August 2015, Shelford PA48/5/PL2 received 25 August 2015, Thornford PA49/5/PL2 received 25 August 2015, Wilton PB52/5/PL2 received 25 August 2015, Bradenham PD48/5/PL2 received 25 August 2015, Downham PD49/5/PL2 received 25 August 2015, Lavenham PD51/5/PL2 received 25 August 2015, Eynsham PD410/4/PL2 received 25 August 2015, Haddenahm PD411/5/PL2 received 25 August 2015, Kirkby PRB/5/PL2 received 25 August 2015, Langdale PT43/5/PL2 received 25 August 2015, Stokesley 762/5/PL2 received 25 August 2015, Gosford PA34/5/PL2 received 25 August 2015, Cotterdale ZA25/5/PL2 received 25 August 2015, Bellerby ZA34/5/PL2 received 25 August 2015, Bainbridge QUAD/5/PL2 received 25 August 2015, Miton ZT310/5/PL2 received 25 August 2015.

- 3. No dwelling shall be occupied until its associated boundary walls, fences, hedgerows and other means of enclosure associated with it have been constructed in accordance with the details approved in accordance with condition 4 above. All boundary walls, fences, hedgerows and other means of enclosure shall be retained and no part thereof shall be removed without the prior written consent of the Local Planning Authority.
- 4. Prior to development commencing detailed cross sections shall be submitted to and approved in writing by the Local Planning Authority, showing the existing ground levels in relation to the proposed ground and finished floor levels for the development. The levels shall relate to a fixed Ordnance Datum. The development shall be constructed in accordance with the approved details and thereafter be retained in the approved form.
- 5. Before each phase of development approved by this planning permission, no development shall take place until such time as a scheme to mitigate against surface water run-off has been submitted to, and approved in writing by, the local planning authority. This scheme should include: a) The design for a detailed surface water drainage scheme. b) Calculations for the site's current green field run-off rate. This rate should be applied to the impermeable area of the proposed development so sufficient surface water storage capacity can be identified. The scheme shall be fully implemented and subsequently maintained, in accordance with the timing / phasing arrangements embodied within the scheme, or within any other period as may subsequently be agreed, in writing, by the local planning authority.
- 6. Development shall not be occupied until a detailed scheme for the disposal of foul water from the development hereby approved has been submitted to and approved in writing by the Local Planning Authority in consultation with Northumbrian Water. Thereafter the development shall take place in accordance with the approved details.
- 7. The site shall be developed with separate systems of drainage for foul and surface water on and off site.
- 8. Unless otherwise approved in writing by the local planning authority, there shall be no piped discharge of surface water from the development prior to the completion of the approved surface water drainage works and no buildings shall be occupied or brought into use prior to completion of the approved foul drainage works.
- 9. No development shall be commenced until an assessment of the risks posed by contamination, carried out in line with the Environment Agency's Model Procedures for the Management of Land Contamination CLR11, has been submitted to and approved by the local planning authority. A scheme for the remediation of any contamination shall be submitted and approved by the Local Planning Authority before any development occurs. The development shall not be occupied until the approved remediation scheme has been implemented and a verification report detailing all works carried out has been submitted to and approved in writing by the Local Planning Authority.
- 10. Unless otherwise approved in writing by the Local Planning Authority, there shall be no excavation or other groundworks, except for investigative works or the depositing of material on the site, until the following drawings and details have been submitted to and approved in writing by the Local Planning Authority (1) Detailed engineering drawings to a scale of not less than 1:500 and based upon an accurate survey showing: (a)

proposed highway layout including adoptable turning heads including the highway boundary (b) dimensions of any carriageway, cycleway, footway, and verges (d) the proposed buildings and site (c) visibility splays layout, including levels (e) accesses and driveways (f) drainage and sewerage system (g) lining and signing (h) traffic calming measures (i) all types of surfacing (including tactiles), kerbing and edging. Longitudinal sections to a scale of not less than 1:500 horizontal and not less than 1:50 vertical along the centre line of each proposed road showing: the existing ground level (b) the proposed road channel and centre line levels (c) full details of surface water drainage proposals. (3) Full highway construction details including: (a) typical highway cross-sections to scale of not less than 1:50 showing a specification for all the types of construction proposed for carriageways, cycleways and footways/footpaths

- (b) when requested cross sections at regular intervals along the proposed roads showing the existing and proposed ground levels (c) kerb and edging construction details (d) typical drainage construction details.
- (4) Details of the method and means of surface water disposal. (5) Details of all proposed street lighting. (6) Drawings for the proposed new roads and footways/footpaths giving all relevant dimensions for their setting out including reference dimensions to existing features. (7) Full working drawings for any structures which affect or form part of the highway network. (8) A programme for completing the works. The development shall only be carried out in full compliance with the approved drawings and details unless agreed otherwise in writing by the Local Planning Authority
- 11. No dwelling to which this planning permission relates shall be occupied until the carriageway and any footway/footpath from which it gains access is constructed to basecourse macadam level and/or block paved and kerbed and connected to the existing highway network with street lighting installed and in operation. The completion of all road works, including any phasing, shall be in accordance with a programme approved in writing with the Local Planning Authority before the first dwelling of the development is occupied.
- 12. Unless otherwise approved in writing by the Local Planning Authority, there shall be no excavation or other groundworks, except for investigative works, or the depositing of material on the site in connection with the construction of the access road or building(s) or other works until the details of the provision of tactile paving and the pedestrian crossing point on Romanby Road have been submitted to and approved in writing by the Local Planning Authority.
- 13. No dwelling shall be occupied until the related parking facilities have been constructed in accordance with the approved drawing Once created these parking areas shall be maintained clear of any obstruction and retained for their intended purpose at all times.
- 14. Notwithstanding the provisions of the Town and Country Planning General Permitted Development Order 1995 or any subsequent Order, the garage(s) shall not be converted into domestic accommodation without the granting of an appropriate planning permission.
- 15. There shall be no access or egress by any vehicles between the highway and the application site until details of the precautions to be taken to prevent the deposit of mud, grit and dirt on public highways by vehicles travelling to and from the site have been submitted to and approved in writing

by the Local Planning Authority in consultation with the Highway Authority. These facilities shall include the provision of wheel washing facilities where considered necessary by the Local Planning Authority. These precautions shall be made available before any excavation or depositing of material in connection with the construction commences on the site and be kept available and in full working order and used until such time as the Local Planning Authority agrees in writing to their withdrawal.

- 16. Unless approved otherwise in writing by the Local Planning Authority there shall be no establishment of a site compound, site clearance, demolition, excavation or depositing of material in connection with the construction on the site until proposals have been submitted to and approved in writing by the Local Planning Authority for the provision of: (i) on-site parking capable of accommodating all staff and sub-contractors vehicles clear of the public highway; and (ii) on-site materials storage area capable of accommodating all materials required for the operation of the site. The approved areas shall be kept available for their intended use at all times that construction works are in operation.
- 17. The development shall be built in accordance with the energy reduction and renewable energy measures as detailed within the submitted Energy Assessment received on 26 August 2015.
- 18. Prior to any development taking place above foundation level, details that show how 'Secured by Design' principles have been incorporated into the scheme shall be submitted for the written approval of the Local Planning Authority and once approved the development shall be implemented in accordance with the approved 'Secured by Design' details prior to occupation or use of any part of the development hereby approved.
- 19. Prior to the commencement of work above foundation level, a scheme of hard and soft landscaping works shall be submitted to and approved in writing by the Local Planning Authority. The submitted scheme shall provide details of the species, numbers and locations of planting, all hard surface materials, timescales for implementation and a maintenance schedule. The approved landscaping scheme shall be implemented and maintained thereafter in accordance with the approved details.
- 20. No development shall take place above foundation level until a detailed scheme for the implementation of the mitigation measures outlined in the ecological report prepared by E3 Ecology Limited, has been submitted to and agreed in writing with the Local Planning Authority. The development shall be implemented in accordance with the agreed scheme and programme for implementation.
- 21. The development shall not be commenced until a plan has been submitted to and approved in writing by the Local Planning Authority, showing all existing trees and hedges which are to be felled or retained, together with the positions and height of protective fences, and the areas for the storage of materials and the stationing of machines and huts.
- 22. Prior to commencement of work a Construction Management Plan including details of hours of operation and delivery times, methods of controlling noise and dust, details of lorry routes to and from the site and site security measures during the construction period, shall be submitted to and approved in writing by the Local Planning Authority. The agreed scheme shall be implemented in accordance with the approved details.

23. No external lighting shall be installed other than in complete accordance with a scheme that has previously been approved in writing by the Local Planning Authority.

The reasons for the above conditions are:-

- 1. To ensure compliance with Section 92 of the Town and Country Planning Act, 1990.
- 2. In order that the development is undertaken in a form that is appropriate to the character and appearance of its surroundings and in accordance with the Development Plan Policy(ies) .
- 3. To protect the amenity of occupiers and neighbouring residents and to ensure that the development is appropriate to the character and appearance of its surroundings in accordance with Policies CP1, DP1, CP17 and DP32 of the Hambleton Local Development Framework.
- 4. To protect the amenity of the neighbouring residents and to ensure that the development is appropriate to the character and appearance of its surroundings in accordance with Policies CP1, DP1, CP17 and DP32 of the Hambleton Local Development Framework.
- 5. To prevent flooding by ensuring the satisfactory storage of/disposal of surface water from the site and to reduce the risk of flooding to the proposed development and future users.
- 6. To prevent the increased risk of flooding from any sources in accordance with the NPPF.
- 7. To ensure that the development can be properly drained.
- 8. To ensure that no foul or surface water discharges take place until proper provision has been made for their disposal.
- 9. In order to take proper account of the risks to the health and safety of the local population, builders and the environment and address these risks and in accordance with the Hambleton Local Development Framework Policy CP21.
- 10. To ensure safe and appropriate access and egress to the dwellings, in the interests of highway safety and the convenience of prospective residents.
- 11. To ensure safe and appropriate access and egress to the dwellings, in the interests of highway safety and the convenience of prospective residents.
- 12. To ensure that the details are satisfactory in the interests of the safety and convenience of highway users.
- 13. To provide for adequate and satisfactory provision of off-street accommodation for vehicles in the interest of safety and the general amenity of the development.
- 14. To ensure the retention of adequate and satisfactory provision of offstreet accommodation for vehicles generated by occupiers of the dwelling and visitors to it, in the interest of safety and the general amenity the development.

- 15. To ensure that no mud or other debris is deposited on the carriageway in the interests of highway safety.
- 16. To provide for appropriate on-site vehicle parking and storage facilities, in the interests of highway safety and the general amenity of the area.
- 17. In order to minimise energy demand, improve energy efficiency and promote energy generated from renewable resources in accordance with policy DP34 of the Hambleton Local Development Framework.
- 18. In order to ensure that the development takes account of the need to reduce opportunities for crime and fear of crime, in accordance with Policy CP20 of the adopted Local Development Framework.
- 19. In order to soften the visual appearance of the development and provide any appropriate screening to adjoining properties in accordance with Policies CP1, DP1, CP17 and DP32 of the Hambleton Local Development Framework.
- 20. To ensure that proper regard is had to the mitigation of the proposed development on existing wild life species and habitat, in accordance with Policies CP16 and DP31 of the adopted Hambleton Local Development Framework.
- 21. To ensure that existing trees which are of amenity value are retained and adequately protected during the construction period, in accordance with Policies CP16 and DP31 of the adopted Hambleton Local Development Framework.
- 22. To protect the amenity of adjacent residents and to accord with Policies CP1 and DP1 of the Hambleton Local Development Framework.
- 23. In order that the Local Planning Authority can consider the impact of the proposed lighting scheme and avoid environmental pollution in accordance with Policy DP32 of the adopted Hambleton Local Development Framework.



Parish: East HarlseyCommittee Date :12 November 2015Ward: Osmotherley & SwainbyOfficer dealing :Mrs H M Laws

Target Date: 17 November 2015

15/02139/FUL

Construction of a detached dwellinghouse with domestic garage. at Bankside Farm East Harlsey North Yorkshire DL6 2DB for Mr & Mrs John Newcombe.

- 1.0 SITE DESCRIPTION AND PROPOSAL
- 1.1 The site, which is currently an agricultural field, lies at the eastern end of the village of East Harlsey on the northern side of the village street. The site covers an area of approximately 0.1 hectares with a frontage onto the village street of 28m. The land is positioned between the existing dwelling at Bankside Farm and the detached dwelling to the west at 8 The Beeches. A strip of land adjacent to 8 The Beeches is excluded from the application site boundary.
- 1.2 The application is bounded at the front by a mature well established hedgerow along its entire length. The hedge lies at the top of a steep verge, up to 1400mm higher than carriageway level. The application site slopes gently downwards on the other side of the hedgerow and is approximately 3.6m lower than the base of the hedge at the rear (northernmost part) of the site. The boundary with the garden of Bankside Farm is a post and rail fence with hedging. There are no boundaries to the west and north of the application site. A mature sycamore lies on the boundary but is within the garden of 8 The Beeches.
- 1.3 It is proposed to construct a two storey detached dwelling on the site. The dwelling would be built at a similar ground level to the dwelling at 8 The Beeches and higher than the dwelling at Bankside Farm. The four bedroomed, double fronted dwelling would have a central oak framed porch feature on the front elevation and an attached single storey section accommodating a garage and a utility room. Rooflights are proposed in the rear elevation although there is no accommodation proposed within the roofspace. A glazed balustrade feature is proposed at first floor on the rear elevation serving the master bedroom. The dwelling would be finished in brickwork and slates. It is proposed to install pv panels on the front elevation.
- 1.4 It is proposed to remove a 5m section of hedge at the eastern end of the site's frontage to create an access into the site. The dwelling would be set back at a distance of almost 15m from the front boundary hedge. The front boundary hedge is otherwise proposed to be retained.
- 1.5 The application submission also includes a tree survey and an ecological assessment.
- 2.0 PLANNING & ENFORCEMENT HISTORY
- 2.1 None
- 3.0 RELEVANT PLANNING POLICIES:
- 3.1 The relevant policy of the Development Plan and any supplementary planning policy advice are as follows;

Core Strategy Policy CP1 - Sustainable development

Core Strategy Policy CP2 - Access

Core Strategy Policy CP4 - Settlement hierarchy

Core Strategy Policy CP16 - Protecting and enhancing natural and man-made assets

Core Strategy Policy CP17 - Promoting high quality design

Core Strategy Policy CP21 - Safe response to natural and other forces

Development Policies DP1 - Protecting amenity

Development Policies DP4 - Access for all

Development Policies DP9 - Development outside Development Limits

Development Policies DP30 - Protecting the character and appearance of the countryside

Development Policies DP32 - General design

Development Policies DP43 - Flooding and floodplains

Interim Guidance Note - adopted by Council on 7th April 2015

National Planning Policy Framework - published 27 March 2012

4.0 CONSULTATIONS

- 4.1 Parish Council comments as follows:
- 1. No identified need for development of properties of such size in this location 2. Discussion with village residents, admittedly not scientific, suggests that opposition to development of agricultural land for housing is widespread.
- 3. This is agricultural land.
- 4. Not infill. Next door on one side is a farm.
- 5. This would establish principle that extending village boundary into agricultural land is acceptable to the PC.
- 6. Developing agricultural land would change the nature of the village.

Therefore refuse

Concern over the fact this is agricultural land and that this may open the gates for further development of agricultural land.

4.2 Parish Council Chairman - Whilst I share the concerns of others re development of agricultural land I think the application is for an appropriate and attractive dwelling that may well enhance that end of the village. The risk of opening the flood gates is low due to the fact that there are very few potential development sites within the village boundary. Approve.

This is on agricultural land not a brownfield site or a conversion of an existing building. Acceptance of this proposal would create a precedent for further development on agricultural land/green belt within the boundary of the village which the majority of residents are opposed to. Refuse.

Having read through the interim policy guidance (IPG) there is nothing in that document as far as I can understand which specifically says that a dwelling cannot be built on agricultural land. The criteria for East Harlsey is, any development must be small, within the village boundary and similar in character to existing properties. A downside might be that it impacts on the open character of the site but wouldn't that be the same for any development in a similar position. On that basis of the IPG I can only approve unless I have missed something really obvious.

- 4.3 NYCC Highways conditions recommended
- 4.4 Yorkshire Water comments not yet received (expiry date for representations 21/10/2015)
- 4.5 Site notice/local residents no comments received (expiry date for representations 5/11/2015)

5.0 OBSERVATIONS

5.1 The main issues for consideration in this case relate to the principle of a new dwelling in this location outside Development Limits, an assessment of the likely impact of the proposed dwelling on the character and appearance of the village in respect of its scale,

siting and design, the effect on the rural landscape, neighbour amenity and highway safety and developer contributions.

5.2 The site falls outside of Development Limits as East Harlsey does not feature within the settlement hierarchy defined within Policy CP4 of the Core Strategy. Policy DP9 states that development will only be granted for development "in exceptional circumstances". The applicant does not claim any of the exceptional circumstances identified in Policy CP4 and, as such, the proposal would be a departure from the development plan. However, it is also necessary to consider more recent national policy in the form of the NPPF. Paragraph 55 of the NPPF states:

"To promote sustainable development in rural areas, housing should be located where it will enhance or maintain the vitality of rural communities. For example, where there are groups of smaller settlements, development in one village may support services in a village nearby. Local planning authorities should avoid new isolated homes in the countryside unless there are special circumstances".

- 5.3 To ensure appropriate consistent interpretation of the NPPF alongside Policies CP4 and DP9, the Council has adopted Interim Policy Guidance (IPG) relating to Settlement Hierarchy and Housing Development in the Rural Areas. This guidance is intended to bridge the gap between CP4/DP9 and the NPPF and relates to residential development within villages. The IPG has brought in some changes and details how Hambleton District Council will now consider development in and around smaller settlements and has included an updated Settlement Hierarchy.
- 5.4 In the settlement hierarchy contained within the IPG, East Harlsey is defined as a 'secondary village'; within the IPG small scale development adjacent to the main built form of the settlement "will be supported where it results in incremental and organic growth". To satisfy criterion 1 of the IPG the proposed development must provide support to local services. The site lies on the edge of the village of East Harlsey which has facilities including a public house and village hall. Criterion 1 would be satisfied.
- 5.5 It is important to consider the likely impact of the proposed development with particular regard to criteria 2, 3 and 4 of the IPG. The proposed dwelling would be on undeveloped land that lies between existing residential properties on the northern side of the village street. There are also houses opposite on the southern side of the street. The following detailed advice within the IPG is considered to be relevant:
- "Proposals will be assessed for their impact on the form and character of a settlement. Consideration should be given to the built form of a settlement, its historical evolution and its logical future growth and how the proposal relates to this."
- "Any detrimental impact on the character, appearance and environmental quality of the surrounding area should be avoided and development should not compromise the open and rural character of the countryside."
- 5.6 The proposed development would infill an existing undeveloped part of the village, which is one of several gaps at this end of the village before it finally becomes truly open countryside. It could be argued that these gaps form part of East Harlsey's character and the infilling of the gaps would create ribbon development extending much further beyond the main part of the village. Additional built development also lies on the opposite site of the road and as such respects the general built form of the village. There would be no harmful impact to the natural, built and historic environment.
- 5.7 The proposed development is of a high standard of design that respects its immediate neighbours and is in keeping with the character and appearance of the village.
- 5.8 The closest neighbours are two storey dwellings to either side and two storey dwellings on the opposite side of the road. The distances between the proposed dwelling and the existing properties are relatively substantial as it is a low density development and would not

adversely affect the amenity of existing residents as a result of overshadowing or overlooking.

- 5.9 The Highway Authority has no objections regarding the proposed development. It is not considered that the proposed development would adversely impact highway safety and conditions are recommended.
- 5.10 An ecological assessment confirms that the site, as grassland, is of low ecological value although the hedgerows and nearby trees offers potential for species habitat. There is no adverse impact anticipated as a result of the development. It is recommended that the new boundaries are hedges, which would mitigate for the loss of the section of hedge removed to create the access along the front boundary.
- 5.11 A tree survey has also been submitted, which confirms that the dwelling as proposed would not have an adverse effect on any mature trees within the vicinity.
- 5.12 Under the Community Infrastructure Levy (CIL) regulations the proposed dwellings are liable for payment of CIL at a rate of £55 per sq. m, the rate adopted by the Council on 7 April 2015.
- 5.13 It is considered that the proposal is in accordance with the Interim Policy Guidance document in that the housing development, in respect of siting, scale, design, materials and access is acceptable and would have no adverse impact on landscape character, residential amenity and highway safety.
- 5.14 The proposed development is acceptable and approval of the application is recommended.

SUMMARY

The site is appropriate in location and size to accommodate residential development without harm to the amenities of neighbours or the surroundings and without harm to road safety. The proposed development complies with the requirements of the Interim Policy Guidance and otherwise conforms with the policy requirements set out in the adopted Hambleton Local Development Framework.

The Local Planning Authority has acted positively and proactively in determining this application by assessing the proposal against all material considerations, including planning policies and any representations that may have been received and subsequently determining to grant planning permission in accordance with the presumption in favour of sustainable development, as set out within the National Planning Policy Framework.

6.0 RECOMMENDATION: that subject to any outstanding consultations the application be GRANTED subject to the following conditions:

- 1. The development hereby permitted shall be begun within three years of the date of this permission.
- 2. Prior to development commencing, details and samples of the materials to be used in the construction of the external surfaces of the development shall be made available on the application site for inspection and the Local Planning Authority shall be advised that the materials are on site and the materials shall be approved in writing by the Local Planning Authority. The development shall be constructed of the approved materials in accordance with the approved method.
- 3. All new, repaired or replaced areas of hard surfacing shall be formed using porous materials or provision shall be made to direct run-off water from

the hard surface to an area that allows the water to drain away naturally within the curtilage of the property.

- 4. The development shall not be commenced until a detailed landscaping scheme indicating the type, height, species and location of all new trees and shrubs, has been submitted to and approved by the Local Planning Authority. No dwelling shall be occupied after the end of the first planting and seeding seasons following the approval of the landscaping scheme, unless those elements of the approved scheme situate within the curtilage of that dwelling have been implemented. Any trees or plants which within a period of 5 years of planting die, are removed, or become seriously damaged or diseased, shall be replaced with others of similar size and species.
- 5. The development hereby approved shall not be undertaken other than in accordance with drawings numbered JAN/1/8/15-02 and JAN/1/8/15-02A that show the ground level of the development at 109.51, other than with the prior approval in writing by the Local Planning Authority.
- 6. No part of the existing boundary hedge along the southern boundary of the site shall be uprooted or removed and the hedge shall not be reduced below a height of 1500mm other than to create an access in accordance with drawing number JAN/1/8/15-02, other than with the prior approval in writing by the Local Planning Authority.
- 7. The development shall not be commenced until details relating to the northern and western boundaries of the development have been submitted to and approved in writing by the Local Planning Authority. The dwelling shall not be occupied until the boundaries have been implemented in accordance with the approved details and thereafter retained.
- 8. There shall be no access or egress by any vehicles between the highway and the application site until full details of any measures required to prevent surface water from non-highway areas discharging on to the existing or proposed highway together with a programme for their implementation have been submitted to and approved in writing by the Local Planning Authority. The works shall be implemented in accordance with the approved details and programme.
- 9. Unless otherwise approved in writing by the Local Planning Authority, there shall be no excavation or other groundworks, except for investigative works, or the depositing of material on the site until the access to the site have been set out and constructed in accordance with the published Specification of the Highway Authority and the following requirements: (i) The crossing of the highway verge shall be constructed in accordance with the approved details and Standard Detail number E6. (ii) That part of the access extending 5 metres into the site from the carriageway of the existing highway shall be at a gradient not exceeding 1 in 15. (iii) The final surfacing of any private access shall not contain any loose material that is capable of being drawn on to the existing public highway. All works shall accord with the approved details unless otherwise agreed in writing by the Local Planning Authority.
- 10. No dwelling shall be occupied until the related parking facilities have been constructed in accordance with the approved drawing no JAN/1/8/15-02. Once created these parking areas shall be maintained clear of any obstruction and retained for their intended purpose at all times.

11. The permission hereby granted shall not be undertaken other than in complete accordance with the location plan and drawings numbered JAN/1/8/15-01, JAN/1/8/15-02 and JAN/1/8/15-02A received by Hambleton District Council on 22 September 2015 unless otherwise approved in writing by the Local Planning Authority.

The reasons for the above conditions are:-

- 1. To ensure compliance with Sections 91 and 92 of the Town and Country Planning Act 1990 and where appropriate as amended by Section 51 of the Planning and Compulsory Purchase Act 2004.
- 2. To ensure that the external appearance of the development is compatible with the immediate surroundings of the site and the area as a whole in accordance with Hambleton Local Development Framework Policy CP17.
- 3. To reduce the volume and rate of surface water that drains to sewers and watercourses and thereby not worsen the potential for flooding in accordance with Hambleton LDF Policies CP21 and DP43.
- 4. In order to soften the visual appearance of the development and provide any appropriate screening in accordance with LDF Policies CP16 and DP30.
- 5. To ensure the building is in keeping with the character and appearance of the locality in accordance with LDF Policies CP17 and DP32.
- 6. The existing hedge is considered to make a worthwhile contribution to the character of the area and is worthy of retention in accordance with LDF Policies CP16 and DP30.
- 7. To ensure that the development is appropriate to the character and appearance of its surroundings in accordance with LDF Policies CP16 and DP30.
- 8. In accordance with LDF Policies CP2 and DP4 and in the interests of highway safety.
- 9. In accordance with LDF Policies CP2 and DP4 and to ensure a satisfactory means of access to the site from the public highway in the interests of vehicle and pedestrian safety and convenience.
- 10. In accordance with LDF Policies CP2 and DP4 and to provide for adequate and satisfactory provision of off-street accommodation for vehicles in the interest of safety and the general amenity of the development.
- 11. In order that the development is undertaken in a form that is appropriate to the character and appearance of its surroundings and in accordance with the Development Plan Policies.